COMMITTEE NOTE

The schedules to be used in cases of individual debtors are revised as part of the Forms Modernization Project, making them easier to read and, as a result, likely to generate more complete and accurate responses. The goals of the Forms Modernization Project include improving the interface between technology and the forms so as to increase efficiency and reduce the need to produce the same information in multiple formats. Therefore, many of the open-ended questions and multiple-part instructions have been replaced with more specific questions. The individual debtor schedules are also renumbered, starting with the number 106 and followed by the letter or name of the schedule to distinguish them from the versions to be used in non-individual cases.

Official Form 106Sum, Summary of Your Assets and Liabilities and Certain Statistical Information, replaces Official Form 6, Summary of Schedules and Statistical Summary of Certain Liability and Related Data (28 U.S.C. § 159), in cases of individual debtors.

The form is reformatted and updated with crossreferences indicating the line numbers of specific schedules from which the summary information is to be gathered. In addition, because most filings are now done electronically, the form no longer requires the debtor to indicate which schedules are attached or to state the number of sheets of paper used for the schedules.

Official Form 106A/B, Schedule A/B: Property, consolidates information about an individual debtor's real and personal property into a single form. It replaces Official Form 6A, Real Property, and Official Form 6B, Personal Property, in cases of individual debtors. In addition to specific questions about the assets, the form also includes open text fields for providing additional information regarding particular assets when appropriate.

The layout and categories of property on Official Form 106A/B have changed. Instead of dividing property interests into two categories (real or personal property), the new form uses seven categories likely to be more familiar to non-lawyers: real estate, vehicles, personal household items, financial assets, business-related property, farm- and commercial fishing-related property, and a catch-all category for property that was not listed elsewhere in the form. The new form categories and the examples provided in many of the categories are designed to prompt debtors to be thorough and list all of their interests in property. The debtor may describe generally items of minimal value (such as children's clothes) by adding the value of the items and reporting the total.

Although a particular item of property may fit into more than one category, the instructions for the form explain that it should be listed only once.

In addition, because property that falls within a particular category may not be specifically elicited by the particular line items on the form, the debtor is asked in Parts 3–6 (lines 14, 35, 44, and 51) to specifically identify and value any other property in the category.

In Part 1, Describe Each Residence, Building, Land, or Other Real Estate You Own or Have an Interest In, the debtor is asked to state the "current value of the portion you own," and to also state who has an interest in the property. In addition, the debtor is asked for the nature of the ownership interest, if known by the debtor. Furthermore, instead of asking for an open-ended description of the property, the form guides the debtor in answering the description question by providing eight options from which to choose: single-family home, duplex or multi-unit building, condominium or cooperative, manufactured or mobile home, land, investment property, timeshare, and other.

Part 2, *Describe Your Vehicles*, also guides the debtor in answering the question, asking for the make, model, year, and mileage of the car or other vehicle. Because mileage is just a general indication of vehicle value, the debtor is not required to list the exact mileage, but instead is prompted to provide the approximate mileage.

Part 3, Describe Your Personal and Household Items, simplifies wording, updates categories, and uses more common terms. For example, "Wearing apparel" is changed to "Clothes" and examples include furs, which were previously grouped with jewelry. Firearms, on the

other hand, which were previously grouped with sports and other hobbies, are now set out as a separate category. Additionally, because a new Part 6 has been added to separately describe-farm related property, Part 3 includes a category for "Non-farm animals."

Part 4, *Describe Your Financial Assets*, prompts a listing of the debtor's financial assets through several questions providing separate space, after each listed type of account or deposit, for the institution or issuer name and the value of the debtor's interest in the asset. Two new categories of financial assets are added: "Bonds, mutual funds, or publicly traded stocks" and "Claims against third parties, whether or not you have filed a lawsuit or made a demand for payment." In addition, qualified ABLE accounts, as defined in 26 U.S.C. § 529A(b), are added to the list of accounts in question 24. This change is made in response to the Tax Increase Prevention Act of 2014, Pub. Law No. 113-295, which excludes ABLE account contributions meeting the specified requirements from property of the estate.

Part 5, Describe Any Business-Related Property You Own or Have an Interest In, provides prompts for listing business-related property, such as accounts receivable, inventory, and machinery, and includes a direction to list business-related real estate in Part 1, to avoid listing real estate twice.

Part 6, Describe Any Farm- and Commercial Fishing-Related Property You Own or Have an Interest In, provides prompts for listing farm- or commercial fishingrelated property, such as farm animals, crops, and feed. It also includes a direction to list any farm- or commercial fishing-related real estate in Part 1.

Part 7, Describe All Property You Own or Have an Interest in That You Did Not List Above, is a catch-all provision that allows the debtor to report property that is difficult to categorize.

Part 8, *List the Totals of Each Part of this Form*, tabulates the total value of the debtor's interest in the listed property. The tabulation includes two subtotals, one for real estate, which corresponds to the real property total that was reported on former Official Form 6A. The second

subtotal is of Parts 2-7, which corresponds to the personal property total that was reported on former Official Form 6B.

Official Form 106C, Schedule C: The Property You Claim as Exempt, replaces Official Form 6C, Property Claimed as Exempt, in cases of individual debtors.

Part 1, *Identify the Property You Claim as Exempt*, includes a table to list the property the debtor seeks to exempt, the value of the property owned by the debtor, the amount of the claimed exemption, and the law that allows the exemption. The first column asks for a brief description of the exempt property, and it also asks for the line number where the property is listed on Schedule A/B. The second column asks for the value of the portion of the asset owned by the debtor, rather than the entire asset. The third column asks for the amount, rather than the value, of the exemption claim.

The form has also been changed in light of the Supreme Court's ruling in *Schwab v. Reilly*, 560 U.S. 770 (2010). Entries in the "amount of the exemption you claim" column may now be listed as either a dollar limited amount or as 100% of fair market value, up to any applicable statutory limit. For example, a debtor might claim 100% of fair market value for a home covered by an exemption capped at \$15,000, and that limit would be applicable. This choice would impose no dollar limit where the exemption is unlimited in dollar amount, such as some exemptions for health aids, certain governmental benefits, and tax-exempt retirement funds.

Official Form 106D, Schedule D: Creditors Who Hold Claims Secured by Property, replaces Official Form 6D, Creditors Holding Secured Claims, in cases of individual debtors.

Part 1, *List Your Secured Claims*, now directs the debtor to list only the last four digits of the account number. Part 1 also adds four checkboxes with which to describe the nature of the lien: an agreement the debtor made (such as mortgage or secured car loan); statutory lien (such as tax lien, mechanic's lien); judgment lien from a lawsuit; and other.

The form adds Part 2, *List Others to Be Notified for a Debt That You Already Listed*. The debtor is instructed to use Part 2 if there is a need to notify someone about the bankruptcy filing other than the creditor for a debt listed in Part 1. For example, if a collection agency is trying to collect for a creditor listed in Part 1, the collection agency would be listed in Part 2.

Official Form 106E/F, Schedule E/F: Creditors Who Have Unsecured Claims, consolidates information about priority and nonpriority unsecured claims into a single form. It replaces Official Form 6E, Creditors Holding Unsecured Priority Claims, and Official Form 6F, Creditors Holding Unsecured Nonpriority Claims, in cases of individual debtors.

Although both priority and nonpriority unsecured claims are reported in Official Form 106E/F, the two types of claims are separately grouped so that the total for each type can be reported for case administration and statistical purposes. The form eliminates the question "consideration for claim" and instructs debtors to list claims in the alphabetical order of creditors as much as possible.

Part 1, List All of Your PRIORITY Unsecured Claims, includes four checkboxes for identifying the type of priority that applies to the claim: domestic support obligations; taxes and certain other debts owed to the government; claims for death or personal injury while intoxicated; and "other." The first three categories are required to be separately reported for statistical purposes. If the debtor selects "other," the debtor must specify the basis of the priority, e.g., wages or employee benefit plan contribution.

Part 2, *List All of Your NONPRIORITY Unsecured Claims*, contains four checkboxes, including three for types of claims that must be separately reported for statistical purposes: student loans; obligations arising out of a separation agreement or divorce not listed as priority claims; and debts to pension or profit-sharing plans and other similar debts. The remaining "other" checkbox treats claims not subject to separate reporting. If the debtor selects "other," the debtor must specify the basis of the claim.

Part 3, *List Others to Be Notified About a Debt That You Already Listed*, is new. The debtor is instructed to use Part 3 only if there is a need to give notice of the bankruptcy to someone other than a creditor listed in Parts 1 and 2. For example, if a collection agency is trying to collect for a creditor listed in Part 1, the collection agency would be listed in Part 3.

Finally, Part 4, Add the Amounts for Each Type of Unsecured Claim, requires the debtor to provide the total amounts of particular types of unsecured claims for statistical reporting purposes and the overall totals of the priority and nonpriority unsecured claims reported in this form.

Official Form 106G, Schedule G: Executory Contracts and Unexpired Leases, replaces Official Form 6G, Executory Contracts and Unexpired Leases, in cases of individual debtors.

The form is simplified. Instead of requiring the debtor to make multiple assertions about each potential executory contract or unexpired lease, the form simply requires the debtor to identify the name and address of the other party to the contract or lease, and to state what the contract or lease deals with. Definitions and examples of executory contracts and unexpired leases are included in the separate instructions for the form.

An additional page is provided in case the debtor has so many executory contracts and unexpired leases that the available page is not adequate. If the debtor needs to use the additional page, the debtor is required to fill in the entry number.

Official Form 106H, Schedule H: Your Codebtors, replaces Official Form 6H, Codebtors, in cases of individual debtors.

The form breaks out the questions about whether there are any codebters, and whether the debtor has lived with a spouse, former spouse, or legal equivalent in a community property state in the prior eight years. It also removes Alaska from the listed community property states. Finally, it asks the debtor to indicate where the debt is listed on Schedule D, Schedule E/F, or Schedule G, thereby eliminating the need to list the name and address of the creditor.

Official Form 106I, Schedule I: Your Income, replaces Official Form 6I, Your Income, in cases of individual debtors.

The form is one of an initial set of forms that were published as part of the Forms Modernization Project in 2012. It is renumbered and internal cross references are updated to conform to the new numbering system now being introduced by the Forms Modernization Project.

Official Form 106J, Schedule J: Your Expenses, replaces Official Form 6J, Your Expenses, in cases of individual debtors.

The form is one of an initial set of forms that were published as part of the Forms Modernization Project in 2012. It is renumbered and internal cross references are updated to conform to the new numbering system now being introduced by the Forms Modernization Project.

The form has been revised to include references to new Schedule J-2: Expenses for Separate Household of Debtor 2 (Official Form 106J-2) at line 1 and new line 22b. The revisions clarify how to calculate monthly net income in joint cases where Debtor 1 and Debtor 2 maintain separate households. Line 22b is added so Schedule J and Schedule J-2 are easily coordinated.

Official Form 106J-2 is new. It is used to report the monthly expenses of Debtor 2 in a joint debtor case only if Debtor 1 and Debtor 2 maintain separate households.

Official Form 106Dec, Declaration About an Individual Debtor's Schedules, replaces Official Form 6, Declaration Concerning Debtor's Schedules, in cases of individual debtors.

The form, which is to be signed by the debtor and filed with the debtor's schedules, deletes the Declaration and Signature of Bankruptcy Petition Preparer (BPP). Instead, the debtor is directed to complete and file Official Form 119, *Bankruptcy Petition Preparer's Notice*,

Declaration, and Signature, if a BPP helped fill out the bankruptcy forms.

Because the form applies only to individual debtors, it no longer contains the Declaration Under Penalty of Perjury on Behalf of a Corporation or Partnership. It also deletes from the declaration the phrase "to the best of my knowledge, information, and belief" in order to conform to the language of 28 U.S.C. § 1746. *See* Rule 1008.

1007–2 <u>Mailing – List or Matrix</u>

- (a) A mailing matrix is required to include the full names and addresses of all creditors and interested parties. A Verification of Matrix must verify that the matrix is accurate and complete to the best of the filer's knowledge, information and belief. If a case is filed electronically, the filer must upload the names and addresses on the mailing matrix to the database, and then must separately file the mailing matrix. If a case is filed on paper, the filer must submit the mailing matrix on paper in the approved format.
- (b) A debtor must include on the mailing matrix the name and address of any domestic support creditor even if all payments to that domestic support creditor are current as of the date of the filing of the bankruptcy petition. The names of minor children and confidential information should not be disclosed.
- (c) A matrix must include the taxing authority for each county where the debtor holds an interest in real estate. In a partnership case, a matrix is required to include the names and addresses of all general and/or limited partners. In a Chapter 11 case, the matrix must include the names and addresses of all equity security holders.
- (d) A debtor must promptly correct errors that are discovered on a mailing matrix.
- (e) When an amendment or a change is made to the mailing matrix which adds a creditor or entity, or changes an address to the mailing matrix, the debtor is required to serve the added creditor or entity with the debtor's Statement of Social Security Number(s); *Notice of Bankruptcy Case, Meeting of Creditors, & Deadlines*; schedule(s) listing the creditor or entity; and any other document affecting the rights of the creditor or entity.
- (f) When an amendment to a schedule adds a creditor or changes an address, the amendment to the mailing matrix must only reflect the additions or changes.
- (g) In the event of a conversion of a case from one chapter to another, the debtor must file an amendment to the mailing matrix which lists only the names and addresses of additional creditors or parties.
- Note: Information for the approved mailing matrix format may be obtained from the Clerk's Office or by clicking on the hyperlink.

The following need not be included on a mailing matrix:

- (1) Debtor(s);
- (2) Debtor's counsel
- (3) Trustee (if applicable)
- (4) United States Trustee, Region 4
- (5) United States Attorney's Office for the Southern District of West Virginia
- (6) Chief, Special Procedures Staff Internal Revenue Service

(Chapter 11 cases only)

(At the address shown on the Register of Addresses of Governmental Units)(7) West Virginia State Department of Tax and Revenue

Reference:FSee Also:SCompare:N

FBR 1007; 11 U.S.C. § 342 SDLR 1009-1 (Amendments to Lists & Schedules)

NDLR 1007-2

-
Chapter you are filing under: Chapter 7 Chapter 11 Chapter 12 Chapter 13

Check if this is an amended filing

12/15

Official Form 101

Voluntary Petition for Individuals Filing for Bankruptcy

The bankruptcy forms use you and Debtor 1 to refer to a debtor filing alone. A married couple may file a bankruptcy case together-called a

joint case—and in joint cases, these forms use you to ask for information from both debtors. For example, if a form asks, "Do you own a car," the answer would be yes if either debtor owns a car. When information is needed about the spouses separately, the form uses Debtor 1 and Debtor 2 to distinguish between them. In joint cases, one of the spouses must report information as Debtor 1 and the other as Debtor 2. The same person must be Debtor 1 in all of the forms.

Be as complete and accurate as possible. If two married people are filing together, both are equally responsible for supplying correct information. If more space is needed, attach a separate sheet to this form. On the top of any additional pages, write your name and case number (if known). Answer every question.

Part 1: **Identify Yourself**

		About Debtor 1:	About Debtor 2 (Spouse Only in a Joint Case):
1.	Your full name	e : <u>An anna 2017</u> 12	1. "************************************
	Write the name that is on your		
	government-issued picture identification (for example, your driver's license or	First name	First name
	passport).	Middle name	Middle name
	Bring your picture identification to your meeting with the trustee.	Last name	Last name
		Suffix (Sr., Jr., II, III)	Suffix (Sr., Jr., II, III)
	All other names you		an anna 12 feanna ann an Suiteanna an Suiteanna an Suiteanna an Suiteanna an Suiteanna an Suiteanna an Suiteann
-	have used in the last 8 years	First name	First name
	Include your married or maiden names.	Middle name	Middle name
		Last name	Last name
		First name	First name
		Middle name	Middle name
		Last name	Last name
1			
3.	Only the last 4 digits of your Social Security	xxx - xx	xxx - xx
	number or federal	OR .	OR
	Individual Taxpayer Identification number	9 xx - xx	9 xx - xx
	(ITIN)		

	About Debtor 1:	About Debtor 2 (Spouse Only in a Joint Case):
Any bu s iness names and Employer dentification Numbers EIN) you have used in	I have not used any business names or EINs.	I have not used any business names or EINs.
the last 8 years	Business name	Business name
doing business as names	Business name	Business name
	<u></u>	EIN
	EIN	
Where you live	99.99.99.99.99.99.99.99.99.99.99.99.99.	If Debtor 2 lives at a different address:
	Number Street	Number Street
	· · · · · · · · · · · · · · · · · · ·	<u></u>
	City State ZIP Code	City State ZIP Co
	County If your mailing address is different from the one above, fill it in here. Note that the court will send any notices to you at this mailing address.	County If Debtor 2's mailing address is different from yours, fill it in here. Note that the court will send any notices to this mailing address.
	Number Street	Number Street
	P.O. Box	P.O. Box
	City State ZIP Code	City State ZIP C
Why you are choosing <i>this district</i> to file for bankruptcy	 Check one: Over the last 180 days before filing this petition, I have lived in this district longer than in any other district. I have another reason. Explain. (See 28 U.S.C. § 1408.) 	 Check one: Over the last 180 days before filing this petition I have lived in this district longer than in any other district. I have another reason. Explain. (See 28 U.S.C. § 1408.)

	ne Last Name
t 2: Tell the Court Abou	it Your Bankruptcy Case
The chapter of the Bankruptcy Code you	Check one. (For a brief description of each, see Notice Required by 11 U.S.C. § 342(b) for Individuals Filing for Bankruptcy (Form 2010)). Also, go to the top of page 1 and check the appropriate box.
are choosing to file	Chapter 7
under	Chapter 11
	Chapter 12
	Chapter 13
How you will pay the fee	□ I will pay the entire fee when I file my petition. Please check with the clerk's office in your local court for more details about how you may pay. Typically, if you are paying the fee yourself, you may pay with cash, cashier's check, or money order. If your attorney is submitting your payment on your behalf, your attorney may pay with a credit card or check with a pre-printed address.
	□ I need to pay the fee in installments. If you choose this option, sign and attach the Application for Individuals to Pay The Filing Fee in Installments (Official Form 103A).
	□ I request that my fee be waived (You may request this option only if you are filing for Chapter By law, a judge may, but is not required to, waive your fee, and may do so only if your income is less than 150% of the official poverty line that applies to your family size and you are unable to pay the fee in installments). If you choose this option, you must fill out the <i>Application to Have th</i> <i>Chapter 7 Filing Fee Waived</i> (Official Form 103B) and file it with your petition.
Have you filed for	□ No
bankruptcy within the last 8 years?	Yes. District When Case number
last o years i	MM / DD / YYYY
	District When Case number MM / DD / YYYY
	District When Case number
	MM / DD / YYYY
Are any bankruptcy	No No
cases pending or being filed by a spouse who is	Yes. Debtor Relationship to you
not filing this case with	District When Case number, if known
you, or by a business partner, or by an	MM / DD / YYYY
affiliate?	
	Debtor Relationship to you
	District When Case number, if known MM / DD / YYYY
Do you rent your residence?	 No. Go to line 12. Yes. Has your landlord obtained an eviction judgment against you and do you want to stay in your residence?
	No. Go to line 12.

3: Report About Any E	Susiness	es You Own as a Sol	e Proprietor			
Are you a sole proprietor of any full- or part-time		o to Part 4.				
ousiness?	Yes.	Name and location of bus	iness			
A sole proprietorship is a						
ousiness you operate as an ndividual, and is not a	. • 1	Name of business, if any	-			-
eparate legal entity such as a corporation, partnership, or						
LC.	Ī	Number Street				
f you have more than one						-
ole proprietorship, use a eparate sheet and attach it						
o this petition.		City		State	ZIP Code	
		Check the appropriate bo	x to describe your busines	SS:		
		Health Care Business	a (as defined in 11 U.S.C.	§ 101(27A))		
		Single Asset Real Es	tate (as defined in 11 U.S.	C. § 101(51B))		
		Stockbroker (as defin	ed in 11 U.S.C. § 101(53A	A)))		
		Commodity Broker (a	s defined in 11 U.S.C. § 1	01(6))		
		None of the above				
debtor? For a definition of <i>small</i> business debtor, see 11 U.S.C. § 101(51D).	No.	the Bankruptcy Code.	oter 11. 11, but I am NOT a small 11 and I am a small busin			
		l am filing under Chapter Bankruptcy Code.	11 and 1 am a small busin	less deptor acci	ording to the defir	nition in the
t 4: Report if You Own	or Have /	Any Hazardous Prop	erty or Any Property 1	That Needs I	mmediate Atto	ention
Do you own or have any	No No	~				
property that poses or is alleged to pose a threat	Yes.	What is the hazard?				
of imminent and						
dentifiable hazard to						
oublic health or safety? Or do you own any						
property that needs		If immediate attention is	needed, why is it needed	?		
mmediate attention?						
erishable goods, or livestock						
hat must be fed, or a building						
		Where is the property?				
		timere le ule properdit	Number Street	-	· · · · ·	
hat needs urgent repairs?			City		State	ZIP Code

Debtor 1

Middle Name

Case number (if kno

Explain Your Efforts to Receive a Briefing About Credit Counseling Part 5:

Last Name

15. Tell the court whether you have received a briefing about credit counseling.

First Name

The law requires that you receive a briefing about credit counseling before you file for bankruptcy. You must truthfully check one of the following choices. If you cannot do so, you are not eligible to file.

If you file anyway, the court can dismiss your case, you will lose whatever filing fee you paid, and your creditors can begin collection activities again.

About Debtor 1:

You must check one:

I received a briefing from an approved credit counseling agency within the 180 days before I filed this bankruptcy petition, and I received a certificate of completion.

Attach a copy of the certificate and the payment plan, if any, that you developed with the agency.

I received a briefing from an approved credit counseling agency within the 180 days before I filed this bankruptcy petition, but I do not have a certificate of completion.

Within 14 days after you file this bankruptcy petition, you MUST file a copy of the certificate and payment plan, if any.

L certify that I asked for credit counseling services from an approved agency, but was unable to obtain those services during the 7 days after I made my request, and exigent circumstances merit a 30-day temporary waiver of the requirement.

To ask for a 30-day temporary waiver of the requirement, attach a separate sheet explaining what efforts you made to obtain the briefing, why you were unable to obtain it before you filed for bankruptcy, and what exigent circumstances required you to file this case.

Your case may be dismissed if the court is dissatisfied with your reasons for not receiving a briefing before you filed for bankruptcy.

If the court is satisfied with your reasons, you must still receive a briefing within 30 days after you file. You must file a certificate from the approved agency, along with a copy of the payment plan you developed, if any. If you do not do so, your case may be dismissed.

Any extension of the 30-day deadline is granted only for cause and is limited to a maximum of 15 days.

I am not required to receive a briefing about credit counseling because of:

- Incapacity. I have a mental illness or a mental deficiency that makes me incapable of realizing or making rational decisions about finances.
- My physical disability causes me Disability. to be unable to participate in a briefing in person, by phone, or through the internet, even after I
- Active duty. I am currently on active military duty in a military combat zone.

If you believe you are not required to receive a briefing about credit counseling, you must file a motion for waiver of credit counseling with the court.

reasonably tried to do so.

About Debtor 2 (Spouse Only in a Joint Case):

You must check one:

I received a briefing from an approved credit counseling agency within the 180 days before I filed this bankruptcy petition, and I received a certificate of completion.

Attach a copy of the certificate and the payment plan, if any, that you developed with the agency.

I received a briefing from an approved credit counseling agency within the 180 days before I filed this bankruptcy petition, but I do not have a certificate of completion.

Within 14 days after you file this bankruptcy petition, you MUST file a copy of the certificate and payment plan, if any.

I certify that I asked for credit counseling services from an approved agency, but was unable to obtain those services during the 7 days after I made my request, and exigent circumstances merit a 30-day temporary waiver of the requirement.

To ask for a 30-day temporary waiver of the requirement, attach a separate sheet explaining what efforts you made to obtain the briefing, why you were unable to obtain it before you filed for bankruptcy, and what exigent circumstances required you to file this case.

Your case may be dismissed if the court is dissatisfied with your reasons for not receiving a briefing before you filed for bankruptcy.

If the court is satisfied with your reasons, you must still receive a briefing within 30 days after you file. You must file a certificate from the approved agency, along with a copy of the payment plan you developed, if any. If you do not do so, your case may be dismissed.

Any extension of the 30-day deadline is granted only for cause and is limited to a maximum of 15 days.

I am not required to receive a briefing about credit counseling because of:

Incapacity. I have a mental illness or a mental deficiency that makes me incapable of realizing or making rational decisions about finances.

Disability.

- My physical disability causes me to be unable to participate in a briefing in person, by phone, or through the internet, even after I reasonably tried to do so.
- Active duty. I am currently on active military duty in a military combat zone.

If you believe you are not required to receive a briefing about credit counseling, you must file a motion for waiver of credit counseling with the court.

Debtor 1		Case number (if km	own)
First Name Middle Name	a Last Name		
art 6: Answer These Ques	tions for Reporting Purpos	ses	
6. What kind of debts do	16a. Are your debts prima as "incurred by an individu	rily consumer debts? Consumer debts and primarily for a personal, family, or hou	ots are defined in 11 U.S.C. § 101(8) sehold purpose."
you have?	No. Go to line 16b.Yes. Go to line 17.		
	16b. Are your debts prima	rily business debts? Business debts nvestment or through the operation of the	
	No. Go to line 16c.Yes. Go to line 17.		
	16c. State the type of debts yo	u owe that are not consumer debts or bu	siness debts.
7. Are you filing under Chapter 7?	No. I am not filing under C	hapter 7. Go to line 18.	
Do you estimate that after any exempt property is	Yes. I am filing under Chap	ter 7. Do you estimate that after any exe es are paid that funds will be available to	
excluded and	D No		
administrative expenses are paid that funds will be available for distribution to unsecured creditors?	C Yes		
8. How many creditors do	1 -49	1,000-5,000	25,001-50,000
you estimate that you owe?	50-99 100-199 200-999	5,001-10,00010,001-25,000	 50,001-100,000 More than 100,000
9. How much do you	\$0-\$50,000	\$1,000,001-\$10 million	\$500,000,001-\$1 billion
estimate your assets to be worth?	□ \$50,001-\$100,000 □ \$100,001-\$500,000	 \$10,000,001-\$50 million \$50,000,001-\$100 million 	\$1,000,000,001-\$10 billion \$10,000,000,001-\$50 billion
	S500,001-\$1 million	\$100,000,001-\$500 million	More than \$50 billion
0. How much do you	\$0-\$50,000	\$1,000,001-\$10 million	\$500,000,001-\$1 billion
estimate your liabilities to be?	\$50,001-\$100,000	□ \$10,000,001-\$50 million	\$1,000,000,001-\$10 billion
to be?	 \$100,001-\$500,000 \$500,001-\$1 million 	 \$50,000,001-\$100 million \$100,000,001-\$500 million 	 \$10,000,000,001-\$50 billion More than \$50 billion
Part 7: Sign Below			
For you	I have examined this petition, a correct.	and I declare under penalty of perjury that	t the information provided is true and
	If I have chosen to file under C of title 11, United States Code. under Chapter 7.	hapter 7, I am aware that I may proceed I understand the relief available under e	, if eligible, under Chapter 7, 11,12, or 13 ach chapter, and I choose to proceed
	If no attorney represents me and I did not pay or agree to pay someone who is not an attorney to help me fill out this document, I have obtained and read the notice required by 11 U.S.C. § 342(b).		
		with the chapter of title 11, United States	
	l understand making a false st with a bankruptcy case can res 18 U.S.C. §§ 152, 1341, 1519,	sult in fines up to \$250,000, or imprisonm	g money or property by fraud in connection lent for up to 20 years, or both.
	×	× .	
	Signature of Debtor 1		re of Debtor 2
	Executed on	Execute	ed on
1	MM / DD		MM / DD / YYYY

Voluntary Petition for Individuals Filing for Bankruptcy

ptor 1 First Name Middle Name	Last Name	Case number (if kno	
23			
your attorney, if you are resented by one	to proceed under Chapter 7, 11, 12, or	r 13 of title 11, United States Code n the person is eligible. I also cert	ify that I have delivered to the debtor(s)
ou are not represented an attorney, you do not	knowledge after an inquiry that the info	ormation in the schedules filed with	h the petition is incorrect.
ed to file this page.	×	Date	
	Signature of Attorney for Debtor		MM / DD /YYYY
	Printed name	· · ·	
			-
	Firm name		
	Number Street		
	· · · · · · · · · · · · · · · · · · ·		
			·
	City	State	ZIP Code
		-	
	Contact phone	Email add	dress
			· · · · · · · · · · · · · · · · · · ·
-	Barnumber	State	

Debtor 1

Case number (if know

For you if you are filing this bankruptcy without an attorney

First Name

If you are represented by an attorney, you do not need to file this page. The law allows you, as an individual, to represent yourself in bankruptcy court, but you should understand that many people find it extremely difficult to represent themselves successfully. Because bankruptcy has long-term financial and legal consequences, you are strongly urged to hire a qualified attorney.

To be successful, you must correctly file and handle your bankruptcy case. The rules are very technical, and a mistake or inaction may affect your rights. For example, your case may be dismissed because you did not file a required document, pay a fee on time, attend a meeting or hearing, or cooperate with the court, case trustee, U.S. trustee, bankruptcy administrator, or audit firm if your case is selected for audit. If that happens, you could lose your right to file another case, or you may lose protections, including the benefit of the automatic stay.

You must list all your property and debts in the schedules that you are required to file with the court. Even if you plan to pay a particular debt outside of your bankruptcy, you must list that debt in your schedules. If you do not list a debt, the debt may not be discharged. If you do not list property or properly claim it as exempt, you may not be able to keep the property. The judge can also deny you a discharge of all your debts if you do something dishonest in your bankruptcy case, such as destroying or hiding property, falsifying records, or lying. Individual bankruptcy cases are randomly audited to determine if debtors have been accurate, truthful, and complete. Bankruptcy fraud is a serious crime; you could be fined and imprisoned.

If you decide to file without an attorney, the court expects you to follow the rules as if you had hired an attorney. The court will not treat you differently because you are filing for yourself. To be successful, you must be familiar with the United States Bankruptcy Code, the Federal Rules of Bankruptcy Procedure, and the local rules of the court in which your case is filed. You must also be familiar with any state exemption laws that apply.

Are you aware that filing for bankruptcy is a serious action with long-term financial and legal consequences?

🔲 No

Yes

Are you aware that bankruptcy fraud is a serious crime and that if your bankruptcy forms are inaccurate or incomplete, you could be fined or imprisoned?

O No

🗋 Yes

Did you pay or agree to pay someone who is not an attorney to help you fill out your bankruptcy forms?

Yes. Name of Person

Attach Bankruptcy Petition Preparer's Notice, Declaration, and Signature (Official Form 119).

By signing here, I acknowledge that I understand the risks involved in filing without an attorney. I have read and understood this notice, and I am aware that filing a bankruptcy case without an attorney may cause me to lose my rights or property if I do not properly handle the case.

X	× * *	
Signature of Debtor 1		Signature of Debtor 2
Date		Date MM / DD / YYYY
Contact phone	· .	Contact phone
Cell phone		Cell phone
Email address	· · · ·	Email address

Fill in this information to identify your case:			
Debtor 1	Last Name		
Debtor 2 (Spouse, if filing) First Name Middle Name	Last Name		
United States Bankruptcy Court for the:	District of (State)		Check if this is an
Case number (If known)			amended filing

Official Form 106Sum

Summary of Your Assets and Liabilities and Certain Statistical Information 12/15

Be as complete and accurate as possible. If two married people are filing together, both are equally responsible for supplying correct information. Fill out all of your schedules first; then complete the information on this form. If you are filing amended schedules after you file your original forms, you must fill out a new *Summary* and check the box at the top of this page.

Part 1: Summarize Your Assets

		Your assets Value of what you own
	1. Schedule A/B: Property (Official Form 106A/B)	
	1a. Copy line 55, Total real estate, from Schedule A/B	
	1b. Copy line 62, Total personal property, from Schedule A/B	
	1c. Copy line 63, Total of all property on Schedule A/B	
		\$
	Part 2: Summarize Your Liabilities	
		Your liabilities
		Amount you owe
	 Schedule D: Creditors Who Have Claims Secured by Property (Official Form 106D) 2a. Copy the total you listed in Column A, Amount of claim, at the bottom of the last page of Part 1 of Schedule D 	\$
	 Schedule E/F: Creditors Who Have Unsecured Claims (Official Form 106E/F) 3a. Copy the total claims from Part 1 (priority unsecured claims) from line 6e of Schedule E/F 	\$
	3b. Copy the total claims from Part 2 (nonpriority unsecured claims) from line 6j of Schedule E/F	···· + \$
	Your total liabilities	s
	Part 3: Summarize Your Income and Expenses	
	4. Schedule I: Your Income (Official Form 106I)	
ĺ	Copy your combined monthly income from line 12 of Schedule I	\$
	Copy your combined monthly moonic norm into 12 or concoder a statistical statistic	
	5. Schedule J: Your Expenses (Official Form 106J)	
	Copy your monthly expenses from line 22c of Schedule J	\$

þ	Debtor 1 Case number (<i>if known</i>)
l	Part 4: Answer These Questions for Administrative and Statistical Records
6	6. Are you filing for bankruptcy under Chapters 7, 11, or 13?
	 No. You have nothing to report on this part of the form. Check this box and submit this form to the court with your other schedules. Yes
7	7. What kind of debt do you have?
	Your debts are primarily consumer debts. Consumer debts are those "incurred by an individual primarily for a personal, family, or household purpose." 11 U.S.C. § 101(8). Fill out lines 8-9g for statistical purposes. 28 U.S.C. § 159.
	Your debts are not primarily consumer debts. You have nothing to report on this part of the form. Check this box and submit this form to the court with your other schedules.
8	From the Statement of Your Current Monthly Income: Copy your total current monthly income from Official Form 122A-1 Line 11; OR, Form 122B Line 11; OR, Form 122C-1 Line 14.
9	e. Copy the following special categories of claims from Part 4, line 6 of <i>Schedule E/F</i> : Total claim
	From Part 4 on Schedule E/F, copy the following:
1	9a. Domestic support obligations (Copy line 6a.)
	9b. Taxes and certain other debts you owe the government. (Copy line 6b.) \$
	9c. Claims for death or personal injury while you were intoxicated. (Copy line 6c.)
	9d. Student loans. (Copy line 6f.)
	9e. Obligations arising out of a separation agreement or divorce that you did not report as priority claims. (Copy line 6g.)
	9f. Debts to pension or profit-sharing plans, and other similar debts. (Copy line 6h.) + \$
	9g. Total. Add lines 9a through 9f. \$
-	

Fill in this information to identify your case and this filing:					
Debtor 1					
	First Name	Middle Name	Last Name		
Debtor 2 (\$pouse, if filing)	First Name	Middle Name	Last Name		
United States E	ankruptcy Court for the:		District of(State)		
Case number					

Check if this is an amended filing

Official Form 106A/B

Schedule A/B: Property

12/15

In each category, separately list and describe items. List an asset only once. If an asset fits in more than one category, list the asset in the category where you think it fits best. Be as complete and accurate as possible. If two married people are filing together, both are equally responsible for supplying correct information. If more space is needed, attach a separate sheet to this form. On the top of any additional pages, write your name and case number (if known). Answer every question.

Part 1:	Describe Each Residence, Building,	Land, or Other Real Estate You Own or Hav	e an Interest in	
1. Do yo	ou own or have any legal or equitable interes	st in any residence, building, land, or similar prop	erty?	
	o. Go to Part 2. es. Where is the property?			
1.1.	Street address, if available, or other description	 What is the property? Check all that apply. Single-family home Duplex or multi-unit building 	Do not deduct secured cla the amount of any secure Creditors Who Have Clair	d claims on Schedule D:
	Street address, if available, or other description	 Condominium or cooperative Manufactured or mobile home 	Current value of the entire property?	Current value of the portion you own?
		Land	\$	\$
	City State ZIP Code	 Investment property Timeshare Other 	Describe the nature of interest (such as fee the entireties, or a life	simple, tenancy by
		Who has an interest in the property? Check one.		,,
		Debtor 1 only		
	County	 Debtor 2 only Debtor 1 and Debtor 2 only 	Check if this is co (see instructions)	mmunity property
		At least one of the debtors and another Other information you wish to add about this it property identification number:		
If you	own or have more than one, list here:			
1.2.		 What is the property? Check all that apply. Single-family home Duplex or multi-unit building 	Do not deduct secured cla the amount of any secure <i>Creditors Who Have Clair</i>	d claims on Schedule D:
	Street address, if available, or other description	 Condominium or cooperative Manufactured or mobile home 	Current value of the entire property?	Current value of the portion you own?
		Land	\$	\$
	City State ZIP Code		Describe the nature of your ownership interest (such as fee simple, tenancy by the entireties, or a life estate), if known	
		Who has an interest in the property? Check one.		
		Debtor 1 only		
	Caush	Debtor 2 only		
	County	Debtor 1 and Debtor 2 only	Check if this is co	ommunity property
		At least one of the debtors and another	(see instructions)	
		Other information you wish to add about this ite property identification number:		

Debtor 1		Case number (if ki	nown)	
	First Name Middle Name Last Name			
·		· · ·		
		What is the property? Check all that apply.	Do not deduct secured cla	ime or exemptions. But
		Single-family home	the amount of any secured	
1.3.	Street address, if available, or other description	Duplex or multi-unit building	Creditors Who Have Clain	ns Secured by Property.
	Street address, if available, or other description	Condominium or cooperative	Current value of the	Current value of th
		Manufactured or mobile home	entire property?	portion you own?
			\$	\$
		Investment property	- ,	
· · .	City State ZIP Code		Describe the nature of	of your ownership
	City State ZIP Code	Other	interest (such as fee	
· .			the entireties, or a life	e estate), if known.
		Who has an interest in the property? Check one.		
		Debtor 1 only		
	County	Debtor 2 only		
		Debtor 1 and Debtor 2 only	Check if this is co	mmunity property
		At least one of the debtors and another	(see instructions)	
			m evel tt	
		Other information you wish to add about this ite property identification number:	m, such as local	
art 2:	Describe Your Vehicles			
o you (own, lease, or have legal or equitable interes	st in any vehicles, whether they are registered or the second secon	not? Include any vehicles and Unexpired Leases.	S
o you o ou own	own, lease, or have legal or equitable interest that someone else drives. If you lease a vehicl	e, also report it on Schedule G: Executory Contracts	not? Include any vehicles and Unexpired Leases.	S
o you o ou own Cars	own, lease, or have legal or equitable interest that someone else drives. If you lease a vehicl , vans, trucks, tractors, sport utility vehicles	e, also report it on Schedule G: Executory Contracts	not? Include any vehicles and Unexpired Leases.	S
o you o bu own Cars	own, lease, or have legal or equitable interest that someone else drives. If you lease a vehicl , vans, trucks, tractors, sport utility vehicles lo	e, also report it on Schedule G: Executory Contracts	not? Include any vehicles and Unexpired Leases.	S
o you o ou own Cars	own, lease, or have legal or equitable interest that someone else drives. If you lease a vehicl , vans, trucks, tractors, sport utility vehicles lo	e, also report it on Schedule G: Executory Contracts	not? Include any vehicles and Unexpired Leases.	S
o you o ou own Cars N Y	own, lease, or have legal or equitable interes that someone else drives. If you lease a vehicle , vans, trucks, tractors, sport utility vehicles to es	e, also report it on Schedule G: Executory Contracts	and Unexpired Leases.	
oyou ouown Cars	own, lease, or have legal or equitable interes that someone else drives. If you lease a vehicle , vans, trucks, tractors, sport utility vehicles lo es Make:	e, also report it on <i>Schedule G: Executory Contracts</i> of motorcycles Who has an interest in the property? Check one.	Do not deduct secured cla the amount of any secure	aims or exemptions. Put d claims on <i>Schedule D</i> :
o you o ou own Cars N Y	own, lease, or have legal or equitable interes that someone else drives. If you lease a vehicle , vans, trucks, tractors, sport utility vehicles to es	e, also report it on <i>Schedule G: Executory Contracts</i> , motorcycles Who has an interest in the property? Check one. Debtor 1 only	and Unexpired Leases. Do not deduct secured cla	aims or exemptions. Put d claims on <i>Schedule D</i> :
o you o ou own Cars N Y	own, lease, or have legal or equitable interes that someone else drives. If you lease a vehicle , vans, trucks, tractors, sport utility vehicles lo es Make:	e, also report it on <i>Schedule G: Executory Contracts</i> , motorcycles Who has an interest in the property? Check one. Debtor 1 only Debtor 2 only	Do not deduct secured cla the amount of any secure <i>Creditors Who Have Clair</i> Current value of the	aims or exemptions. Put d claims on <i>Schedule D:</i> <i>ns Secured by Property.</i> Current value of th
o you o ou own Cars N Y	own, lease, or have legal or equitable interest that someone else drives. If you lease a vehicle , vans, trucks, tractors, sport utility vehicles lo es Make: Model:	 e, also report it on <i>Schedule G: Executory Contracts</i> motorcycles Who has an interest in the property? Check one. Debtor 1 only Debtor 2 only Debtor 1 and Debtor 2 only 	Do not deduct secured cla the amount of any secure <i>Creditors Who Have Clair</i>	aims or exemptions. Put d claims on <i>Schedule D:</i> <i>ns Secured by Property</i> .
o you o ou own Cars N Y	own, lease, or have legal or equitable interest that someone else drives. If you lease a vehicle , vans, trucks, tractors, sport utility vehicles lo es Make: Year: Approximate mileage:	e, also report it on <i>Schedule G: Executory Contracts</i> , motorcycles Who has an interest in the property? Check one. Debtor 1 only Debtor 2 only	Do not deduct secured cla the amount of any secure <i>Creditors Who Have Clair</i> Current value of the	aims or exemptions. Put d claims on <i>Schedule D:</i> <i>ns Secured by Property.</i> Current value of th
o you o ou own Cars N Y	own, lease, or have legal or equitable interest that someone else drives. If you lease a vehicles , vans, trucks, tractors, sport utility vehicles lo es Make: Model: Year:	 e, also report it on <i>Schedule G: Executory Contracts</i> , motorcycles Who has an interest in the property? Check one. Debtor 1 only Debtor 2 only Debtor 1 and Debtor 2 only At least one of the debtors and another 	Do not deduct secured cla the amount of any secure <i>Creditors Who Have Clair</i> Current value of the	aims or exemptions. Put d claims on <i>Schedule D:</i> <i>ns Secured by Property.</i> Current value of th
o you o ou own Cars N Y	own, lease, or have legal or equitable interest that someone else drives. If you lease a vehicle , vans, trucks, tractors, sport utility vehicles lo es Make: Year: Approximate mileage:	 e, also report it on <i>Schedule G: Executory Contracts</i> motorcycles Who has an interest in the property? Check one. Debtor 1 only Debtor 2 only Debtor 1 and Debtor 2 only 	Do not deduct secured cla the amount of any secure <i>Creditors Who Have Clair</i> Current value of the	aims or exemptions. Put d claims on Schedule D: ns Secured by Property. Current value of th
o you o ou own Cars N N Y	own, lease, or have legal or equitable interest that someone else drives. If you lease a vehicle , vans, trucks, tractors, sport utility vehicles lo es Make: Year: Approximate mileage:	 e, also report it on Schedule G: Executory Contracts of motorcycles Who has an interest in the property? Check one. Debtor 1 only Debtor 2 only Debtor 1 and Debtor 2 only At least one of the debtors and another Check if this is community property (see 	Do not deduct secured cla the amount of any secure <i>Creditors Who Have Clair</i> Current value of the	aims or exemptions. Put d claims on Schedule D ns Secured by Property. Current value of tl
o you o ou own Cars N Y	own, lease, or have legal or equitable interest that someone else drives. If you lease a vehicle , vans, trucks, tractors, sport utility vehicles lo es Make: Year: Approximate mileage:	 e, also report it on Schedule G: Executory Contracts of motorcycles Who has an interest in the property? Check one. Debtor 1 only Debtor 2 only Debtor 1 and Debtor 2 only At least one of the debtors and another Check if this is community property (see 	Do not deduct secured cla the amount of any secure <i>Creditors Who Have Clair</i> Current value of the	aims or exemptions. Put d claims on <i>Schedule D:</i> <i>ns Secured by Property.</i> Current value of th
o you o ou own Cars N Y 3.1.	own, lease, or have legal or equitable interest that someone else drives. If you lease a vehicle , vans, trucks, tractors, sport utility vehicles lo es Make: Year: Approximate mileage:	 e, also report it on Schedule G: Executory Contracts of motorcycles Who has an interest in the property? Check one. Debtor 1 only Debtor 2 only Debtor 1 and Debtor 2 only At least one of the debtors and another Check if this is community property (see 	Do not deduct secured cla the amount of any secure <i>Creditors Who Have Clair</i> Current value of the	aims or exemptions. Put d claims on <i>Schedule D:</i> <i>ns Secured by Property.</i> Current value of th
o you o bu own Cars N Y 3.1.	own, lease, or have legal or equitable interest that someone else drives. If you lease a vehicle, vans, trucks, tractors, sport utility vehicles to the set of the set	 e, also report it on Schedule G: Executory Contracts of motorcycles Who has an interest in the property? Check one. Debtor 1 only Debtor 2 only Debtor 1 and Debtor 2 only At least one of the debtors and another Check if this is community property (see instructions) 	and Unexpired Leases. Do not deduct secured cla the amount of any secure <i>Creditors Who Have Clair</i> Current value of the entire property? \$	aims or exemptions. Put d claims on <i>Schedule D:</i> <i>ns Secured by Property.</i> Current value of th portion you own? \$
Cars Cars N Y 3.1.	own, lease, or have legal or equitable interest that someone else drives. If you lease a vehicle, vans, trucks, tractors, sport utility vehicles to the set of the set	 e, also report it on Schedule G: Executory Contracts of motorcycles Who has an interest in the property? Check one. Debtor 1 only Debtor 2 only Debtor 1 and Debtor 2 only At least one of the debtors and another Check if this is community property (see instructions) Who has an interest in the property? Check one. 	Do not deduct secured cla the amount of any secure <i>Creditors Who Have Clair</i> Current value of the entire property? \$ Do not deduct secured cla	aims or exemptions. Put d claims on <i>Schedule D:</i> <i>ns Secured by Property.</i> Current value of th portion you own? \$
Cars Cars N Y 3.1.	own, lease, or have legal or equitable interest that someone else drives. If you lease a vehicle, vans, trucks, tractors, sport utility vehicles to the set of the set	 e, also report it on Schedule G: Executory Contracts of motorcycles Who has an interest in the property? Check one. Debtor 1 only Debtor 2 only Debtor 1 and Debtor 2 only At least one of the debtors and another Check if this is community property (see instructions) Who has an interest in the property? Check one. Debtor 1 only 	and Unexpired Leases. Do not deduct secured cla the amount of any secure <i>Creditors Who Have Clair</i> Current value of the entire property? \$	aims or exemptions. Put d claims on <i>Schedule D:</i> <i>ms Secured by Property.</i> Current value of th portion you own? \$
Cars Cars N Y 3.1.	own, lease, or have legal or equitable interest that someone else drives. If you lease a vehicles of you lease a vehicles of the someone else drives. If you lease a vehicles of the someone else drives. If you lease a vehicles of the someone else drives. If you lease a vehicles of the someone else drives. If you lease a vehicles of the someone else drives. If you lease a vehicles of the someone else drives. If you lease a vehicle someone else drives. If you lease drives a vehicle someone else drives. If you lease a	 e, also report it on Schedule G: Executory Contracts of motorcycles Who has an interest in the property? Check one. Debtor 1 only Debtor 2 only Debtor 1 and Debtor 2 only At least one of the debtors and another Check if this is community property (see instructions) Who has an interest in the property? Check one. Debtor 1 only Debtor 2 only 	Do not deduct secured cla the amount of any secure <i>Creditors Who Have Clair</i> Current value of the entire property? \$ Do not deduct secured cla the amount of any secure <i>Creditors Who Have Clair</i>	aims or exemptions. Put d claims on <i>Schedule D:</i> <i>ns Secured by Property.</i> Current value of th portion you own? \$
o you own Cars N Y 3.1.	own, lease, or have legal or equitable interest that someone else drives. If you lease a vehicle, vans, trucks, tractors, sport utility vehicles to the set of the set	 e, also report it on Schedule G: Executory Contracts of motorcycles Who has an interest in the property? Check one. Debtor 1 only Debtor 2 only Debtor 1 and Debtor 2 only At least one of the debtors and another Check if this is community property (see instructions) Who has an interest in the property? Check one. Debtor 1 only Debtor 2 only Debtor 1 only 	Do not deduct secured cla the amount of any secure <i>Creditors Who Have Clair</i> Current value of the entire property? \$ Do not deduct secured cla the amount of any secure	aims or exemptions. Put d claims on <i>Schedule D:</i> <i>ns Secured by Property.</i> Current value of th portion you own? \$
o you own Cars N Y 3.1.	own, lease, or have legal or equitable interest that someone else drives. If you lease a vehicles of you lease a vehicles of the someone else drives. If you lease a vehicles of the someone else drives. If you lease a vehicles of the someone else drives. If you lease a vehicles of the someone else drives. If you lease a vehicles of the someone else drives. If you lease a vehicles of the someone else drives. If you lease a vehicle someone else drives. If you lease drives a vehicle someone else drives. If you lease a	 e, also report it on Schedule G: Executory Contracts of motorcycles Who has an interest in the property? Check one. Debtor 1 only Debtor 2 only Debtor 1 and Debtor 2 only At least one of the debtors and another Check if this is community property (see instructions) Who has an interest in the property? Check one. Debtor 1 only Debtor 2 only 	Do not deduct secured cla the amount of any secure <i>Creditors Who Have Clair</i> Current value of the entire property? \$ Do not deduct secured cla the amount of any secure <i>Creditors Who Have Clair</i> Current value of the	aims or exemptions. Put d claims on Schedule D: ns Secured by Property. Current value of th portion you own? \$
Cars Cars N Y 3.1.	own, lease, or have legal or equitable interest that someone else drives. If you lease a vehicle, vans, trucks, tractors, sport utility vehicles to the set of the set	 e, also report it on Schedule G: Executory Contracts of motorcycles Who has an interest in the property? Check one. Debtor 1 only Debtor 2 only Debtor 1 and Debtor 2 only At least one of the debtors and another Check if this is community property (see instructions) Who has an interest in the property? Check one. Debtor 1 only Debtor 2 only At least one of the debtors and another 	Do not deduct secured cla the amount of any secure <i>Creditors Who Have Clair</i> Current value of the entire property? \$ Do not deduct secured cla the amount of any secure <i>Creditors Who Have Clair</i> Current value of the	aims or exemptions. Put d claims on <i>Schedule D:</i> <i>ns Secured by Property.</i> Current value of th portion you own? \$
o you own Cars N Y 3.1.	own, lease, or have legal or equitable interest that someone else drives. If you lease a vehicles of you lease a vehicles of the set o	 e, also report it on Schedule G: Executory Contracts of motorcycles Who has an interest in the property? Check one. Debtor 1 only Debtor 2 only Debtor 1 and Debtor 2 only At least one of the debtors and another Check if this is community property (see instructions) Who has an interest in the property? Check one. Debtor 1 only Debtor 2 only Debtor 1 only 	Do not deduct secured cla the amount of any secure <i>Creditors Who Have Clair</i> Current value of the entire property? \$ Do not deduct secured cla the amount of any secure <i>Creditors Who Have Clair</i> Current value of the	aims or exemptions. Put d claims on <i>Schedule D:</i> <i>ns Secured by Property.</i> Current value of th portion you own? \$

- - 			•
	When here an interpret in the property? Check are		
Make:	Who has an interest in the property? Check one.	Do not deduct secured cla the amount of any secured	
Model:	Debtor 1 only	Creditors Who Have Clain	
Year:	Debtor 2 only	Current value of the	Current value of t
	Debtor 1 and Debtor 2 only	entire property?	portion you own?
Approximate mileage:	At least one of the debtors and another		
Other information:	Check if this is community property (see	\$	\$
	instructions)		
Make:	Who has an interest in the property? Check one.	Do not deduct secured cla	
	Debtor 1 only	the amount of any secure Creditors Who Have Clain	
Model:	Debtor 2 only	Creditors who have Clain	ns Secured by Propert
Year:	Debtor 1 and Debtor 2 only	Current value of the	Current value of
Approximate mileage:	At least one of the debtors and another	entire property?	portion you own?
Other information:			
	Check if this is community property (see	\$	\$
	instructions)		
Yes	Who has an interest in the property? Check one.	Do not deduct required al	ing of compliants D
Make:		Do not deduct secured cla the amount of any secure Condition Mithe Nam Claim	d claims on Schedule
Make:	Who has an interest in the property? Check one.		d claims on Schedule
Make:	Debtor 1 only	the amount of any secure	d claims on Schedule ns Secured by Propen
Make:	Debtor 1 only	the amount of any secure Creditors Who Have Clair	d claims on Schedule ns Secured by Propert Current value of
Make: Model: Year:	 Debtor 1 only Debtor 2 only Debtor 1 and Debtor 2 only At least one of the debtors and another 	the amount of any secure Creditors Who Have Clair Current value of the	d claims on Schedule i ns Secured by Propert Current value of
Make: Model: Year:	 Debtor 1 only Debtor 2 only Debtor 1 and Debtor 2 only 	the amount of any secure Creditors Who Have Clair Current value of the	d claims on Schedule ns Secured by Propert Current value of
Make: Model: Year:	 Debtor 1 only Debtor 2 only Debtor 1 and Debtor 2 only At least one of the debtors and another Check if this is community property (see 	the amount of any secure Creditors Who Have Clair Current value of the	d claims on Schedule ns Secured by Propert Current value of
Make: Model: Year:	 Debtor 1 only Debtor 2 only Debtor 1 and Debtor 2 only At least one of the debtors and another Check if this is community property (see instructions) 	the amount of any secure Creditors Who Have Clair Current value of the	d claims on Schedule ns Secured by Propert Current value of
Make: Model: Year: Other information:	 Debtor 1 only Debtor 2 only Debtor 1 and Debtor 2 only Debtor 1 and Debtor 2 only At least one of the debtors and another Check if this is community property (see instructions) 	the amount of any secure Creditors Who Have Clair Current value of the entire property? \$ Do not deduct secured cla	d claims on Schedule ns Secured by Proper Current value of portion you own \$
Make: Model: Year: Other information: ou own or have more than one Make:	 Debtor 1 only Debtor 2 only Debtor 1 and Debtor 2 only Debtor 1 and Debtor 2 only At least one of the debtors and another Check if this is community property (see instructions) , list here: Who has an interest in the property? Check one. Debtor 1 only 	the amount of any secure Creditors Who Have Clair Current value of the entire property? \$	d claims on Schedule ns Secured by Proper Current value of portion you own \$
Make: Model: Year: Other information: ou own or have more than one Make: Model:	 Debtor 1 only Debtor 2 only Debtor 1 and Debtor 2 only At least one of the debtors and another Check if this is community property (see instructions) , list here: Who has an interest in the property? Check one. Debtor 1 only Debtor 2 only 	the amount of any secure Creditors Who Have Clair Current value of the entire property? \$ Do not deduct secured clair the amount of any secure Creditors Who Have Clair	d claims on Schedule ns Secured by Propen Current value of portion you own \$
Make: Model: Year: Other information: ou own or have more than one Make: Model: Year:	 Debtor 1 only Debtor 2 only Debtor 1 and Debtor 2 only At least one of the debtors and another Check if this is community property (see instructions) Itist here: Who has an interest in the property? Check one. Debtor 1 only Debtor 2 only Debtor 1 and Debtor 2 only 	the amount of any secure Creditors Who Have Clair Current value of the entire property? \$ Do not deduct secured cli the amount of any secure	d claims on Schedule ns Secured by Proper Current value of portion you own \$
Make: Model: Year: Other information: Tou own or have more than one Make: Model:	 Debtor 1 only Debtor 2 only Debtor 1 and Debtor 2 only At least one of the debtors and another Check if this is community property (see instructions) , list here: Who has an interest in the property? Check one. Debtor 1 only Debtor 2 only 	the amount of any secure Creditors Who Have Clair Current value of the entire property? \$ Do not deduct secured clair the amount of any secure Creditors Who Have Clair Current value of the	d claims on Schedule ns Secured by Proper Current value of portion you own \$
Make: Model: Year: Other information: fou own or have more than one Make: Model: Year:	 Debtor 1 only Debtor 2 only Debtor 1 and Debtor 2 only At least one of the debtors and another Check if this is community property (see instructions) I list here: Who has an interest in the property? Check one. Debtor 1 only Debtor 2 only Debtor 1 and Debtor 2 only At least one of the debtors and another 	the amount of any secure Creditors Who Have Clair Current value of the entire property? \$ Do not deduct secured clair the amount of any secure Creditors Who Have Clair Current value of the	d claims on Schedule ns Secured by Proper Current value of portion you own \$
Make: Model: Year: Other information: fou own or have more than one Make: Model: Year:	 Debtor 1 only Debtor 2 only Debtor 1 and Debtor 2 only At least one of the debtors and another Check if this is community property (see instructions) I list here: Who has an interest in the property? Check one. Debtor 1 only Debtor 2 only Debtor 1 and Debtor 2 only At least one of the debtors and another 	the amount of any secure Creditors Who Have Clair Current value of the entire property? \$ Do not deduct secured clair the amount of any secure Creditors Who Have Clair Current value of the	d claims on Schedule ns Secured by Proper Current value of portion you own \$
Make: Model: Year: Other information: ou own or have more than one Make: Model: Year:	 Debtor 1 only Debtor 2 only Debtor 1 and Debtor 2 only At least one of the debtors and another Check if this is community property (see instructions) I list here: Who has an interest in the property? Check one. Debtor 1 only Debtor 2 only Debtor 1 and Debtor 2 only At least one of the debtors and another 	the amount of any secure Creditors Who Have Clair Current value of the entire property? \$ Do not deduct secured clair the amount of any secure Creditors Who Have Clair Current value of the	d claims on Schedule ns Secured by Propen Current value of portion you own \$
Make: Model: Year: Other information: fou own or have more than one Make: Model: Year:	 Debtor 1 only Debtor 2 only Debtor 1 and Debtor 2 only At least one of the debtors and another Check if this is community property (see instructions) I list here: Who has an interest in the property? Check one. Debtor 1 only Debtor 2 only Debtor 1 and Debtor 2 only At least one of the debtors and another 	the amount of any secure Creditors Who Have Clair Current value of the entire property? \$ Do not deduct secured clair the amount of any secure Creditors Who Have Clair Current value of the	d claims on Schedule ns Secured by Proper Current value of portion you own \$
Make: Model: Year: Other information: fou own or have more than one Make: Model: Year:	 Debtor 1 only Debtor 2 only Debtor 1 and Debtor 2 only At least one of the debtors and another Check if this is community property (see instructions) I list here: Who has an interest in the property? Check one. Debtor 1 only Debtor 2 only Debtor 1 and Debtor 2 only At least one of the debtors and another 	the amount of any secure Creditors Who Have Clair Current value of the entire property? \$ Do not deduct secured clair the amount of any secure Creditors Who Have Clair Current value of the	d claims on Schedule ns Secured by Proper Current value of portion you own \$
Make: Model: Year: Other information: fou own or have more than one Make: Model: Year:	 Debtor 1 only Debtor 2 only Debtor 1 and Debtor 2 only At least one of the debtors and another Check if this is community property (see instructions) I list here: Who has an interest in the property? Check one. Debtor 1 only Debtor 2 only Debtor 1 and Debtor 2 only At least one of the debtors and another 	the amount of any secure Creditors Who Have Clair Current value of the entire property? \$ Do not deduct secured clair the amount of any secure Creditors Who Have Clair Current value of the	d claims on Schedule ns Secured by Proper Current value of portion you own \$
Make: Model: Year: Other information: wou own or have more than one Make: Model: Year: Other information:	 Debtor 1 only Debtor 2 only Debtor 1 and Debtor 2 only At least one of the debtors and another Check if this is community property (see instructions) I ist here: Who has an interest in the property? Check one. Debtor 1 only Debtor 2 only Debtor 2 only Debtor 1 and Debtor 2 only At least one of the debtors and another Check if this is community property (see instructions) 	the amount of any secure Creditors Who Have Clair Current value of the entire property? \$ Do not deduct secured cl the amount of any secure Creditors Who Have Clair Current value of the entire property? \$	d claims on Schedule ns Secured by Propen Current value of portion you own \$
Make: Model: Year: Other information: Make: Model: Year: Other information: Other information:	 Debtor 1 only Debtor 2 only Debtor 1 and Debtor 2 only At least one of the debtors and another Check if this is community property (see instructions) I list here: Who has an interest in the property? Check one. Debtor 1 only Debtor 2 only Debtor 1 and Debtor 2 only At least one of the debtors and another 	the amount of any secure Creditors Who Have Clair Current value of the entire property? \$ Do not deduct secured cli the amount of any secure Creditors Who Have Clair Current value of the entire property? \$ \$	d claims on Schedule ns Secured by Proper Current value of portion you own \$
Make: Model: Year: Other information: Make: Model: Year: Other information: Other information:	Debtor 1 only Debtor 2 only Debtor 1 and Debtor 2 only At least one of the debtors and another Check if this is community property (see instructions) Iist here: Who has an interest in the property? Check one. Debtor 1 only Debtor 2 only Debtor 2 only Debtor 1 and Debtor 2 only At least one of the debtors and another Check if this is community property (see instructions)	the amount of any secure Creditors Who Have Clair Current value of the entire property? \$ Do not deduct secured cli the amount of any secure Creditors Who Have Clair Current value of the entire property? \$ \$	d claims on Schedule ns Secured by Proper Current value of portion you own \$
Make: Model: Year: Other information: Make: Model: Year: Other information: Other information:	Debtor 1 only Debtor 2 only Debtor 1 and Debtor 2 only At least one of the debtors and another Check if this is community property (see instructions) Iist here: Who has an interest in the property? Check one. Debtor 1 only Debtor 2 only Debtor 2 only Debtor 1 and Debtor 2 only At least one of the debtors and another Check if this is community property (see instructions)	the amount of any secure Creditors Who Have Clair Current value of the entire property? \$ Do not deduct secured cli the amount of any secure Creditors Who Have Clair Current value of the entire property? \$ \$	d claims on Schedule I ns Secured by Propert Current value of portion you own \$

	Debtor 1 First Name Middle Name Last Name	Case number (if known)	
Pa	art 3: Describe Your Personal and Household Items		
D	o you own or have any legal or equitable interest in any of the following items?		Current value of the portion you own? Do not deduct secured claims or exemptions.
6.	Household goods and furnishings		se del secono de la construcción de La construcción de la construcción d
	Examples: Major appliances, furniture, linens, china, kitchenware		
	No		atmosting
	Yes. Describe		\$
7.	Electronics		
	Examples: Televisions and radios; audio, video, stereo, and digital equipment; comp collections; electronic devices including cell phones, cameras, media play	uters, printers, scanners; music yers, games	
	No .		
	Yes. Describe		\$
8.	Collectibles of value Examples: Antiques and figurines; paintings, prints, or other artwork; books, pictures stamp, coin, or baseball card collections; other collections, memorabilia,	, or other art objects; collectibles	
	Ves. Describe		\$
			Ψ
9	 Equipment for sports and hobbies Examples: Sports, photographic, exercise, and other hobby equipment; bicycles, por and kayaks; carpentry tools; musical instruments No 	ol tables, golf clubs, skis; canoes	
	Yes. Describe		\$
10	0. Firearms		
	Examples: Pistols, rifles, shotguns, ammunition, and related equipment		
	Yes. Describe		\$
1	1. Clothes		
	Examples: Everyday clothes, furs, leather coats, designer wear, shoes, accessories		
		-	
	Yes. Describe		\$
12	2. Jewelry		
	Examples: Everyday jewelry, costume jewelry, engagement rings, wedding rings, he gold, silver	virloom jewelry, watches, gems,	
	Ves. Describe		\$
1	3. Non-farm animals <i>Examples:</i> Dogs, cats, birds, horses		
	Yes. Describe		\$
1	4. Any other personal and household items you did not already list, including an	y health aids you did not list	annaé
	□ No		
	Yes. Give specific		\$
	information		
1	5. Add the dollar value of all of your entries from Part 3, including any entries for for Part 3. Write that number here		\$
	for Part 3. Write that number here	7	L

First Name	Middle Name Last Name	Case number (if known)	
4: Describe You	ur Financial Assets		
ou own or have any	legal or equitable interest in	n any of the following?	Current value of the
			portion you own? Do not deduct secured cla
			or exemptions.
ash		 M. M. O. ARABOZO M. CONTRACTOR AND CONTRACTOR INTERACTION AND AND ADDRESS REPORTED IN 2. March 201 (201) 	
	nave in your wallet, in your ho	ome, in a safe deposit box, and on hand when you file your petition	
No			
Yes	1. 1	Cash:	\$
eposits of money ixamples: Checking, s and other si	avings, or other financial according and a second sec	ounts; certificates of deposit; shares in credit unions, brokerage houses, multiple accounts with the same institution, list each.	
No			
Yes		Institution name:	
	17.1. Checking account:		\$
	17.2. Checking account:		\$
	17.3. Savings account:		¢
		· · · · · · · · · · · · · · · · · · ·	¢
	17.4. Savings account:		\$
	17.5. Certificates of deposit:		\$
	17.6. Other financial account	· ·	\$
	17.7. Other financial account	Ľ	\$
	17.8. Other financial account	±	\$
	17.9. Other financial account	to	\$
			Ψ
ando mutual funda	or publicly traded stocks		
xamples: Bond funds,	investment accounts with bro	okerage firms, money market accounts	
No			
	Institution or issuer name:		
Yes			\$
J Yes			
J Yes			. \$
J Yes			. \$. \$
Yes			\$\$
Yes			\$\$
lon-publicly traded s	tock and interests in incor		. \$ \$
lon-publicly traded s n LLC, partnership,	tock and interests in incorr and joint venture	porated and unincorporated businesses, including an interest in	- \$ \$
ion-publicly traded s in LLC, partnership, 】 No	tock and interests in incorr and joint venture Name of entity:	porated and unincorporated businesses, including an interest in % of ownership:	\$
Ion-publicly traded s in LLC, partnership, No Yes. Give specific information about	tock and interests in incorr and joint venture Name of entity:	porated and unincorporated businesses, including an interest in % of ownership: %	\$\$
lon-publicly traded s in LLC, partnership, No Yes. Give specific	tock and interests in incorr and joint venture Name of entity:	porated and unincorporated businesses, including an interest in % of ownership:	\$ \$
Ion-publicly traded s in LLC, partnership, ☐ No ☐ Yes. Give specific information about	tock and interests in incorr and joint venture Name of entity:	porated and unincorporated businesses, including an interest in % of ownership: %	\$\$ \$\$ \$\$ \$\$

vernment and corpo	rate bonds and other negotiable and non-negotiable instruments	
ootiable instruments i	nclude personal checks, cashiers' checks, promissory notes, and money orders.	
n-negotiable instrume	nts are those you cannot transfer to someone by signing or delivering them.	
No	Issuer name:	
Yes. Give specific		
them.		\$
		\$
		\$
tirement or pension	accounts	
amples: Interests in IF	A, ERISA, Keogh, 401(k), 403(b), thrift savings accounts, or other pension or profit-sharing plans	
No		
Yes. List each		
account separately	Type of account: Institution name:	
	401(k) or similar plan:	\$
		\$
	Pension plan:	Ψ
	IRA:	\$
	Retirement account:	\$
	Keogh:	\$
	Additional account:	\$
	Additional account:	\$
ur share of all unused	deposits you have made so that you may continue service or use from a company	
ur share of all unused		
ur share of all unused amples: Agreements npanies, or others	deposits you have made so that you may continue service or use from a company	
ur share of all unused amples: Agreements npanies, or others No	deposits you have made so that you may continue service or use from a company with landlords, prepaid rent, public utilities (electric, gas, water), telecommunications	
ur share of all unused amples: Agreements npanies, or others No	deposits you have made so that you may continue service or use from a company	
ur share of all unused amples: Agreements npanies, or others No	deposits you have made so that you may continue service or use from a company with landlords, prepaid rent, public utilities (electric, gas, water), telecommunications	\$
ur share of all unused amples: Agreements npanies, or others No	deposits you have made so that you may continue service or use from a company with landlords, prepaid rent, public utilities (electric, gas, water), telecommunications Institution name or individual:	\$ \$
ur share of all unused amples: Agreements npanies, or others No	deposits you have made so that you may continue service or use from a company with landlords, prepaid rent, public utilities (electric, gas, water), telecommunications Institution name or individual:	\$ \$ \$
ur share of all unused amples: Agreements npanies, or others No	deposits you have made so that you may continue service or use from a company with landlords, prepaid rent, public utilities (electric, gas, water), telecommunications Institution name or individual: Electric: Gas: Heating oil:	\$ \$ \$
ur share of all unused amples: Agreements npanies, or others No	deposits you have made so that you may continue service or use from a company with landlords, prepaid rent, public utilities (electric, gas, water), telecommunications Institution name or individual: Electric: Gas: Heating oil: Security deposit on rental unit:	\$ \$ \$
ur share of all unused amples: Agreements npanies, or others No	deposits you have made so that you may continue service or use from a company with landlords, prepaid rent, public utilities (electric, gas, water), telecommunications Institution name or individual: Electric: Gąs: Heating oil: Security deposit on rental unit: Prepaid rent:	\$ \$ \$ \$
ur share of all unused amples: Agreements npanies, or others No	deposits you have made so that you may continue service or use from a company with landlords, prepaid rent, public utilities (electric, gas, water), telecommunications Institution name or individual: Electric: Gas: Heating oil: Security deposit on rental unit:	\$\$ \$\$ \$\$\$
ur share of all unused amples: Agreements npanies, or others No	deposits you have made so that you may continue service or use from a company with landlords, prepaid rent, public utilities (electric, gas, water), telecommunications Institution name or individual: Electric: Gas: Heating oil: Security deposit on rental unit: Prepaid rent:	\$ \$ \$ \$ \$ \$
ur share of all unused amples: Agreements npanies, or others No	deposits you have made so that you may continue service or use from a company with landlords, prepaid rent, public utilities (electric, gas, water), telecommunications Institution name or individual: Electric: Gas: Heating oil: Security deposit on rental unit: Prepaid rent: Telephone:	\$\$ \$\$ \$\$ \$\$ \$\$
ur share of all unused amples: Agreements npanies, or others No	deposits you have made so that you may continue service or use from a company with landlords, prepaid rent, public utilities (electric, gas, water), telecommunications Institution name or individual: Electric: Gąs: Heating oil: Security deposit on rental unit: Prepaid rent: Telephone: Water:	\$\$ \$\$ \$\$ \$\$ \$\$
ur share of all unused amples: Agreements npanies, or others No	deposits you have made so that you may continue service or use from a company with landlords, prepaid rent, public utilities (electric, gas, water), telecommunications Institution name or individual: Electric: Gas: Heating oil: Security deposit on rental unit: Prepaid rent: Telephone: Water: Rented furniture:	\$\$ \$\$ \$\$ \$\$ \$\$ \$\$ \$\$
ur share of all unused amples: Agreements npanies, or others No Yes	deposits you have made so that you may continue service or use from a company with landlords, prepaid rent, public utilities (electric, gas, water), telecommunications Institution name or individual: Electric: Gas: Heating oil: Security deposit on rental unit: Telephone: Water: Rented furniture: Other:	\$ \$ \$ \$ \$ \$ \$ \$ \$
ur share of all unused amples: Agreements npanies, or others No Yes	deposits you have made so that you may continue service or use from a company with landlords, prepaid rent, public utilities (electric, gas, water), telecommunications Institution name or individual: Electric: Gas: Heating oil: Security deposit on rental unit: Prepaid rent: Telephone: Water: Rented furniture:	\$\$ \$\$ \$\$ \$\$ \$\$ \$\$ \$\$
ur share of all unused amples: Agreements npanies, or others No Yes	deposits you have made so that you may continue service or use from a company with landlords, prepaid rent, public utilities (electric, gas, water), telecommunications Institution name or individual: Electric: Gas: Heating oil: Security deposit on rental unit: Telephone: Water: Rented furniture: Other:	\$\$ \$\$ \$\$ \$\$ \$\$ \$\$
ur share of all unused amples: Agreements No Yes nuities (A contract fo No	deposits you have made so that you may continue service or use from a company with landlords, prepaid rent, public utilities (electric, gas, water), telecommunications Institution name or individual: Electric: Gas: Heating oil: Security deposit on rental unit: Telephone: Water: Rented furniture: Other:	\$\$ \$\$ \$\$ \$\$ \$\$ \$\$
ur share of all unused amples: Agreements No Yes nuities (A contract fo No	deposits you have made so that you may continue service or use from a company with landlords, prepaid rent, public utilities (electric, gas, water), telecommunications Institution name or individual: Electric: Gąs: Heating oil: Security deposit on rental unit: Prepaid rent: Telephone: Water: Rented furniture: Other: a periodic payment of money to you, either for life or for a number of years)	\$\$ \$\$ \$\$ \$\$ \$\$ \$\$
ur share of all unused amples: Agreements No Yes nuities (A contract fo No	deposits you have made so that you may continue service or use from a company with landlords, prepaid rent, public utilities (electric, gas, water), telecommunications Institution name or individual: Electric: Gąs: Heating oil: Security deposit on rental unit: Prepaid rent: Telephone: Water: Rented furniture: Other: a periodic payment of money to you, either for life or for a number of years)	\$\$ \$\$
amples: Agreements mpanies, or others No Yes	deposits you have made so that you may continue service or use from a company with landlords, prepaid rent, public utilities (electric, gas, water), telecommunications Institution name or individual: Electric: Gąs: Heating oil: Security deposit on rental unit: Prepaid rent: Telephone: Water: Rented furniture: Other: a periodic payment of money to you, either for life or for a number of years)	\$\$ \$\$

Debtor 1 First Name Middle Name	Last Name Case number (if known)	
24. Interests in an education IRA, in an accord 26 U.S.C. §§ 530(b)(1), 529A(b), and 529(punt in a qualified ABLE program, or under a qualified sta	ate tuition program.	
	-7.7		
	name and description. Separately file the records of any inter-	eete 11 11 S C 8 521(c)	h.
institution	lane and description. Separately me the records of any men		
			\$
			\$
			\$
25. Trusts, equitable or future interests in p exercisable for your benefit	roperty (other than anything listed in line 1), and rights o	r powers	
🖵 No			7
Yes. Give specific			\$
information about them			•
26. Patents, copyrights, trademarks, trade s	secrets, and other intellectual property		
	es, proceeds from royalties and licensing agreements		
D No			
Yes. Give specific		-	
information about them			\$
	· · · · · · · · · · · · · · · · · · ·		
27. Licenses, franchises, and other general <i>Examples</i> : Building permits, exclusive licer	nses, cooperative association holdings, liquor licenses, profe	ssional licenses	
□ No			7
Yes. Give specific information about them			S
			↓ Ψ <u>·</u>
Money or property owed to you?			Current value of the
money of property office to you.			portion you own?
			Do not deduct secured claims or exemptions.
28. Tax refunds owed to you			
NoYes. Give specific information			-
about them, including whether	-	Federal:	\$
you already filed the returns and the tax years.	, · ·	State:	\$
and the tax years.		Local:	\$
29. Family support			•
Examples: Past due or lump sum alimony,	spousal support, child support, maintenance, divorce settlen	nent, property settleme	nt
No			
Yes. Give specific information		Alimony:	¢
		Maintenance:	\$ \$
		Support:	\$ \$
		Divorce settlement:	\$ \$
		Property settlement:	\$ \$
		roperty settement.	*
30. Other amounts someone owes you	nce payments, disability benefits, sick pay, vacation pay, wo	rkers' componenties	
Social Security benefits; unpaid	d loans you made to someone else	inters compensation,	
D No.			
Yes. Give specific information.			
			\$
1			

. D-htar				Case number (if known)	
Debtor 1	First Name	Middle Name	Last Name		
			# 4 4 4 7		
	rests in insurance			the base of the second s	
Exa	mples: Health, disa	bility, or life insuran	ce; health savings account (HSA); cre	edit, homeowner's, or renter's insurance	
	No				
	Yes. Name the insu		Company name:	Beneficiary:	Surrender or refund value:
	of each policy	and list its value			
					\$
-					\$
					S
32. Any	interest in prope	rty that is due you	from someone who has died	nalised as an automatic antitlad to reasily	
lf yo	u are the beneficia erty because some	ry of a living trust, e	xpect proceeds from a life insurance	policy, or are currently entitled to receive	
				w daw construction a construction with a construction with the state of the state o	
	res. Give specific	information			\$
			not you have filed a lawsuit or ma	de a demand for payment	
			s, insurance claims, or rights to sue		
	No	,			
		h claim		-	
_			-	·	\$
34. Othe	er contingent and	unliquidated clain	is of every nature, including count	erclaims of the debtor and rights	
	et off claims				
	No				
	Yes. Describe eacl	h claim		-	
		· .		-	\$
					· · ·
35. Any	financial assets	you did not alread	list		
	No				
	Yes. Give specific	information			S .
			-		
		of all of your ontri	s from Part 4, including any entries	e for pages you have attached	
36. Add	Dart 4 Write that	of all of your entrie	s from Fart 4, including any entries	s for pages you have attached	⇒ s
	Fait 4. Wille that				
	-				
	_				
Part 5	Describe	Any Business-	Related Property You Own	or Have an Interest In. List a	ny real estate in Part 1.
37. Do y	ou own or have a	any legal or equita	ole interest in any business-related	I property?	
	No. Go to Part 6.				
	Yes. Go to line 38.				
					Current value of the
					portion you own?
					Do not deduct secured claims
					or exemptions.
38. Acc	ounts receivable	or commissions y	ou already earned		
	No			•	
	Yes. Describe	*****		-	
1					\$
30 0#	ce equipment fu	rnishings, and sup	plies		
				s, rugs, telephones, desks, chairs, electronic de	vices
Fra			·		1
	No				
					\$

Debtor 1		Case nur	mber (if known)	
First Name	Middle Name Last Name			
Machinery, fixtures, e	quipment, supplies you use in bu	usiness, and tools of your trade		
No		-		****
Yes. Describe				\$
Inventory				
No		-		
Yes. Describe				\$
Interests in partnersh	ips or joint ventures			
No No				
Yes. Describe	Name of entity:		% of ownership:	
			%	\$
			%	\$
	-	·	%	\$
Quetemerliste meilin	a lists or other compilations			
No	g lists, or other compilations			
	include personally identifiable in	nformation (as defined in 11 U.S.C. § 10	01(41A)) ?	
D No				
🔲 Yes. Desc	ribe			\$
				φ
Any husiness-related	property you did not already list			
No	property you and not anounly not			
Yes. Give specific				\$
information				\$
				\$\$
		· · · · · · · · · · · · · · · · · · ·		*
· · · ·				\$
				\$
				\$
Add the doilar value	of all of your entries from Part 5.	including any entries for pages you h	ave attached	
for Part 5. Write that	number here		→	\$
art 6: Describe A	ny Farm- and Commercial Fi	shing-Related Property You Own	or Have an Interest I	n
if you own o	r have an interest in farmland, lis	st it in Part 1.	· · ·	
		any form or commercial fiching relate	ad property?	
Do you own or have a No. Go to Part 7.	iny legal of equitable interest in a	any farm- or commercial fishing-relate	sa property:	
Yes. Go to line 47.				
		· ·		Current value of the
				portion you own?
				Do not deduct secured claims or exemptions.
7. Farm animals				and an and a second
Examples: Livestock,	poultry, farm-raised fish			
D No		· · · · ·		
Yes	-			
				\$

Debter 1				Case number (if known)	
Debtor 1	First Name	Middle Name Last Name		· · ·	
Crons oi	ithor growing	or harvested			
	itilei growing				7
	Sive specific				S
	ation				.
9. Farm and No	fishing equi	pment, implements, machinery, fixtur	es, and tools of trade		
Yes			-		
					\$
	fishing supp	lies, chemicals, and feed			
No Ves		-			
					\$
1. Any farm-	and comme	rcial fishing-related property you did	not already list		
D No					-
	Give specific ation.				\$
2 Add the d	sollar value o	f all of your entries from Part 6, inclu	ding any entries for page	es you have attached	
for Part 6	. Write that n	umber here		→	\$
-					
Part 7:	Describe A	All Property You Own or Have	an Interest in Tha	t You Did Not List Above	
	ave other pro	operty of any kind you did not already	/list?		
Examples:	Season tickets,	country club membership			
No No	0		· · ·		\$
	Give specific nation.			-	\$
					\$
		f all of your entries from Part 7. Write	that number boro	-	s
4. Add the d	iollar value o	Tail of your entries from Part 7. write	that number nere		
Part 8:	List the To	otals of Each Part of this For	n		
5. Part 1: To	otal real estat	e, line 2	· · ·	→	\$
56. Part 2: To	otal vehicles,	line 5	\$		
		and household items, line 15	\$		
			*		
58. Part 4: To	otal financial	assets, line 36	۵		
59. Part 5: To	otal business	-related property, line 45	\$	— . · · · ·	1997 - A.
0. Part 6: To	otal farm- and	d fishing-related property, line 52	\$		
51. Part 7: To	otal other pro	operty not listed, line 54	+ \$		
2 Total ner	eonal proper	ty. Add lines 56 through 61.	s	Copy personal property total →	+ s
2. Total per	sonai proper	ty. Add intes of through of the	· • • • • • • • • • • • • • • • • • • •		- 4
		- Cabadula A/D. Add line 55 + line 60			¢
53. Total of a	all property o	n Schedule A/B. Add line 55 + line 62.			\$

Fill in this in	formation to i	dentify your case:	
Debtor 1	First Name	Middle Name	Last Name
Debtor 2 (Spouse, if filing)	First Name	Middle Name	Last Name
United States 8	Bankruptcy Court	for the:	District of (State)
Case number (If known)		· · · · · · · · · · · · · · · · · · ·	

Check if this is an amended filing

Official Form 106C

Schedule C: The Property You Claim as Exempt

12/15

Be as complete and accurate as possible. If two married people are filing together, both are equally responsible for supplying correct information. Using the property you listed on *Schedule A/B: Property* (Official Form 106A/B) as your source, list the property that you claim as exempt. If more space is needed, fill out and attach to this page as many copies of *Part 2: Additional Page* as necessary. On the top of any additional pages, write your name and case number (if known).

For each item of property you claim as exempt, you must specify the amount of the exemption you claim. One way of doing so is to state a specific dollar amount as exempt. Alternatively, you may claim the full fair market value of the property being exempted up to the amount of any applicable statutory limit. Some exemptions—such as those for health aids, rights to receive certain benefits, and tax-exempt retirement funds—may be unlimited in dollar amount. However, if you claim an exemption of 100% of fair market value under a law that limits the exemption to a particular dollar amount and the value of the property is determined to exceed that amount, your exemption would be limited to the applicable statutory amount.

Part 1: Identify the Property You Claim as Exempt

1. Which set of exemptions are you claiming? Check one only, even if your spouse is filing with you.

□ You are claiming state and federal nonbankruptcy exemptions. 11 U.S.C. § 522(b)(3)

☐ You are claiming federal exemptions. 11 U.S.C. § 522(b)(2)

2. For any property you list on Schedule A/B that you claim as exempt, fill in the information below.

	······································	WAY - COME, 15 YO MANDARA & CANADARA & CANAD	
Brief description of the property and line on Schedule A/B that lists this property	Current value of the portion you own	Amount of the exemption you claim	Specific laws that allow exemption
	Copy the value from Schedule A/B	Check only one box for each exemption.	
Brief description:	\$	□ \$ □ 100% of fair market value, up to	
Line from Schedule A/B:		any applicable statutory limit	
Brief description:	\$	□ \$	· · · · · · · · · · · · · · · · · · ·
Line from Schedule A/B:		100% of fair market value, up to any applicable statutory limit	
Brief description:	\$	□ \$	
Line from Schedule A/B:		100% of fair market value, up to any applicable statutory limit	
Are you claiming a homestead exemption o	of more than \$155,675?	•	
(Subject to adjustment on 4/01/16 and every 3	years after that for case	es filed on or after the date of adjustment.)
D No			
Yes. Did you acquire the property covered	by the exemption within	1,215 days before you filed this case?	
No No			-
Yes			

Debtor 1

Part

Middle Name Last Name

Case number (if know

2:	Additional	Page
----	------------	------

First Name

Brief description of the property and line on Schedule A/B that lists this property	Current value of the portion you own	Amount of the exemption you claim	Specific laws that allow exemption
	Copy the value from Schedule A/B	Check only one box for each exemption	
Brief description:	\$	□ \$	· · · · · · · · · · · · · · · · · · ·
Line from		100% of fair market value, up to any applicable statutory limit	
Brief description:	\$	□ \$	
Line from Schedule A/B:		100% of fair market value, up to any applicable statutory limit	· · · · · · · · · · · · · · · · · · ·
Brief description:	\$	s	
Line from	No 1910 a	100% of fair market value, up to any applicable statutory limit	· · · · · · · · · · · · · · · · · · ·
Brief description:	\$	_ _ \$	
Line from Schedule A/B:		100% of fair market value, up to any applicable statutory limit	
Brief description:	\$	\$	
Line from Schedule A/B:		100% of fair market value, up to any applicable statutory limit	
Brief description:	\$	\$	
Line from Schedule A/B:		100% of fair market value, up to any applicable statutory limit	·
Brief description:	\$	s	· · · · · · · · · · · · · · · · · · ·
Line from Schedule A/B:		100% of fair market value, up to any applicable statutory limit	
Brief description:	\$	\$	
Line from Schedule A/B:		100% of fair market value, up to any applicable statutory limit	
Brief description:	\$	\$	
Line from Schedule A/B:		100% of fair market value, up to any applicable statutory limit	
Brief description:	\$	_ 🖸 \$	
Line from Schedule A/B:		100% of fair market value, up to any applicable statutory limit	
Brief description:	\$	_ 🛛 \$	
Line from Schedule A/B:		100% of fair market value, up to any applicable statutory limit	
Brief description:	\$	_ 🗅 \$	
Line from Schedule A/B:		100% of fair market value, up to any applicable statutory limit	

Fill in this information to identify your case:						
Debtor 1	First Name	Middle Name	Last Name			
Debtor 2 (Spouse, if filing)	First Name	Middle Name	Last Name			
United States E	Bankruptcy Court for the:		District of (State)			
Case number (If known)						

Check if this is an amended filing

12/15

Official Form 106D

Schedule D: Creditors Who Have Claims Secured by Property

Be as complete and accurate as possible. If two married people are filing together, both are equally responsible for supplying correct information. If more space is needed, copy the Additional Page, fill it out, number the entries, and attach it to this form. On the top of any additional pages, write your name and case number (if known).

1 Do any creditors have claims secured by your property?

□ No. Check this box and submit this form to the court with your other schedules. You have nothing else to report on this form.

Yes. Fill in all of the information below.

Part 1: List All Secured Claims

2	for each claim. If more than one creditor h	nore than one secured claim, list the creditor separately as a particular claim, list the other creditors in Part 2. habetical order according to the creditor's name.	Column A Amount of claim Do not deduct the value of collateral.	Column B Value of collateral that supports this claim	Column C Unsecured portion If any
2	.1	Describe the property that secures the claim:	\$	\$	\$
	Creditor's Name Number Street City State ZIP Code Who owes the debt? Check one. Debtor 1 only Debtor 2 only	As of the date you file, the claim is: Check all that apply. Contingent Unliquidated Disputed Nature of lien. Check all that apply. An agreement you made (such as mortgage or secured car loan)			
	 Debtor 1 and Debtor 2 only At least one of the debtors and another Check if this claim relates to a community debt 	 Statutory lien (such as tax lien, mechanic's lien) Judgment lien from a lawsuit Other (including a right to offset)	- -		
	Date debt was incurred	Last 4 digits of account number			
	2.2	Describe the property that secures the claim:	\$	\$	\$
	Creditor's Name				
		As of the date you file, the claim is: Check all that apply.			
	City State ZIP Code	Unliquidated Disputed			
	City State ZIP Code Who owes the debt? Check one.				
		Disputed	-		
	Who owes the debt? Check one. Debtor 1 only Debtor 2 only Debtor 1 and Debtor 2 only At least one of the debtors and another Check if this claim relates to a community debt Date debt was incurred	 Disputed Nature of lien. Check all that apply. An agreement you made (such as mortgage or secured car loan) Statutory lien (such as tax lien, mechanic's lien) Judgment lien from a lawsuit 	_		

Schedule D: Creditors Who Have Claims Secured by Property

First Name Middle Name	Last Name	-		
Additional Page rt 1: After listing any entries on this p by 2.4, and so forth.	page, number them beginning with 2.3, followed	Column A Amount of claim Do not deduct the value of collateral.	Column B Value of collateral that supports this claim	Column C Unsecured portion If any
	Describe the property that secures the claim:	\$	\$	\$
Creditor's Name				
Number Street				
	As of the date you file, the claim is: Check all that apply.	1		
City State ZIP Code				
the sume the debte of the state				
/ho owes the debt? Check one.	Nature of lien. Check all that apply.			
Debtor 1 only Debtor 2 only	An agreement you made (such as mongage or secured car loan)			
Debtor 1 and Debtor 2 only	Statutory lien (such as tax lien, mechanic's lien)			
At least one of the debtors and another	Judgment lien from a lawsuit Other (including a right to offset)			
Check if this claim relates to a		-		
community debt				
ate debt was incurred	Last 4 digits of account number			
	Describe the property that secures the claim:	\$	\$\$	\$
Creditor's Name			· · · ·	
Number Street				
	As of the date you file, the claim is: Check all that apply.			
City State ZIP Code	Unliquidated Disputed			
Vho owes the debt? Check one.	Nature of lien. Check all that apply.			
Debtor 1 only	An agreement you made (such as mortgage or secured			
Debtor 2 only	car loan)			
 Debtor 1 and Debtor 2 only At least one of the debtors and another 	 Statutory lien (such as tax lien, mechanic's lien) Judgment lien from a lawsuit 			
	Other (including a right to offset)			
Check if this claim relates to a community debt				
Date debt was incurred	Last 4 digits of account number			
n para manan dan kari kari ang kari ang	Describe the property that secures the claim:	\$	\$	\$
Creditor's Name		1		-
Number Street				
	As of the date you file, the claim is: Check all that apply.			
City State ZIP Code				
	Disputed			
Vho owes the debt? Check one.	Nature of lien. Check all that apply.			
Debtor 1 only	An agreement you made (such as mortgage or secured			
Debtor 2 only Debtor 1 and Debtor 2 only	car loan) Statutory lien (such as tax lien, mechanic's lien)			
At least one of the debtors and another	Judgment lien from a lawsuit			
Check if this claim relates to a community debt	Cther (including a right to offset)			
Date debt was incurred	Last 4 digits of account number	•		
Add the dollar value of your entrie	s in Column A on this page. Write that number here:	s		
바다. 이 안 같아? 승규가 가지 않는 것 같은 것 같아요. 가지 않는 것	, add the dollar value totals from all pages.	() *		

Debtor 1	Case number (if known)
First Name Middle Name Last Name	Janady I Infod
Part 2: List Others to Be Notified for a Debt That You A	bitcy for a debt that you already listed in Part 1. For example, if a collection
agency is trying to collect from you for a debt you owe to someone else	e, list the creditor in Part 1, and then list the collection agency here. Similarly, if Part 1, list the additional creditors here. If you do not have additional persons to
	On which line in Part 1 did you enter the creditor?
Name	Last 4 digits of account number
Number Street	
City State ZIP Code	3
	On which line in Part 1 did you enter the creditor?
Name	Last 4 digits of account number
Number Street	
· · · · · · · · · · · · · · · · · · ·	
City State ZIP Code	
	On which line in Part 1 did you enter the creditor?
Name	Last 4 digits of account number
Number Street	
City State ZIP Code	
	On which line in Part 1 did you enter the creditor?
Name	Last 4 digits of account number
Number Street	·
City State ZIP Code	
	On which line in Part 1 did you enter the creditor?
Name	Last 4 digits of account number
Number Street	
	,
City State ZIP Code	
Name	On which line in Part 1 did you enter the creditor? Last 4 digits of account number
Number Street	
City State ZIP Code	e e e e e e e e e e e e e e e e e e e

page ____ of ____

	Fill in this in	formation to identify	your case:	
ì	Debtor 1		Middle Marco	Last Name
	Debtor 2	First Name	Middle Name	Last Name
	(Spouse, if filing)	First Name	Middle Name	Last Name
	United States E	Bankruptcy Court for the:		District of(State)
	Case number (If known)	<u> </u>		

Check if this is an amended filing

Official Form 106E/F

Schedule E/F: Creditors Who Have Unsecured Claims

12/15

Be as complete and accurate as possible. Use Part 1 for creditors with PRIORITY claims and Part 2 for creditors with NONPRIORITY claims. List the other party to any executory contracts or unexpired leases that could result in a claim. Also list executory contracts on *Schedule A/B: Property* (Official Form 106A/B) and on *Schedule G: Executory Contracts and Unexpired Leases* (Official Form 106G). Do not include any creditors with partially secured claims that are listed in *Schedule D: Creditors Who Have Claims Secured by Property*. If more space is needed, copy the Part you need, fill it out, number the entries in the boxes on the left. Attach the Continuation Page to this page. On the top of any additional pages, write your name and case number (if known).

_	art 1: List All of Your PRIORITY Unsecure				-
1.	Do any creditors have priority unsecured claims No. Go to Part 2. Yes.	against you?			
2.	each claim listed, identify what type of claim it is. If nonpriority amounts. As much as possible, list the c unsecured claims, fill out the Continuation Page of	editor has more than one priority unsecured claim, list the a claim has both priority and nonpriority amounts, list the laims in alphabetical order according to the creditor's ne Part 1. If more than one creditor holds a particular claim the first form in the insteading backlot)	at claim here ar ame. If you hav	nd show both e more than tw	priority and vo priority
	(For an explanation of each type of claim, see the in	istructions for this form in the instruction bookiet.)	Total claim	Priority amount	Nonpriority amount
2.1	Priority Creditor's Name	Last 4 digits of account number	\$	\$	\$\$
	Number Street	When was the debt incurred?			
		As of the date you file, the claim is: Check all that apply	1.		
	City State ZIP Code Who incurred the debt? Check one.	 Contingent Unliquidated Disputed 			~
	Debtor 2 only	Type of PRIORITY unsecured claim:			
	 Debtor 1 and Debtor 2 only At least one of the debtors and another 	 Domestic support obligations Taxes and certain other debts you owe the government 			
	Check if this claim is for a community debt	Claims for death or personal injury while you were intoxicated			
	Is the claim subject to offset? No Yes	Other. Specify			
2.2		Last 4 digits of account number	\$	\$	\$
	Priority Creditor's Name	When was the debt incurred?			
	Number Street	As of the date you file, the claim is: Check all that apply	1.		
	City State ZIP Code	Contingent Unliquidated			
	Who incurred the debt? Check one.				
	Debtor 2 only	Type of PRIORITY unsecured claim: Domestic support obligations			
	 Debtor 1 and Debtor 2 only At least one of the debtors and another 	Taxes and certain other debts you owe the government			
	Check if this claim is for a community debt	Claims for death or personal injury while you were intoxicated			
	Is the claim subject to offset?	Other. Specify	-		

	tor 1 First Name Middle Name Last Name	Case number (# known)			<u> </u>
	rt 1: Your PRIORITY Unsecured Claims er listing any entries on this page, number them	beginning with 2.3, followed by 2.4, and so forth.	Total claim	Priority amount	Nonpriority amount
				amount	amount
		Last & disite of account number	s	\$	s
	Priority Creditor's Name	Last 4 digits of account number	Ψ	Ψ	- ¥
		When was the debt incurred?			
	Number Street				
		As of the date you file, the claim is: Check all that apply.			
		Contingent			
	City State ZIP Code				
	Who incurred the debt? Check one.	Disputed			
	Debtor 1 only	Type of PRIORITY unsecured claim:			
	Debtor 2 only				
	Debtor 1 and Debtor 2 only	 Domestic support obligations Taxes and certain other debts you owe the government 			
۰.	At least one of the debtors and another	 Taxes and certain other debts you owe the government Claims for death or personal injury while you were 			
	Check if this claim is for a community debt	intoxicated			
		Other. Specify			
	Is the claim subject to offset?				
	□ No				
	C Yes				
-					
_	I	Last 4 digits of account number	\$	\$	\$
	Phority Creditor's Name				
	Number Street	When was the debt incurred?			
	Number Street	As of the date you file, the claim is: Check all that apply.			
		As of the date you me, the claim is. Check an that apply.			
	City State ZIP Code				
	Who incurred the debt? Check one.	Disputed			
	Debtor 1 only	Type of PRIORITY unsecured claim:			
	Debtor 2 only				
	Debtor 1 and Debtor 2 only	Domestic support obligations			
	At least one of the debtors and another	 Taxes and certain other debts you owe the government Claims for death or personal injury while you were 			
	Check if this claim is for a community debt	intoxicated			
		Other. Specify			
	Is the claim subject to offset?	and the second			
	No No				
	Yes				

	Priority Creditor's Name	Last 4 digits of account number	\$	\$	\$
	Phoney oreallor smaller	When was the debt incurred?			
	Number Street				
		As of the date you file, the claim is: Check all that apply.			
	City State ZIP Code				
	Who incurred the debt? Check one.				
	Debtor 1 only	Type of PRIORITY unsecured claim:			
	Debtor 2 only	Domestic support obligations			
	Debtor 1 and Debtor 2 only	Taxes and certain other debts you owe the government			
	At least one of the debtors and another	Claims for death or personal injury while you were			
	Check if this claim is for a community debt	intoxicated	1997		
		Other. Specify			
	Is the claim subject to offset?				
	D No				
	Yes				

Debto		· .	Case number (if known)	
	First Name Middle Name Last Name			
Par	t 2: List All of Your NONPRIORITY Unsecure	ed Claims		
	Do any creditors have nonpriority unsecured claims No. You have nothing to report in this part. Submit th Yes		ourt with your other schedules.	· · · · · · · · · · · · · · · · · · ·
n ir	contrionity unsecured claim, list the creditor separately for	or each claim.	der of the creditor who holds each claim. If a creditor has for each claim listed, identify what type of claim it is. Do not the other creditors in Part 3.If you have more than three nor	list claims already
			Eller C. Status and C. Martin Miller Global systems in the second system of the sys	Total claim
4.1	No contracts Opentitede Name		Last 4 digits of account number	S
	Nonpriority Creditor's Name		When was the debt incurred?	
	Number Street			~
	City State ZIP C	Code	As of the date you file, the claim is: Check all that apply.	
	Who incurred the debt? Check one.		Contingent Unliquidated	
and who shall	Debtor 1 only		Disputed	
10-10-10 (Friday)	Debtor 2 only			
	Debtor 1 and Debtor 2 only		Type of NONPRIORITY unsecured claim:	
	At least one of the debtors and another		Student loans	
	Check if this claim is for a community debt		Obligations arising out of a separation agreement or divorce that you did not report as priority claims	
	Is the claim subject to offset?		Debts to pension or profit-sharing plans, and other similar debts	
	☐ No ☐ Yes		Other. Specify	
4.2			Last 4 digits of account number	\$
	Nonpriority Creditor's Name		When was the debt incurred?	
	Number Street		As of the date you file, the claim is: Check all that apply.	
*	City State ZIP C	Code		
	Who incurred the debt? Check one.		Unliquidated	
	Debtor 1 only		Disputed	
-	Debtor 2 only		Type of NONPRIORITY unsecured claim:	
-	Debtor 1 and Debtor 2 only			
	At least one of the debtors and another		 Student loans Obligations arising out of a separation agreement or divorce 	
	Check if this claim is for a community debt		that you did not report as priority claims	
	Is the claim subject to offset?		Debts to pension or profit-sharing plans, and other similar debts	
	No		Other. Specify	
	Yes			
4.3		,	Last 4 digits of account number	
	Nonpriority Creditor's Name		When was the debt incurred?	\$
-	Number Street			
	Citv State ZIP C	Code	As of the date you file, the claim is: Check all that apply.	
1. Par				
-	Who incurred the debt? Check one.		Unliquidated	
	Debtor 1 only Debtor 2 only		Disputed	
	Debtor 2 only Debtor 1 and Debtor 2 only			
	At least one of the debtors and another		Type of NONPRIORITY unsecured claim:	
			Student loans	
	Check if this claim is for a community debt		Obligations arising out of a separation agreement or divorce that you did not report as priority claims	
	Is the claim subject to offset?		Debts to pension or profit-sharing plans, and other similar debts	
100000	Ves		Other. Specify	

0	h	to	-

Middle Name

Last Name

First Name

Case number (if known)_

isting any entries on this page, number them beginning with 4.4		Total cla
	Last 4 digits of account number	\$
onpriority Creditor's Name		Ψ
	When was the debt incurred?	
umber Street	As of the date you file, the claim is: Check all that apply.	
ty State ZIP Code	Contingent	
ho incurred the debt? Check one.	Disputed	
Debtor 1 only		
Debtor 2 only	Type of NONPRIORITY unsecured claim:	
Debtor 1 and Debtor 2 only	Student loans	
At least one of the debtors and another	Obligations arising out of a separation agreement or divorce that	
Check if this claim is for a community debt	you did not report as priority claims	
- · · · · · · · ·	 Debts to pension or profit-sharing plans, and other similar debts Other. Specify 	
the claim subject to offset?		
No		
] Yes		
	ana gapané ili minimukana kana kanggara - ingo gan ing nakarikan yang kang nakara di mana kang hang pana - ing	
	Last 4 digits of account number	\$
onpriority Creditor's Name	When was the debt incurred?	
umber Street	As of the date you file, the claim is: Check all that apply.	
ty State ZIP Code		
	Unliquidated	
/ho incurred the debt? Check one.	Disputed	
Debtor 1 only		
Debtor 2 only	Type of NONPRIORITY unsecured claim:	
Debtor 1 and Debtor 2 only	Student loans	
At least one of the debtors and another	Obligations arising out of a separation agreement or divorce that	
Check if this claim is for a community debt	you did not report as priority claims	
	Debts to pension or profit-sharing plans, and other similar debts	
the claim subject to offset?	Other. Specify	
No		
Yes		
e de la sector com a devenu de sue trappe a la proprie manage de parte a se contra presente sur se baixante apresente de la minar esta entra presente en	Last 4 digits of account number	\$
onpriority Creditor's Name		
	When was the debt incurred?	
umber Street	As of the date you file, the claim is: Check all that apply.	
ity State ZIP Code	Contingent	
Vho incurred the debt? Check one.	Disputed	
Debtor 1 only		
Debtor 2 only	Type of NONPRIORITY unsecured claim:	
Debtor 1 and Debtor 2 only	Student loans	
At least one of the debtors and another	Obligations arising out of a separation agreement or divorce that	
Check if this claim is for a community debt	you did not report as priority claims	
	Debts to pension or profit-sharing plans, and other similar debts	
s the claim subject to offset?	Other. Specify	

	to		

Case number (if known)

Part 3: List Others to Be Notified About a Debt That You Already Listed

5. Use this page only if you have others to be notified about your bankruptcy, for a debt that you already listed in Parts 1 or 2. For example, if a collection agency is trying to collect from you for a debt you owe to someone else, list the original creditor in Parts 1 or 2, then list the collection agency here. Similarly, if you have more than one creditor for any of the debts that you listed in Parts 1 or 2, list the additional creditors here. If you do not have additional persons to be notified for any debts in Parts 1 or 2, do not fill out or submit this page.

		On which entry in Part 1 or Part 2 did you list the original creditor?
Name		Line of (Check one): D Part 1: Creditors with Priority Unsecured Claims
Number Street		Part 2: Creditors with Phony Unsecured Claims Part 2: Creditors with Nonpriority Unsecured Claims
		Last 4 digits of account number
City	State ZIP Code	
Name		On which entry in Part 1 or Part 2 did you list the original creditor?
Name		Line of (Check one):
Number Street		Part 2: Creditors with Nonpriority Unsecured
		Claims
	21-1-1-21D Orde	Last 4 digits of account number
City	State ZIP Code	
Nomo		On which entry in Part 1 or Part 2 did you list the original creditor?
Name		Line of (Check one): Deart 1: Creditors with Priority Unsecured Claims
Number Street		Part 2: Creditors with Nonpriority Unsecured
		Claims
		Last 4 digits of account number
City	State ZIP Code	
Name		On which entry in Part 1 or Part 2 did you list the original creditor?
		Line of (Check one): D Part 1: Creditors with Priority Unsecured Claims
Number Street		Part 2: Creditors with Nonpriority Unsecured
		Claims
<u></u>	State ZIP Code	Last 4 digits of account number
City		
Name		On which entry in Part 1 or Part 2 did you list the original creditor?
		Line of (Check one): D Part 1: Creditors with Priority Unsecured Claims
Number Street		Part 2: Creditors with Nonpriority Unsecured
		Claims
<u>City</u>	State ZIP Code	Last 4 digits of account number
City .	State Zir Gude	On which entry in Part 1 or Part 2 did you list the original creditor?
Name		on which entry in rare roor are 2 did you not the original creditor?
		Line of (Check one): D Part 1: Creditors with Priority Unsecured Claims
Number Street		Claims
		Ciains
City	State ZIP Code	Last 4 digits of account number
Name		On which entry in Part 1 or Part 2 did you list the original creditor?
		Line of (Check one): D Part 1: Creditors with Priority Unsecured Claims
Number Street		Part 2: Creditors with Nonpriority Unsecured
		Claims
City	State ZIP Code	Last 4 digits of account number
Ulty .		

	bi		

First Name

Case number (if known)

Part 4: Add the Amounts for Each Type of Unsecured Claim

Last Name

6. Total the amounts of certain types of unsecured claims. This information is for statistical reporting purposes only. 28 U.S.C. § 159. Add the amounts for each type of unsecured claim.

			Total claim
otal claims	6a. Domestic support obligations	6a.	\$
rom Part 1	6b. Taxes and certain other debts you owe the government	6b.	\$
	6c. Claims for death or personal injury while you were intoxicated	6c.	\$
	6d. Other. Add all other priority unsecured claims. Write that amount here.	6d	+ \$
	6e. Total. Add lines 6a through 6d.	6e.	\$
			Total claim
Fotal claims	6f. Student loans	6f.	\$
rom Part 2	6g. Obligations arising out of a separation agreement or divorce that you did not report as priority claims	6g.	\$
	6h. Debts to pension or profit-sharing plans, and other similar debts	6h.	\$
	 Other. Add all other nonpriority unsecured claims. Write that amount here. 	6i.	+ \$
	6j. Total. Add lines 6f through 6i.	6j.	· · · · · · · · · · · · · · · · · · ·

Fill in this information to identify your case:									
Debtor									
	First Name	Middle Name	Last Name						
Debtor 2			Last Name						
(Spouse if filing)	First Name	Middle Name	Last Name						
United States F	Bankruptcy Court for t	the:	District of						
onition official of			(State)						
Case number									
(If known)									

Official Form 106G

Schedule G: Executory Contracts and Unexpired Leases

12/15

Be as complete and accurate as possible. If two married people are filing together, both are equally responsible for supplying correct information. If more space is needed, copy the additional page, fill it out, number the entries, and attach it to this page. On the top of any additional pages, write your name and case number (if known).

1. Do you have any executory contracts or unexpired leases?

No. Check this box and file this form with the court with your other schedules. You have nothing else to report on this form.

Yes. Fill in all of the information below even if the contracts or leases are listed on Schedule A/B: Property (Official Form 106A/B).

2. List separately each person or company with whom you have the contract or lease. Then state what each contract or lease is for (for example, rent, vehicle lease, cell phone). See the instructions for this form in the instruction booklet for more examples of executory contracts and unexpired leases.

Person o	r company with wi	hom you	have the con	itract or lease	State what the contract or lease is for
	100000000				
Name					
Number	Street			· · · · · · · · · · · · · · · · · · ·	
-				· · · · ·	
City		State	ZIP Code		
				-	
Name					
Number	Street				
City		State	ZIP Code		
			-		
Name					
Number	Street				
City		State	ZIP Code	· · · · · · · · · · · · · · · · · · ·	
	-				
Name					
Number	Street			· · ·	
City	~	State	ZIP Code		
Name		-			
Number	Street				
City		State	ZIP Code		

otor	1					Case number (if known)
	Fir	st Name Mid	dle Name	Last Name		
	Ad	Iditional Pag	ge if You Ha	we More Contracts or Leas	ses	
				have the contract or lease		What the contract or lease is for
	Person or	company wit	n whom you	lave the contract of lease		
					· .	
	Name					
,	Number	Street			,;	
	City		State	ZIP Code		
	City		State			
-						
	Name					
	Number	Street			· ,	
	City		State	ZIP Code		
j		-				
	Name					
	Number	Street			-	
	City		State	ZIP Code		
			96469117020220220220777 8797 070791783888888888		electrosoft Australian	
-						
	Name					
	Number	Street				
	City		State	ZIP Code		
	a a second de la companya de la comp			· · · · · Balanters	Marten	
and						
	Name					
	Number	Street				
	City		State	ZIP Code		
	· · · · · · · · · · · · · · · · · · ·	************************************	- AND AND AND A CONTRACT	, and the second s	inducera içina	
	Nama	-				
	Name					
	Number	Street				
	City		State	ZIP Code	<u> </u>	
	and water ou a					
	Name					
	Name					
	Number	Street		5. S.		
	City		State	ZIP Code		
ľ						
	Name			· · · · · · · · · · · · · · · · · · ·		
	Naille					
	Number	Street				
	City		State	ZIP Code		

Fill in this information to identify your case:									
Debtor 1	First Name	Middle Name	Last Name						
Debtor 2 (Spouse, if filing)	First Name	Middle Name	Last Name						
United States E	of								
Case number (If known)									

Official Form 106H

ic	hedule H: Your Codebtors	12/15
re fil nd n	btors are people or entities who are also liable for any debts you may hing together, both are equally responsible for supplying correct inform umber the entries in the boxes on the left. Attach the Additional Page t number (if known). Answer every question.	nation. If more space is needed, copy the Additional Page, fill it out,
	Do you have any codebtors? (If you are filing a joint case, do not list either No	spouse as a codebtor.)
	Yes	
. \	Nithin the last 8 years, have you lived in a community property state or Arizona, California, Idaho, Louisiana, Nevada, New Mexico, Puerto Rico, Te	territory? (Community property states and territories include xas, Washington, and Wisconsin.)
(No. Go to line 3.	
(Yes. Did your spouse, former spouse, or legal equivalent live with you at	the time?
	No	
	Yes. In which community state or territory did you live?	Fill in the name and current address of that person.
	Name of your spouse, former spouse, or legal equivalent	
	Number Street	
	City State ZIP	Code
3. I	n Column 1, list all of your codebtors. Do not include your spouse as a	codebtor if your spouse is filing with you. List the person
1	shown in line 2 again as a codebtor only if that person is a guarantor o	r cosigner. Make sure you have listed the creditor on
	Schedule D (Official Form 106D), Schedule E/F (Official Form 106E/F), o Schedule E/F, or Schedule G to fill out Column 2.	or Schedule G (Official Form 106G). Use Schedule D,
100		Optimum Of The secolitization theory was such the data
12 20 10 1	Column 1: Your codebtor	Column 2: The creditor to whom you owe the debt
		Check all schedules that apply:
1		Schedule D, line
	Name	Schedule E/F, line
	Number Street	Schedule G, line
	City State Zi	P Code
2		
	Name	Schedule D, line
	Nation Oracle	Schedule E/F, line Schedule G, line
	Number Street	
	City State Zi	P Code
3		Schedule D, line
	Name	Schedule E/F, line
	Number Street	Schedule G, line
	Number Street	

City

State

ZIP Code

Debtor	r 1					Case number (if known)
Debioi		First Name	Middle Name	Last Name		
	A	dditional	Page to List	More Codebtors		
	Column [·]	1: Your code	obtor			Column 2: The creditor to whom you owe the debt
						Check all schedules that apply:
3	Name					_ Schedule D, line
	Numo					Schedule E/F, line
	Number	Street				Schedule G, line
	City			State	ZIP Code	-
3	Only					
	Name					_ Chedule D, line
						Schedule E/F, line
	Number	Street				- Schedule G, line
	City			State	ZIP Code	-
3						- Green Schedule D, line
	Name					Schedule E/F, line
	Marthan	Olizat				Schedule G, line
	Number	Street				
	City			State	ZIP Code	_
3						_ Schedule D, line
μ	Name					Schedule E/F, line
						Schedule G, line
	Number	Street				
	City			State	ZIP Code	
<u>з</u>						
\square	Name			- 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1		Schedule D, line
						Schedule E/F, line
	Number	Street				Schedule G, line
	City			State	ZIP Code	_
3	0.0					
<u> </u>	Name					C Schedule D, line
						Schedule E/F, line
	Number	Street				Schedule G, line
	City			State	ZIP Code	-
3.	5.1					
<u>[-]</u>	Name					_ Green Schedule D, line
						Schedule E/F, line
	Number	Street				Schedule G, line
	City			State	ZIP Code	
3	City					
	Name			· · · · · · · · · · · · · · · · ·		Schedule D, line
						Schedule E/F, line
	Number	Street			-	Schedule G, line
				01414	7ID Code	_
NUMBER OF TAXABLE	City			State	ZIP Code	

Fill in this in	formation to identify ye	our case:				
Debtor 1	First Name	Middle Name	Last Name			
Debtor 2 (Spouse, if filing)	First Name	Middle Name	Last Name			
United States E	Bankruptcy Court for the:		District of	(State)	- '	
Case number						Check if this is:
(If known)						An amended filing
-				-		A supplement showing postpetition chapter 13 income as of the following date:
Official Fo	orm 1061					MM / DD / YYYY
Sched	ule I: You	r Income	•,			12/15

Be as complete and accurate as possible. If two married people are filing together (Debtor 1 and Debtor 2), both are equally responsible for supplying correct information. If you are married and not filing jointly, and your spouse is living with you, include information about your spouse. If you are separated and your spouse is not filing with you, do not include information about your spouse. If more space is needed, attach a separate sheet to this form. On the top of any additional pages, write your name and case number (if known). Answer every question.

Part 1: Describe Employment

 Fill in your employment information. 		Debtor 1			Debtor 2 or non-filing	j spouse
If you have more than one job, attach a separate page with information about additional employers.	Employment status	 Employed Not employed 	d		EmployedNot employed	
Include part-time, seasonal, or self-employed work.						
Occupation may include student or homemaker, if it applies.	Occupation					
	Employer's name					
	Employer's address					
	minprojet e autose	Number Street			Number Street	
	. *					
		City	State	ZIP Code	City St	ate ZIP Code
		Oity	otato	211 0000		
	How long employed the	re?				
Part 2: Give Details About	t Monthly Income					
Estimate monthly income as of spouse unless you are separated		n. If you have nothin	ng to re	oort for any line, wr	ite \$0 in the space. Include	your non-filing
If you or your non-filing spouse h below. If you need more space, a	ave more than one employe	er, combine the infor iis form.	mation	for all employers for	or that person on the lines	
				For Debtor 1	For Debtor 2 or non-filing spouse	
2. List monthly gross wages, sal deductions). If not paid monthly	lary, and commissions (be , calculate what the monthly	efore all payroll wage would be.	2.	\$	\$	
3. Estimate and list monthly ove	rtime pay.		3. +	\$	+ \$	
4. Calculate gross income. Add I	ine 2 + line 3.	· · ·	4.	\$	\$	

Debtor 1		Case number (if kno	own)	
First Name Middle Name Last Name				
		For Debtor 1	For Debtor 2 or non-filing spouse	
	→ 4.	\$	¢.	
Copy line 4 here	4.	φ	Φ	
5. List all payroll deductions:				-
5a. Tax, Medicare, and Social Security deductions	5a.	\$	\$	
5b. Mandatory contributions for retirement plans	5b.	\$	\$	
5c. Voluntary contributions for retirement plans	5c.	\$	\$	
5d. Required repayments of retirement fund loans	5d.	\$	\$	
5e. Insurance	5e.	\$	\$	
5f. Domestic support obligations	5f.	\$	\$	
5g. Union dues	5g.	\$	\$	
5h. Other deductions. Specify:	5h.	+ \$	+ \$	
 6. Add the payroll deductions. Add lines 5a + 5b + 5c + 5d + 5e +5f + 5g + 5h 	n. 6.	٩	\$	
	. 0.	φ	Ψ	
7. Calculate total monthly take-home pay. Subtract line 6 from line 4.	7.	\$	\$	
 8. List all other income regularly received: 8a. Net income from rental property and from operating a business, 				
profession, or farm				
Attach a statement for each property and business showing gross receipts, ordinary and necessary business expenses, and the total monthly net income.	8a.	\$	\$	
8b. Interest and dividends	8b.	\$	\$	
8c. Family support payments that you, a non-filing spouse, or a depend	dent			
regularly receive				
Include alimony, spousal support, child support, maintenance, divorce settlement, and property settlement.	8c.	\$	\$	
8d. Unemployment compensation	8d.	\$	\$	
8e. Social Security	8e.	\$	\$	
8f. Other government assistance that you regularly receive Include cash assistance and the value (if known) of any non-cash assista that you receive, such as food stamps (benefits under the Supplemental Nutrition Assistance Program) or housing subsidies.	ance	\$	\$	
Specify:	_ 0/.	·		-
8g. Pension or retirement income	8g.	\$	\$	
8h. Other monthly income. Specify:	8h.	+\$	+\$	-
9. Add all other income. Add lines 8a + 8b + 8c + 8d + 8e + 8f +8g + 8h.	9.	\$	\$	
 Calculate monthly income. Add line 7 + line 9. Add the entries in line 10 for Debtor 1 and Debtor 2 or non-filing spouse. 	10.	\$	+ \$	= \$
11. State all other regular contributions to the expenses that you list in Sch	edule J			
Include contributions from an unmarried partner, members of your household friends or relatives.				
Do not include any amounts already included in lines 2-10 or amounts that an				-
Specify:				.+ \$
12. Add the amount in the last column of line 10 to the amount in line 11. The Write that amount on the Summary of Your Assets and Liabilities and Certain				\$
				Combined monthly income
13. Do you expect an increase or decrease within the year after you file this	s form	?		
Yes. Explain:		• •		
				· · · · · · · · · · · · · · · · · · ·

Fill in this information to identify your case:	
Debtor 1 First Name Middle Name Last Name Debtor 2 (Spouse, if filing) First Name Middle Name Last Name United States Bankruptcy Court for the:	Check if this is: An amended filing A supplement showing postpetition chapter 13 expenses as of the following date: MM / DD / YYYY
Official Form 106J	*

Schedule J: Your Expenses

12/15

Be as complete and accurate as possible. If two married people are filing together, both are equally responsible for supplying correct information. If more space is needed, attach another sheet to this form. On the top of any additional pages, write your name and case number (if known). Answer every question.

l	Part 1: Describe Your Hou	sehold	·		
1.	Is this a joint case? No. Go to line 2. Yes. Does Debtor 2 live in a s No		an anti-Maria kald of Daktor 2		
2	Do you have dependents? Do not list Debtor 1 and Debtor 2. Do not state the dependents' names.	 e Official Form 106J-2, <i>Expenses for S</i> No Yes. Fill out this information for each dependent 	Dependent's relationship to Debtor 1 or Debtor 2	Dependent's age	Does dependent live with you? No Yes No Yes No Yes No Yes No Yes
3	Do your expenses include				No Yes
	expenses of people other than yourself and your dependents?	No Yes Monthly Expenses			
	expenses as of a date after the bar applicable date. nclude expenses paid for with nor	r bankruptcy filing date unless you a nkruptcy is filed. If this is a suppleme n-cash government assistance if you d it on <i>Schedule I: Your Income</i> (Offi	ental <i>Schedule J</i> , check the box		n and fill in the
- 1		expenses for your residence. Include		4. \$ 4a. \$	E 7778
	4b. Property, homeowner's, or r	renter's insurance		4b. \$	

4d. Homeowner's association or condominium dues

Schedule J: Your Expenses

4c.

4d.

\$

Del	btor 1 Case	number (if known)	
	First Name Last Name		
			Your expenses
-	Additional mortgage payments for your residence, such as home equity loans	5.	\$
э.	Autilional mortgage payments for your residence, such as nome equity isone		
6.	Utilities:		
	6a. Electricity, heat, natural gas	6a.	\$
	6b. Water, sewer, garbage collection	6b.	\$
	6c. Telephone, cell phone, Internet, satellite, and cable services	6c.	\$
	6d. Other. Specify:	6d.	\$
7.	Food and housekeeping supplies	7.	\$
8.	Childcare and children's education costs	8.	\$
9.	Clothing, laundry, and dry cleaning	9.	\$
10.	Personal care products and services	10.	\$
11.	Medical and dental expenses	. 11.	\$
12.	Transportation. Include gas, maintenance, bus or train fare. Do not include car payments.	12.	\$
13.	Entertainment, clubs, recreation, newspapers, magazines, and books	13.	\$
14.	Charitable contributions and religious donations	14.	\$
15.	Insurance. Do not include insurance deducted from your pay or included in lines 4 or 20.		
	15a. Life insurance	15a.	\$
	15b. Health insurance	15b.	\$
	15c. Vehicle insurance	15c.	\$
	15d. Other insurance. Specify:	15d.	\$
16.	Taxes. Do not include taxes deducted from your pay or included in lines 4 or 20.	16.	\$
17.	Installment or lease payments:	17-	¢
	17a. Car payments for Vehicle 1	17a.	¢
	17b. Car payments for Vehicle 2	17b.	\$
	17c. Other. Specify:	17c.	\$
	17d. Other. Specify:	17d.	\$
18.	Your payments of alimony, maintenance, and support that you did not report as deduct your pay on line 5, <i>Schedule I, Your Income</i> (Official Form 106I).	ted from 18	\$
19.	Other payments you make to support others who do not live with you.		
	Specify:	19.	\$
20.	an an an Cabadula II II an A an E af this form on an Cabadula II.	Your Income.	
	20a. Mortgages on other property	20a.	\$
	20b. Real estate taxes	20b.	\$
	20c. Property, homeowner's, or renter's insurance	20c.	\$
	20d. Maintenance, repair, and upkeep expenses	20d.	\$
	20e. Homeowner's association or condominium dues	20e.	
		······································	······································

	irst Name Middle Name Last Name			
				••••••••••••••••••••••••••••••••••••••
Other Spe	ecify:		21.	+\$
ouror: op				
alculate	your monthly expenses.			
2a. Add li	nes 4 through 21.		22a.	\$
2b. Copy	line 22 (monthly expenses for Debtor 2), if any, from Official For	m 106J-2	22b.	\$
2c. Add li	ne 22a and 22b. The result is your monthly expenses.		22c.	\$
				L
alculate v	our monthly net income.			
	line 12 (your combined monthly income) from Schedule I.		23a.	\$
3b. Copy	your monthly expenses from line 22c above.		23b.	- \$
a Cubb	ract your monthly expenses from your monthly income.			
	esult is your monthly net income.		23c.	\$
		. 4		
	pect an increase or decrease in your expenses within the year		?	
For exampl	e, do you expect to finish paying for your car loan within the year ayment to increase or decrease because of a modification to the	or do you expect your		
or o		e terms of your mongage?		
mortgage p	ayment to increase or decrease because or a modification to the			
mortgage p				•
mortgage p	Explain here:			•
mortgage p				•
mortgage p				
mortgage p				

Fill in this information to identify your case:	
Debtor 1 First Name Middle Name Last Name Debtor 2 (Spouse, if filing) First Name Middle Name Last Name United States Bankruptcy Court for the:	Check if this is: An amended filing A supplement showing postpetition chapter 13 expenses as of the following date: <u>MM / DD / YYYY</u>
Official Form 106 I-2	

Schedule J-2: Expenses for Separate Household of Debtor 2

Use this form for Debtor 2's separate household expenses ONLY IF Debtor 1 and Debtor 2 maintain separate households. If Debtor 1 and Debtor 2 have one or more dependents in common, list the dependents on both Schedule J and this form. Answer the questions on this form only with respect to expenses for Debtor 2 that are not reported on Schedule J. Be as complete and accurate as possible. If more space is needed, attach another sheet to this form. On the top of any additional pages, write your name and case number (if known). Answer every question.

F	Part 1: Describe Your Hous	sehold				
1.	Do you and Debtor 1 maintain se	parate households?				
	No. Do not complete this forYes	n.				
2.	Do you have dependents?	No No	Dependent's relationship to	0	ependent's	Does dependent live
	Do not list Debtor 1 but list all other dependents of Debtor 2	Yes. Fill out this information for each dependent	Debtor 2:	a	ge	with you?
	regardless of whether listed as a	each dependent				No Yes
	dependent of Debtor 1 on Schedule J.					
	Do not state the dependents'			_		☐ No ☐ Yes
	names.					D No
						Yes
			· · · · · · · · · · · · · · · · · · ·			D No
		x				Yes
						No Ves
3.	Do your expenses include expenses of people other than yourself, your dependents, and Debtor 1?	No Yes		*******		
P	Part 2: Estimate Your Ongoi	ng Monthly Expenses				
e	expenses as of a date after the ban	bankruptcy filing date unless you a kruptcy is filed. -cash government assistance if you		ent in a	Chapter 13 c	case to report
S	such assistance and have included	I it on Schedule I: Your Income (Offi	cial Form 106I.)		Your expe	nses
ľ	 The rental or home ownership e any rent for the ground or lot. 	xpenses for your residence. Include	first mortgage payments and	4.	\$	
	If not included in line 4:					
	4a. Real estate taxes			4a.	\$	
	4b. Property, homeowner's, or re	enter's insurance		4b.		
	4c. Home maintenance, repair,	and upkeep expenses		4c.	\$	· · · · · · · · · · · · · · · · · · ·
	4d. Homeowner's association or	condominium dues	5	4d.	\$	

Schedule J-2: Expenses for Separate Household of Debtor 2

12/15

D	e	b	to	r	1

First Name Middle Name

Last Name

Case number (if known)

		Your expenses
5. Additional mortgage payments for your residence, such as home equity loans	5.	\$
6. Utilities: 6a. Electricity, heat, natural gas	6a.	\$
	6b.	\$
	6c.	\$
6c. Telephone, cell phone, internet, satellite, and cable services 6d. Other. Specify:	6d.	\$
	7.	\$
7. Food and housekeeping supplies		
8. Childcare and children's education costs	8.	\$
9. Clothing, laundry, and dry cleaning	9.	\$
0. Personal care products and services	10.	\$
1. Medical and dental expenses	11.	\$
2. Transportation. Include gas, maintenance, bus or train fare.	12.	\$
Do not include car payments.	13.	\$
3. Entertainment, clubs, recreation, newspapers, magazines, and books		\$ \$
4. Charitable contributions and religious donations	14.	۶
 Insurance. Do not include insurance deducted from your pay or included in lines 4 or 20. 		
15a. Life insurance	15a.	\$
15b. Health insurance	15b.	\$
15c. Vehicle insurance	15c.	\$
15d. Other insurance. Specify:	15d.	\$
 Taxes. Do not include taxes deducted from your pay or included in lines 4 or 20. Specify:	16.	\$
7. Installment or lease payments:		
17a. Car payments for Vehicle 1	17a.	\$
17b. Car payments for Vehicle 2	17b.	\$
	17c.	\$
17c. Other. Specify:		¢
17d. Other. Specify:	17d.	Φ
B. Your payments of alimony, maintenance, and support that you did not report as deducted from your pay on line 5, Schedule I, Your Income (Official Form 106I).	18.	\$
9. Other payments you make to support others who do not live with you.	19.	\$
		τ
20. Other real property expenses not included in lines 4 or 5 of this form or on Schedule I: Your Inco	me.	
20a. Mortgages on other property	20a.	\$
20b. Real estate taxes	20b.	\$
20c. Property, homeowner's, or renter's insurance	20c.	\$
20d. Maintenance, repair, and upkeep expenses	20d.	\$
20e. Homeowner's association or condominium dues	20e.	\$

Schedule J-2: Expenses for Separate Household of Debtor 2

First Name Middle Name Last Name Other. Specify:	e 22b of Schedule J to calculate	21.	+\$	
Your monthly expenses. Add lines 5 through 21.	e 22b of Schedule J to calculate	• • • • • • • • • • • • • • • • • • •	+\$	
Your monthly expenses. Add lines 5 through 21.	e 22b of Schedule J to calculate	• • • • • • • • • • • • • • • • • • •	+\$	
Your monthly expenses. Add lines 5 through 21.	e 22b of Schedule J to calculate	the		
The result is the monthly expenses of Debtor 2. Copy the result to line	e 22b of Schedule J to calculate	the		
The result is the monthly expenses of Debtor 2. Copy the result to line	e 22b of Schedule J to calculate	the		
otal expenses for Debtor 1 and Debtor 2.		uie oo	¢	
		22.	\$	
ne not used on this form.				
o you expect an increase or decrease in your expenses within th	ne year after you file this form	?		
or example, do you expect to finish paying for your car loan within the	e year or do you expect your			
ortgage payment to increase or decrease because of a modification	to the terms of your mortgage?			
No.				
Yes. Explain here:				
				MMAAAMME' 10-1000-1000-1000-1000-1000-
		•		
	· · ·			
			· · · · ·	

Fill in this inf	formation to identify y	our case:	
Debtor 1	First Name	Middle Name	Last Name
Debtor 2 (Spouse, if filing)	First Name	Middle Name	Last Name
United States E Case number	Bankruptcy Court for the: _		District of (State)
(If known)			· · ·

12/15

Official Form 106Dec

Declaration About an Individual Debtor's Schedules

If two married people are filing together, both are equally responsible for supplying correct information.

You must file this form whenever you file bankruptcy schedules or amended schedules. Making a false statement, concealing property, or obtaining money or property by fraud in connection with a bankruptcy case can result in fines up to \$250,000, or imprisonment for up to 20 years, or both. 18 U.S.C. §§ 152, 1341, 1519, and 3571.

d you pay or agree to p	ay someone	who is NO	T an atto	orney to help you	ill out bankrup	tcy forms?	•		
No									
Yes. Name of person					Attach Bankrupto	y Petition Pre	parer's Notice	, Declaration,	and
					Signature (Officia	Form 119).			
Index nonality of parium	I doclare tha	t I have rea	nd the su	immary and scher	ules filed with	this decla	ration and		
Inder penalty of perjury, hat they are true and co	I declare tha rect.	t I have rea	nd the su	ummary and sched	lules filed with	this decla	ration and		
Inder penalty of perjury, hat they are true and cor	l declare tha rect.	t I have rea	nd the su	ummary and schee	ules filed with	this decla	ration and		
Inder penalty of perjury, hat they are true and co	l declare tha rect.	t I have rea	nd the su	ummary and schee	lules filed with	this decla	ration and		
nder penalty of perjury, nat they are true and cor	l declare tha rect.	t I have rea	nd the su	ummary and schee	lules filed with	this decla	ration and		
hat they are true and con	l declare tha rect.	t I have rea				this decla	ration and		
hat they are true and con	l declare tha rect.	t I have rea		ummary and sched		this decla	ration and		
Inder penalty of perjury, hat they are true and con Signature of Debtor 1 Date	I declare tha rect.	t I have rea				this decla	ration and	•	

Fill in this inf	formation to identify	your case:	
Debtor 1			
	First Name	Middle Name	Last Name
Debtor 2			
(Spouse, if filing)	First Name	Middle Name	Last Name
United States E	ankruptcy Court for the:		District of
			(State)
Case number			<u> </u>
(If known)			

Official Form 107

Statement of Financial Affairs for Individuals Filing for Bankruptcy

ut Your Marital Status and Whore You Lived Refere

12/15

Be as complete and accurate as possible. If two married people are filing together, both are equally responsible for supplying correct information. If more space is needed, attach a separate sheet to this form. On the top of any additional pages, write your name and case number (if known). Answer every question.

Fai	t 1:	Give Details About	t four marital State				
1. \	Vhat	is your current marital	status?				
		larried lot married					
I		ng the last 3 years, have lo res. List all of the places y					
		Debtor 1:		Dates Debtor 1 lived there	Debtor 2:		Dates Debtor 2 lived there
					Same as Debtor 1		Same as Debtor 1
		Number Street		From To	Number Street		From To
		City	State ZIP Code	* 	City State	ZIP Code	
					Same as Debtor 1		Same as Debtor 1
		Number Street		From To	Number Street		From To
		City	State ZIP Code		City State	ZIP Code	
	and	in the last 8 years, did y <i>territories</i> include Arizona No	ou ever live with a sp , California, Idaho, Lou	isiana, Nevada, Nev	alent in a community property stat v Mexico, Puerto Rico, Texas, Washin	e or territory? (Cor	mmunity property states sin.)
	י ם י	res. Make sure you fill ou	t Schedule H: Your Coo	debtors (Official For	n 106H).		

Official Form 107

First Name Middle Name Last №	varne			
id you have any income from employment ill in the total amount of income you received you are filing a joint case and you have inco	from all jobs and all busir	nesses, including part-tin	ne activities.	dar years?
No Yes. Fill in the details.				
	Debtor 1		Debtor 2	
	Sources of income Check all that apply.	Gross income (before deductions and exclusions)	Sources of income Check all that apply.	Gross income (before deductions a exclusions)
From January 1 of current year until the date you filed for bankruptcy:	 Wages, commissions, bonuses, tips Operating a business 	\$	 Wages, commissions, bonuses, tips Operating a business 	\$
For last calendar year:	Wages, commissions, bonuses, tips	\$	Wages, commissions, bonuses, tips	\$
(January 1 to December 31,)	Operating a business		Operating a business	
For the calendar year before that:	Wages, commissions, bonuses, tips	¢	Wages, commissions, bonuses, tips	8
(January 1 to December 31,	Operating a business	φ	Operating a business	Ψ
Did you receive any other income during the include income regardless of whether that inc and other public benefit payments; pensions; vinnings. If you are filing a joint case and you	ome is taxable. Examples rental income; interest; div have income that you rec	of other income are alin vidends; money collected eived together, list it only	d from lawsuits; royalties; an y once under Debtor 1.	Security, unemploymend gambling and lotte
Did you receive any other income during the include income regardless of whether that income of the public benefit payments; pensions;	ome is taxable. Examples rental income; interest; div have income that you rec each source separately. Do	of other income are alin vidends; money collected eived together, list it only	d from lawsuits; royalties; au y once under Debtor 1. t you listed in line 4.	Security, unemployme
Did you receive any other income during the include income regardless of whether that include income regardless of whether that include other public benefit payments; pensions; vinnings. If you are filing a joint case and you ist each source and the gross income from e No	ome is taxable. Examples rental income; interest; div have income that you rec	of other income are alin vidends; money collected eived together, list it only	d from lawsuits; royalties; an y once under Debtor 1. t you listed in line 4. Debtor 2	Security, unemployme nd gambling and lotte
Did you receive any other income during the include income regardless of whether that include income regardless of whether that include other public benefit payments; pensions; vinnings. If you are filing a joint case and you ist each source and the gross income from e No	ome is taxable. Examples rental income; interest; div have income that you rec each source separately. Do	of other income are alin vidends; money collected eived together, list it only	d from lawsuits; royalties; au y once under Debtor 1. t you listed in line 4.	nd gambling and lotte Gross income from each source
Did you receive any other income during the include income regardless of whether that include income regardless of whether that include other public benefit payments; pensions; vinnings. If you are filing a joint case and you ist each source and the gross income from e No	ome is taxable. Examples rental income; interest; div have income that you rec each source separately. Do Debtor 1 Sources of income	of other income are alin vidends; money collected eived together, list it only o not include income that Gross income from each source (before deductions and	d from lawsuits; royalties; an y once under Debtor 1. t you listed in line 4. Debtor 2 Sources of income	nd gambling and lotte Gross income from each source (before deductions a
Did you receive any other income during the include income regardless of whether that include income regardless of whether that include income public benefit payments; pensions; winnings. If you are filing a joint case and you ist each source and the gross income from e No Yes. Fill in the details.	ome is taxable. Examples rental income; interest; div have income that you rec each source separately. Do Debtor 1 Sources of income	of other income are alin vidends; money collected eived together, list it only o not include income that Gross income from each source (before deductions and	d from lawsuits; royalties; an y once under Debtor 1. t you listed in line 4. Debtor 2 Sources of income	nd gambling and lotte Gross income from each source (before deductions a
 Did you receive any other income during the include income regardless of whether that include income regardless of whether that include income regardless of whether that include income public benefit payments; pensions; winnings. If you are filing a joint case and you ist each source and the gross income from etails are and the gross income from etails. No Yes. Fill in the details. 	ome is taxable. Examples rental income; interest; div have income that you rec each source separately. Do Debtor 1 Sources of income	of other income are alin vidends; money collected eived together, list it only o not include income that Gross income from each source (before deductions and	d from lawsuits; royalties; an y once under Debtor 1. t you listed in line 4. Debtor 2 Sources of income	nd gambling and lotte Gross income from each source (before deductions a
 Did you receive any other income during the include income regardless of whether that include income regardless of whether that include income regardless of whether that include income public benefit payments; pensions; winnings. If you are filing a joint case and you ist each source and the gross income from etails are and the gross income from etails. No Yes. Fill in the details. 	ome is taxable. Examples rental income; interest; div have income that you rec each source separately. Do Debtor 1 Sources of income	of other income are alin vidends; money collected eived together, list it only o not include income that Gross income from each source (before deductions and	d from lawsuits; royalties; an y once under Debtor 1. t you listed in line 4. Debtor 2 Sources of income	nd gambling and lotte Gross income from each source (before deductions a
 Did you receive any other income during the include income regardless of whether that income other public benefit payments; pensions; vinnings. If you are filing a joint case and you sist each source and the gross income from ether income from ether income source and the details. No Yes. Fill in the details. From January 1 of current year until the date you filed for bankruptcy:	ome is taxable. Examples rental income; interest; div have income that you rec each source separately. Do Debtor 1 Sources of income	of other income are alin vidends; money collected eived together, list it only o not include income that Gross income from each source (before deductions and	d from lawsuits; royalties; an y once under Debtor 1. t you listed in line 4. Debtor 2 Sources of income	nd gambling and lotte Gross income from each source (before deductions a
 Did you receive any other income during the include income regardless of whether that income other public benefit payments; pensions; winnings. If you are filing a joint case and you sist each source and the gross income from ether income from ether income and the gross income from ether income et	ome is taxable. Examples rental income; interest; div have income that you rec each source separately. Do Debtor 1 Sources of income	of other income are alin vidends; money collected eived together, list it only o not include income that Gross income from each source (before deductions and	d from lawsuits; royalties; an y once under Debtor 1. t you listed in line 4. Debtor 2 Sources of income	nd gambling and lotte Gross income from each source (before deductions a
Did you receive any other income during the include income regardless of whether that income other public benefit payments; pensions; winnings. If you are filing a joint case and you ist each source and the gross income from etails are income from etails. No Yes. Fill in the details. From January 1 of current year until the date you filed for bankruptcy: For last calendar year: (January 1 to December 31,) YYYY For the calendar year before that:	ome is taxable. Examples rental income; interest; div have income that you rec each source separately. Do Debtor 1 Sources of income	of other income are alin vidends; money collected eived together, list it only o not include income that Gross income from each source (before deductions and	d from lawsuits; royalties; an y once under Debtor 1. t you listed in line 4. Debtor 2 Sources of income	nd gambling and lotte Gross income from each source (before deductions a
Did you receive any other income during the include income regardless of whether that income other public benefit payments; pensions; vinnings. If you are filing a joint case and you sist each source and the gross income from ether income from ether income source and the gross income from ether income from ether income source and the gross income from ether income from ether income source and the gross income from ether income from ether income source and the gross income from ether income from ether income source and the gross income from ether income from ether income source and the gross income from ether income from ether income source and the gross income from ether income from ether income source and the gross income from ether income source and the gross income from ether income from ether income source and the gross income from ether income from ether income source and the gross income from ether income from ether income source and the gross and the gross income source and the g	ome is taxable. Examples rental income; interest; div have income that you rec each source separately. Do Debtor 1 Sources of income	of other income are alin vidends; money collected eived together, list it only on ot include income tha Gross income from each source (before deductions and exclusions) \$\$ \$\$ \$\$ \$	d from lawsuits; royalties; an y once under Debtor 1. t you listed in line 4. Debtor 2 Sources of income	nd gambling and lotte Gross income from each source (before deductions a

Statement of Financial Affairs for Individuals Filing for Bankruptcy

ptor 1	First Name	Middle Nam	ne	Last Name		Case n	umber (if known)	
art 3:	List Ce	tain Paym	ents You	Made Before	• You Filed f	for Bankruptcy	· · ·	
				· ·				
Are eith	er Debto	r 1's or Debt	tor 2's debt	ts primarily co	nsumer debts	?		
	Neither	Debtor 1 no	r Debtor 2 l	has primarily (consumer deb	ots. Consumer debts are	e defined in 11 U.S.C. § 101(8) as
	"incurre	d by an individ	dual primari	ily for a person	al, family, or ho	ousehold purpose."		
	During	ne 90 days be	efore you fil	ed for bankrup	tcy, did you pa	y any creditor a total of \$	56,225° or more?	
	No.	Go to line 7.		2				
	Yes	total amount	t vou paid th	hat creditor. Do	not include pa	\$6,225* or more in one of ayments for domestic superts to an attorney for the	or more payments and the pport obligations, such as his bankruptcy case.	* .
	* Subje						fter the date of adjustment.	
□ Yes	Debtor	1 or Debtor	े 2 or both h	ave primarily (consumer det	ots.		
						y any creditor a total of	\$600 or more?	
~		Go to line 7.						
			Sada anadikas	to whom you	naid a total of (\$600 or more and the to	tal amount you paid that	
	L Yes	creditor. Do	not include	payments for o	domestic suppo	ort obligations, such as y for this bankruptcy cas	child support and	
					Dates of	Total amount paid	Amount you still owe	Was this payment fo
					payment			
						\$	\$	Mortgage
	Cr	editor's Name		-	-			
	N N	mber Street						Credit card
	N.	IDEI SUEEL						Loan repayment
	_							Guppliers or vend
	Cit		State	ZIP Code				Other
		v						
	- 2000000000	y			2-00-000 - 0.0 - 0.0 - 0.0 - 0.0 - 0.0 - 0.0 - 0.0 - 0.0 - 0.0 - 0.0 - 0.0 - 0.0 - 0.0 - 0.0 - 0.0 - 0.0 - 0.0		*	*
	- 	y		aun - 10 - 181 - 1875 1875 1875 1875 1875 1875 1875 1875	<u>в народор, Альсковско соодин</u> турски. -	anaan meneration and a factor of the second second and a second	s	. Mortgage
		y editor's Name	20000000000000000000000000000000000000		9. 49.992 - Co. C.	\$	\$	☐ Mortgage ☐ Car
	· .	editor's Name				\$	\$	
	· .	- 1994 - 000000 4000 500 4000 4000 400 - 4 -	annan de Alexandre			\$ <u>.</u>	\$	Car
	· .	editor's Name				\$	\$	Car Credit card Loan repayment
		editor's Name mber Street	Caste	718 Code		\$	\$	Car Credit card Loan repayment
	· .	editor's Name mber Street	State	ZIP Code		\$	\$	 Car Credit card Loan repayment Suppliers or veno
		editor's Name mber Street	State	ZIP Code			\$	Car Credit card Loan repayment Suppliers or vence Other
		editor's Name Imber Street	State	ZIP Code		\$ \$	\$	 Car Credit card Loan repayment Suppliers or vend Other Mortgage
		editor's Name mber Street	State	ZIP Code			\$	 Car Credit card Loan repayment Suppliers or vend Other Mortgage Car
		editor's Name Imber Street	State	ZIP Code			\$	 Car Credit card Loan repayment Suppliers or vend Other Mortgage Car Credit card
		editor's Name mber Street ly editor's Name	State	ZIP Code			\$	 Car Credit card Loan repayment Suppliers or vend Other
		editor's Name mber Street ly editor's Name	State	ZIP Code			\$	 Car Credit card Loan repayment Suppliers or vend Other
		editor's Name mber Street y editor's Name mber Street	State	ZIP Code			\$ \$	 Car Credit card Loan repayment Suppliers or vend Other Mortgage Car Credit card

Statement of Financial Affairs for Individuals Filing for Bankruptcy

	First Name	Middle Name	Last Nam	ne					
side	ers include you	r relatives; an	y general partr	ners; relatives	of any g	jeneral partners; pa	rtnerships of which	no was an insider? you are a general partner; securities; and any managing	
jent	t, including one as child suppo	for a busines	ss you operate	as a sole pro	prietor. 1	11 U.S.C. § 101. Inc	clude payments for	domestic support obligations,	
N						•			
I Y	es. List all payr	ments to an in	Isider.		tes of yment	Total amount paid	Amount you still owe	Reason for this payment	
						\$	\$		
	Insider's Name					·			
	Number Street								
	-	-	State ZIP Co			,			
-win	City		State ZIP Co						
	Insider's Name					\$	\$		
							*		
	Number Street								e
	City	e vou filed fo	State ZIP Co		e any pa	ayments or transfe	er any property on	account of a debt that benefited	
ithi in clu	in 1 year befor nsider? de payments o No	n debts guara	or bankruptcy,	, did you mal Ined by an ins		ayments or transfe	er any property on	account of a debt that benefited	
ithi nin clu l N	in 1 year befor nsider? de payments o	n debts guara	or bankruptcy,	, did you mal ned by an ins der.	ider.		-		
ithi nin clu l N	in 1 year befor nsider? de payments o No	n debts guara	or bankruptcy,	, did you mal ned by an ins der. Dat		ayments or transfe Total amount paid	-	account of a debt that benefited Reason for this payment Include creditor's name	
ithi nin clu l N	in 1 year befor nsider? de payments o No Yes. List all pay	n debts guara	or bankruptcy,	, did you mal ned by an ins der. Dat	ider. es of	Total amount	Amount you still	Reason for this payment	
ithi nin clu l N	in 1 year befor nsider? de payments o No	n debts guara	or bankruptcy,	, did you mal ned by an ins der. Dat	ider. es of	Total amount	Amount you still	Reason for this payment	
ithi nin clu l N	in 1 year befor nsider? de payments o No Yes. List all pay	n debts guara	or bankruptcy,	, did you mal ned by an ins der. Dat	ider. es of	Total amount	Amount you still	Reason for this payment	
ithi in clu	in 1 year befor isider? de payments o No Yes. List all pay	n debts guara	or bankruptcy,	, did you mal ned by an ins der. Dat	ider. es of	Total amount	Amount you still	Reason for this payment	
ithi nin clu l N	in 1 year befor hsider? de payments o No Yes. List all pay Insider's Name	n debts guara	or bankruptcy, anteed or cosig	, did you mal	ider. es of	Total amount	Amount you still	Reason for this payment	
ithi in clu	in 1 year befor isider? de payments o No Yes. List all pay	n debts guara	or bankruptcy,	, did you mal	ider. es of	Total amount	Amount you still	Reason for this payment	
ithi in clu	in 1 year befor isider? de payments o No Yes. List all pay Insider's Name Number Street City	n debts guara	or bankruptcy, anteed or cosig	, did you mal	ider. es of	Total amount	Amount you still	Reason for this payment	
ithi in clu	in 1 year befor hsider? de payments o No Yes. List all pay Insider's Name	n debts guara	or bankruptcy, anteed or cosig	, did you mal	ider. es of	Total amount paid \$	Amount you still	Reason for this payment	
ithi in clu	in 1 year befor isider? de payments o No Yes. List all pay Insider's Name Number Street City	n debts guara	or bankruptcy, anteed or cosig	, did you mal	ider. es of	Total amount paid \$	Amount you still	Reason for this payment	
ithi in clu	in 1 year befor hsider? de payments o No Yes. List all pay Insider's Name Number Street City	n debts guara	or bankruptcy, anteed or cosig	, did you mal	ider. es of	Total amount paid \$	Amount you still	Reason for this payment	
ithi nin clu	in 1 year befor hsider? de payments o No Yes. List all pay Insider's Name Number Street City	n debts guara	or bankruptcy, anteed or cosig	, did you mal aned by an ins der. Dat pay der.	ider. es of	Total amount paid \$	Amount you still	Reason for this payment	· · ·

1	Last Name		Case number (if known)		
First Name Middle Name.	Equivante				
t 4: Identify Legal Actions, Rep	ossessions, a	and Foreclosures			
Vithin 1 year before you filed for bankr			court action or administra	tive proceeding	2
ist all such matters, including personal in	niury cases sma	all claims actions divorce	s collection suits paternity a	ctions, support or	custody modification
nd contract disputes.	ijury cases, sine		o, oonoonon oono, paronny a		
No					
Yes. Fill in the details.					
	Nature of	the case	Court or agency	n de la serie d	Status of the case
			,		_
Case title	i		Court Name		Pending
					On appeal
			Number Street		
Case number			City State	ZIP Code	
-			ony orde		
Case title			Court Name		Pending
					🗋 On appeai
·	-		Number Street		Concluded
Case number					
			City State		
Vithin 1 year before you filed for bankn Check all that apply and fill in the details I No. Go to line 11.		y of your property repos	City State	ZIP Code hed, attached, s	eized, or levied?
Check all that apply and fill in the details		y of your property repos			eized, or levied?
Check all that apply and fill in the details No. Go to line 11.		045 20 S. 2007-5			
Check all that apply and fill in the details No. Go to line 11.		y of your property repos Describe the property		hed, attached, s	
Check all that apply and fill in the details No. Go to line 11.		045 20 S. 2007-5		hed, attached, s	
Check all that apply and fill in the details No. Go to line 11.		045 20 S. 2007-5		hed, attached, s	
 Check all that apply and fill in the details No. Go to line 11. Yes. Fill in the information below. 		045 20 S. 2007-5		hed, attached, s	
Check all that apply and fill in the details No. Go to line 11. Yes. Fill in the information below.		Describe the property		hed, attached, s	
Check all that apply and fill in the details No. Go to line 11. Yes. Fill in the information below.		Describe the property Explain what happened	J seessed, foreclosed, garnis	hed, attached, s	
Check all that apply and fill in the details No. Go to line 11. Yes. Fill in the information below.		Describe the property Explain what happened	J seessed, foreclosed, garnis	hed, attached, s	
Check all that apply and fill in the details No. Go to line 11. Yes. Fill in the information below.		Describe the property Explain what happened Property was repos Property was forect	J seessed, foreclosed, garnis seessed. losed.	hed, attached, s	
Check all that apply and fill in the details No. Go to line 11. Yes. Fill in the information below.		Describe the property Explain what happened Property was repos Property was forect Property was garning	J ssessed, foreclosed, garnis ssessed. losed. shed.	hed, attached, s	
Creditor's Name		Describe the property Explain what happened Property was repos Property was forect Property was garning	J seessed, foreclosed, garnis seessed. losed.	hed, attached, s	
Check all that apply and fill in the details No. Go to line 11. Yes. Fill in the information below.	below.	Describe the property Explain what happened Property was repos Property was forect Property was garning	J ssessed, foreclosed, garnis ssessed. losed. shed.	hed, attached, s	Value of the property
Check all that apply and fill in the details No. Go to line 11. Yes. Fill in the information below.	below.	Describe the property Explain what happened Property was repos Property was forect Property was garnia Property was attact	J ssessed, foreclosed, garnis ssessed. losed. shed.	hed, attached, s	Value of the property
Check all that apply and fill in the details No. Go to line 11. Yes. Fill in the information below.	below.	Describe the property Explain what happened Property was repos Property was forect Property was garnia Property was attact	J ssessed, foreclosed, garnis ssessed. losed. shed.	hed, attached, s	Value of the property
Check all that apply and fill in the details No. Go to line 11. Yes. Fill in the information below.	below.	Describe the property Explain what happened Property was repos Property was forect Property was garnia Property was attact	J ssessed, foreclosed, garnis ssessed. losed. shed.	hed, attached, s	Value of the property
Check all that apply and fill in the details No. Go to line 11. Yes. Fill in the information below.	below.	Describe the property Explain what happened Property was repos Property was forect Property was garnia Property was attact	J ssessed, foreclosed, garnis ssessed. losed. shed.	hed, attached, s	Value of the property
Check all that apply and fill in the details No. Go to line 11. Yes. Fill in the information below.	below.	Describe the property Explain what happened Property was repos Property was forect Property was garnia Property was attact	J ssessed, foreclosed, garnis ssessed. losed. shed.	hed, attached, s	Value of the property
Check all that apply and fill in the details No. Go to line 11. Yes. Fill in the information below.	below.	Describe the property Explain what happened Property was repos Property was forect Property was garnit Property was attact Describe the property	J ssessed, foreclosed, garnis ssessed. losed. shed.	hed, attached, s	Value of the property
Check all that apply and fill in the details in the No. Go to line 11. Yes. Fill in the information below. Creditor's Name City State Creditor's Name Creditor's Name	below.	Describe the property Explain what happened Property was repos Property was forect Property was attact Describe the property Explain what happened	J seessed, foreclosed, garnis seessed. losed. shed. hed, seized, or levied.	hed, attached, s	Value of the property
Check all that apply and fill in the details in the No. Go to line 11. Yes. Fill in the information below. Creditor's Name Number Street City State Creditor's Name Creditor's Name	below.	Describe the property Explain what happened Property was repose Property was forect Property was garning Property was direct Explain what happened Property was repose	ssessed. losed. shed. hed, seized, or levied.	hed, attached, s	Value of the property
Check all that apply and fill in the details in the No. Go to line 11. Yes. Fill in the information below. Creditor's Name Number Street City State Creditor's Name Creditor's Name	below.	Describe the property Explain what happened Property was repose Property was forect Property was attact Describe the property Explain what happened Explain what happened Property was attact Describe the property Property was repose Property was repose Property was forect	ssessed. losed. shed. hed, seized, or levied.	hed, attached, s	Value of the property
Check all that apply and fill in the details I No. Go to line 11. Yes. Fill in the information below. Creditor's Name Number Street City State Creditor's Name Number Street	below.	Describe the property Explain what happened Property was repose Property was forect Property was garning Property was direct Explain what happened Property was repose	ssessed. hed, seized, or levied.	hed, attached, s	

Statement of Financial Affairs for Individuals Filing for Bankruptcy

·		se number (if known)		
First Name Middle Name Last N	lame			
	it is a second the second second second second		ant off any and	
thin 90 days before you filed for bankrup	tcy, did any creditor, including a bank or fin	nancial institution,	set on any amo	bunts from your
counts or refuse to make a payment bec	ause you owed a debt?			
No				
Yes. Fill in the details.				
Tes. Fin in the details.			*	
	Describe the action the creditor took		Date action	Amount
			was taken	
Creditor's Name				
				\$
Number Street			1	
	· -			
City State ZIP Code	Last 4 digits of account number: XXXX			
City State ZIP Code		- <u> </u>		
thin 1 year before you filed for bankrupto	cy, was any of your property in the possess	ion of an assignee	e for the benefit	of
editors, a court-appointed receiver, a cus	stodian, or another official?			
No				
Yes				
<u> </u>				
5: List Certain Gifts and Contribu	tions			
	tcy, did you give any gifts with a total value	of more than \$600) per person?	
No	tcy, did you give any gifts with a total value	of more than \$600) per person?	
	tcy, did you give any gifts with a total value	of more than \$600) per person?	
No Yes. Fill in the details for each gift.		of more than \$600		
No Yes. Fill in the details for each gift. Gifts with a total value of more than \$600	tcy, did you give any gifts with a total value Describe the gifts	of more than \$600	Dates you gave	Value
No Yes. Fill in the details for each gift.		of more than \$600		Value
No Yes. Fill in the details for each gift. Gifts with a total value of more than \$600		of more than \$600	Dates you gave	Value
No Yes. Fill in the details for each gift. Gifts with a total value of more than \$600		of more than \$600	Dates you gave	Value
No Yes. Fill in the details for each gift. Gifts with a total value of more than \$600 per person		of more than \$600	Dates you gave	Value \$
No Yes. Fill in the details for each gift. Gifts with a total value of more than \$600		of more than \$600	Dates you gave	Value \$
No Yes. Fill in the details for each gift. Gifts with a total value of more than \$600 per person		of more than \$600	Dates you gave	Value \$
No Yes. Fill in the details for each gift. Gifts with a total value of more than \$600 per person		of more than \$600	Dates you gave	Value \$ \$
No Yes. Fill in the details for each gift. Gifts with a total value of more than \$600 per person		of more than \$600	Dates you gave	Value \$ \$
No Yes. Fill in the details for each gift. Gifts with a total value of more than \$600 per person		of more than \$600	Dates you gave	Value \$ \$
No Yes. Fill in the details for each gift. Gifts with a total value of more than \$600 per person		of more than \$600	Dates you gave	Value \$ \$
No Yes. Fill in the details for each gift. Gifts with a total value of more than \$600 per person Person to Whom You Gave the Gift Number Street		of more than \$600	Dates you gave	Value \$ \$
No Yes. Fill in the details for each gift. Gifts with a total value of more than \$600 per person		of more than \$600	Dates you gave	Value \$ \$
No Yes. Fill in the details for each gift. Gifts with a total value of more than \$600 per person Person to Whom You Gave the Gift Number Street		of more than \$600	Dates you gave	Value \$\$
No Yes. Fill in the details for each gift. Gifts with a total value of more than \$600 per person Person to Whom You Gave the Gift Number Street		of more than \$600	Dates you gave	Value \$\$
No Yes. Fill in the details for each gift. Gifts with a total value of more than \$600 per person Person to Whom You Gave the Gift Number Street City State ZIP Code		of more than \$600	Dates you gave	Value \$ \$
No Yes. Fill in the details for each gift. Gifts with a total value of more than \$600 per person Person to Whom You Gave the Gift Number Street City State ZIP Code Person's relationship to you	Describe the gifts	of more than \$600	Dates you gave the gifts	Value \$\$
No Yes. Fill in the details for each gift. Gifts with a total value of more than \$600 per person Person to Whom You Gave the Gift Number Street City State ZIP Code		of more than \$600	Dates you gave	\$ \$
No Yes. Fill in the details for each gift. Gifts with a total value of more than \$600 per person Person to Whom You Gave the Gift Number Street City State ZIP Code Person's relationship to you	Describe the gifts	of more than \$600	Dates you gave the gifts	\$ \$
No Yes. Fill in the details for each gift. Gifts with a total value of more than \$600 per person Person to Whom You Gave the Gift Number Street City State ZIP Code Person's relationship to you	Describe the gifts	of more than \$600	Dates you gave the gifts	\$ \$
No Yes. Fill in the details for each gift. Gifts with a total value of more than \$600 per person Person to Whom You Gave the Gift Number Street City State ZIP Code Person's relationship to you	Describe the gifts	of more than \$600	Dates you gave the gifts	\$ \$
No Yes. Fill in the details for each gift. Gifts with a total value of more than \$600 per person Person to Whom You Gave the Gift Number Street City State ZIP Code Person's relationship to you	Describe the gifts	of more than \$600	Dates you gave the gifts	\$ \$
No Yes. Fill in the details for each gift. Gifts with a total value of more than \$600 per person Person to Whom You Gave the Gift Number Street City State ZIP Code Person's relationship to you	Describe the gifts	of more than \$600	Dates you gave the gifts	\$ \$
No Yes. Fill in the details for each gift. Gifts with a total value of more than \$600 per person Person to Whom You Gave the Gift Number Street City State ZIP Code Person's relationship to you	Describe the gifts	of more than \$600	Dates you gave the gifts	\$ \$
No Yes. Fill in the details for each gift. Gifts with a total value of more than \$600 per person Person to Whom You Gave the Gift Number Street City State ZIP Code Person's relationship to you	Describe the gifts	of more than \$600	Dates you gave the gifts	\$ \$
No Yes. Fill in the details for each gift. Gifts with a total value of more than \$600 per person Person to Whom You Gave the Gift Number Street City State ZIP Code Person's relationship to you	Describe the gifts	of more than \$600	Dates you gave the gifts	\$ \$
No Yes. Fill in the details for each gift. Gifts with a total value of more than \$600 per person Person to Whom You Gave the Gift Number Street City State ZIP Code Person's relationship to you	Describe the gifts	of more than \$600	Dates you gave the gifts	\$ \$
No Yes. Fill in the details for each gift. Gifts with a total value of more than \$600 per person Person to Whom You Gave the Gift Number Street City State ZIP Code Person's relationship to you	Describe the gifts	of more than \$600	Dates you gave the gifts	\$ \$
No Yes. Fill in the details for each gift. Gifts with a total value of more than \$600 per person Person to Whom You Gave the Gift Number Street City State ZIP Code Person's relationship to you	Describe the gifts	of more than \$600	Dates you gave the gifts	\$ \$
No Yes. Fill in the details for each gift. Gifts with a total value of more than \$600 per person Person to Whom You Gave the Gift Number Street City State ZIP Code Person's relationship to you	Describe the gifts	of more than \$600	Dates you gave the gifts	\$ \$

Statement of Financial Affairs for Individuals Filing for Bankruptcy

First Name Middle Name Last N	ame Case number (if known)		
thin 2 years before you filed for bankrupt	cy, did you give any gifts or contributions with a total value	of more than \$600	to any charity?
No			
Yes. Fill in the details for each gift or contri	bution.		
	Describe what you contributed	Date you	Value
Gifts or contributions to charities that total more than \$600	Describe what you contributed	contributed	value
			¢
Charity's Name		-	Φ
			s
			Ψ
Number Street		-	
City State ZIP Code			
6: List Certain Losses			
r gambling? No Yes. Fill in the details.	y or since you filed for bankruptcy, did you lose anything b	Date of your loss	Volue
gambling?	Describe any insurance coverage for the loss Include the amount that insurance has paid. List pending insurance claims on line 33 of <i>Schedule AVB: Property.</i>	Date of your loss	Value of property lost
gambling? No Yes. Fill in the details. Describe the property you lost and how	Describe any insurance coverage for the loss Include the amount that insurance has paid. List pending insurance	Date of your loss	
gambling? No Yes. Fill in the details. Describe the property you lost and how	Describe any insurance coverage for the loss Include the amount that insurance has paid. List pending insurance	Date of your loss	
gambling? No Yes. Fill in the details. Describe the property you lost and how	Describe any insurance coverage for the loss Include the amount that insurance has paid. List pending insurance	Date of your loss	
gambling? No Yes. Fill in the details. Describe the property you lost and how the loss occurred	Describe any insurance coverage for the loss Include the amount that insurance has paid. List pending insurance claims on line 33 of <i>Schedule A/B: Property</i> .	Date of your loss	
gambling? No Yes. Fill in the details. Describe the property you lost and how	Describe any insurance coverage for the loss Include the amount that insurance has paid. List pending insurance claims on line 33 of <i>Schedule A/B: Property</i> .	Date of your loss	
gambling? No Yes. Fill in the details. Describe the property you lost and how the loss occurred 7: List Certain Payments or Trans //thin 1 year before you filed for bankruptcy or pre-	Describe any insurance coverage for the loss Include the amount that insurance has paid. List pending insurance claims on line 33 of <i>Schedule AVB: Property.</i>	sfer any property to	\$
gambling? No Yes. Fill in the details. Describe the property you lost and how the loss occurred 7: List Certain Payments or Trans //thin 1 year before you filed for bankruptcy or pre-	Describe any insurance coverage for the loss Include the amount that insurance has paid. List pending insurance claims on line 33 of <i>Schedule AVB: Property:</i>	sfer any property to	\$
 gambling? No Yes. Fill in the details. Describe the property you lost and how the loss occurred 7: List Certain Payments or Trans 7: List Certain Payments or Trans 7: Itist certain payments or precision precision	Describe any insurance coverage for the loss Include the amount that insurance has paid. List pending insurance claims on line 33 of <i>Schedule AVB: Property:</i>	sfer any property to	\$
 gambling? No Yes. Fill in the details. Describe the property you lost and how the loss occurred 7: List Certain Payments or Transfithin 1 year before you filed for bankruptcy or preclude any attorneys, bankruptcy petition prepared No 	Describe any insurance coverage for the loss Include the amount that insurance has paid. List pending insurance claims on line 33 of <i>Schedule AVB: Property:</i>	sfer any property to	\$
 gambling? No Yes. Fill in the details. Describe the property you lost and how the loss occurred 7: List Certain Payments or Transfithin 1 year before you filed for bankruptcy or preclude any attorneys, bankruptcy petition prepared No 	Describe any insurance coverage for the loss Include the amount that insurance has paid. List pending insurance claims on line 33 of Schedule AVB: Property. ffers cy, did you or anyone else acting on your behalf pay or tran paring a bankruptcy petition? parers, or credit counseling agencies for services required in your behalf pay or tran parers, or credit counseling agencies for services required in your behalf pay or tran parers, or credit counseling agencies for services required in your behalf pay or transparent pay	sfer any property to our bankruptcy.	s
gambling? No Yes. Fill in the details. Describe the property you lost and how the loss occurred 7: List Certain Payments or Trans Within 1 year before you filed for bankruptcy or preclude any attorneys, bankruptcy petition prepared No Yes. Fill in the details. Person Who Was Paid	Describe any insurance coverage for the loss Include the amount that insurance has paid. List pending insurance claims on line 33 of Schedule AVB: Property. ffers cy, did you or anyone else acting on your behalf pay or tran paring a bankruptcy petition? parers, or credit counseling agencies for services required in your behalf pay or tran parers, or credit counseling agencies for services required in your behalf pay or tran parers, or credit counseling agencies for services required in your behalf pay or transparent pay	sfer any property to our bankruptcy.	s
 gambling? No Yes. Fill in the details. Describe the property you lost and how the loss occurred 7: List Certain Payments or Trans Tithin 1 year before you filed for bankruptco onsulted about seeking bankruptcy or preclude any attorneys, bankruptcy petition prep No Yes. Fill in the details. 	Describe any insurance coverage for the loss Include the amount that insurance has paid. List pending insurance claims on line 33 of Schedule AVB: Property. ffers cy, did you or anyone else acting on your behalf pay or tran paring a bankruptcy petition? parers, or credit counseling agencies for services required in your behalf pay or tran parers, or credit counseling agencies for services required in your behalf pay or tran parers, or credit counseling agencies for services required in your behalf pay or transparent pay	sfer any property to our bankruptcy.	s
gambling? No Yes. Fill in the details. Describe the property you lost and how the loss occurred 7: List Certain Payments or Trans Within 1 year before you filed for bankruptcy or preclude any attorneys, bankruptcy petition prepared No Yes. Fill in the details. Person Who Was Paid	Describe any insurance coverage for the loss Include the amount that insurance has paid. List pending insurance claims on line 33 of Schedule AVB: Property. ffers cy, did you or anyone else acting on your behalf pay or tran paring a bankruptcy petition? parers, or credit counseling agencies for services required in your behalf pay or tran parers, or credit counseling agencies for services required in your behalf pay or tran parers, or credit counseling agencies for services required in your behalf pay or transparent pay	sfer any property to our bankruptcy.	s
gambling? No Yes. Fill in the details. Describe the property you lost and how the loss occurred 7: List Certain Payments or Trans Within 1 year before you filed for bankruptcy or preclude any attorneys, bankruptcy petition prepared No Yes. Fill in the details. Person Who Was Paid	Describe any insurance coverage for the loss Include the amount that insurance has paid. List pending insurance claims on line 33 of Schedule AVB: Property. ffers cy, did you or anyone else acting on your behalf pay or tran paring a bankruptcy petition? parers, or credit counseling agencies for services required in your behalf pay or tran parers, or credit counseling agencies for services required in your behalf pay or tran parers, or credit counseling agencies for services required in your behalf pay or transparent pay	sfer any property to our bankruptcy.	s
gambling? No Yes. Fill in the details. Describe the property you lost and how the loss occurred 7: List Certain Payments or Trans Within 1 year before you filed for bankruptcy or preclude any attorneys, bankruptcy petition prepared No Yes. Fill in the details. Person Who Was Paid	Describe any insurance coverage for the loss Include the amount that insurance has paid. List pending insurance claims on line 33 of Schedule AVB: Property. ffers cy, did you or anyone else acting on your behalf pay or tran paring a bankruptcy petition? parers, or credit counseling agencies for services required in your behalf pay or tran parers, or credit counseling agencies for services required in your behalf pay or tran parers, or credit counseling agencies for services required in your behalf pay or transparent pay	sfer any property to our bankruptcy.	s
gambling? No Yes. Fill in the details. Describe the property you lost and how the loss occurred 7: List Certain Payments or Trans Tithin 1 year before you filed for bankruptcy or preclude any attorneys, bankruptcy petition preperiod about seeking bankruptcy petition preperiod No Yes. Fill in the details. Person Who Was Paid Number	Describe any insurance coverage for the loss Include the amount that insurance has paid. List pending insurance claims on line 33 of Schedule AVB: Property. ffers cy, did you or anyone else acting on your behalf pay or tran paring a bankruptcy petition? parers, or credit counseling agencies for services required in your behalf pay or tran parers, or credit counseling agencies for services required in your behalf pay or tran parers, or credit counseling agencies for services required in your behalf pay or transparent pay	sfer any property to our bankruptcy.	s
gambling? No Yes. Fill in the details. Describe the property you lost and how the loss occurred 7: List Certain Payments or Trans Tithin 1 year before you filed for bankruptcy or preclude any attorneys, bankruptcy petition preperiod about seeking bankruptcy petition preperiod No Yes. Fill in the details. Person Who Was Paid Number	Describe any insurance coverage for the loss Include the amount that insurance has paid. List pending insurance claims on line 33 of Schedule AVB: Property. ffers cy, did you or anyone else acting on your behalf pay or tran paring a bankruptcy petition? parers, or credit counseling agencies for services required in your behalf pay or tran parers, or credit counseling agencies for services required in your behalf pay or tran parers, or credit counseling agencies for services required in your behalf pay or transparent pay	sfer any property to our bankruptcy.	s

Statement of Financial Affairs for Individuals Filing for Bankruptcy

	Description and value of any property t	ransferred	Date payment or transfer was made	Amount of payment
Person Who Was Paid				\$
Number Street	- ·			
	*			\$
City State ZIP Code	-			
·				
Email or website address				
Person Who Made the Payment, if Not You				
	-	-		
No Yes. Fill in the details.	Description and value of any property (transferred	Date payment or	Amount of new
	Description and value of any property t	Iransierred	Date payment or transfer was made	Amount of pay
Person Who Was Paid	-		-	. A
· .	-			\$
Number Street				¢
				<u>م</u>
City State ZIP Code Vithin 2 years before you filed for bankru	- ptcy, did you sell, trade, or otherwise	transfer any property	to anyone, other than	property
	r business or financial affairs? made as security (such as the granting of			
Vithin 2 years before you filed for bankru ransferred in the ordinary course of your include both outright transfers and transfers on ot include gifts and transfers that you ha	r business or financial affairs? made as security (such as the granting of	of a security interest or i	nortgage on your prop y or payments received	erty).
Vithin 2 years before you filed for bankru ransferred in the ordinary course of your include both outright transfers and transfers on ot include gifts and transfers that you ha	r business or financial affairs? made as security (such as the granting of ave already listed on this statement. Description and value of property	of a security interest or i Describe any propert	nortgage on your prop y or payments received	erty). Date transf
Vithin 2 years before you filed for bankru ransferred in the ordinary course of your include both outright transfers and transfers include gifts and transfers that you had No Yes. Fill in the details.	r business or financial affairs? made as security (such as the granting of ave already listed on this statement. Description and value of property	of a security interest or i Describe any propert	nortgage on your prop y or payments received	erty). Date transf
Vithin 2 years before you filed for bankru ransferred in the ordinary course of your include both outright transfers and transfers on ot include gifts and transfers that you ha No Yes. Fill in the details.	r business or financial affairs? made as security (such as the granting of ave already listed on this statement. Description and value of property	of a security interest or i Describe any propert	nortgage on your prop y or payments received	erty). Date transf
Vithin 2 years before you filed for bankru ransferred in the ordinary course of your include both outright transfers and transfers include gifts and transfers that you had No Yes. Fill in the details.	r business or financial affairs? made as security (such as the granting of ave already listed on this statement. Description and value of property	of a security interest or i Describe any propert	nortgage on your prop y or payments received	erty). Date transf
Within 2 years before you filed for bankru transferred in the ordinary course of your nclude both outright transfers and transfers bo not include gifts and transfers that you had No Yes. Fill in the details. Person Who Received Transfer Number Street	r business or financial affairs? made as security (such as the granting of ave already listed on this statement. Description and value of property	of a security interest or i Describe any propert	nortgage on your prop y or payments received	erty). Date transf
Vithin 2 years before you filed for bankru ransferred in the ordinary course of your include both outright transfers and transfers include gifts and transfers that you had No Yes. Fill in the details.	r business or financial affairs? made as security (such as the granting of ave already listed on this statement. Description and value of property	of a security interest or i Describe any propert	nortgage on your prop y or payments received	erty). Date transf
Within 2 years before you filed for bankru transferred in the ordinary course of your nclude both outright transfers and transfers bo not include gifts and transfers that you had No Yes. Fill in the details. Person Who Received Transfer Number Street	r business or financial affairs? made as security (such as the granting of ave already listed on this statement. Description and value of property	of a security interest or i Describe any propert	nortgage on your prop y or payments received	erty). Date transf
Vithin 2 years before you filed for bankru ransferred in the ordinary course of your nclude both outright transfers and transfers bo not include gifts and transfers that you had No Yes. Fill in the details. Person Who Received Transfer Number Street City State ZIP Code Person's relationship to you	r business or financial affairs? made as security (such as the granting of ave already listed on this statement. Description and value of property	of a security interest or i Describe any propert	nortgage on your prop y or payments received	erty). Date transf
Within 2 years before you filed for bankru ransferred in the ordinary course of your include both outright transfers and transfers to not include gifts and transfers that you hat No No Yes. Fill in the details. Person Who Received Transfer Number Street City State ZIP Code	r business or financial affairs? made as security (such as the granting of ave already listed on this statement. Description and value of property	of a security interest or i Describe any propert	nortgage on your prop y or payments received	erty). Date transf
Vithin 2 years before you filed for bankru ransferred in the ordinary course of your nclude both outright transfers and transfers bo not include gifts and transfers that you had No Yes. Fill in the details. Person Who Received Transfer Number Street City State ZIP Code Person's relationship to you	r business or financial affairs? made as security (such as the granting of ave already listed on this statement. Description and value of property	of a security interest or i Describe any propert	nortgage on your prop y or payments received	erty). Date transf
Within 2 years before you filed for bankru ransferred in the ordinary course of your include both outright transfers and transfers to not include gifts and transfers that you hat No No Yes. Fill in the details. Person Who Received Transfer Number Street City State Person's relationship to you Person Who Received Transfer	r business or financial affairs? made as security (such as the granting of ave already listed on this statement. Description and value of property	of a security interest or i Describe any propert	nortgage on your prop y or payments received	erty). Date transf
Within 2 years before you filed for bankru ransferred in the ordinary course of your include both outright transfers and transfers to not include gifts and transfers that you hat No No Yes. Fill in the details. Person Who Received Transfer Number Street City State Person's relationship to you Person Who Received Transfer	r business or financial affairs? made as security (such as the granting of ave already listed on this statement. Description and value of property	of a security interest or i Describe any propert	nortgage on your prop y or payments received	erty). Date transf

Statement of Financial Affairs for Individuals Filing for Bankruptcy

	Name			
fithin 10 years before you filed for bankru re a beneficiary? (These are often called a	ptcy, did you transfer any property sset-protection devices.)	/ to a self-settled trust or	similar device of wh	ich you
O No				
Yes. Fill in the details.				
	Description and value of the proper	rty transferred		Date transfer was made
			STATE AND AND A STATE OF A STATE	
Name of trust				
				4949
	-			
	Automation			
		Davies and Phanese		
8: List Certain Financial Account	ts, Instruments, Safe Deposit	Boxes, and Storage	Units	
Vithin 1 year before you filed for bankrup	tcv, were any financial accounts or	r instruments held in you	r name, or for your b	enefit,
		-		
losed, sold, moved, or transferred?			In hamba and the set	
nclude checking, savings, money market	, or other financial accounts; certil	icates of deposit; shares	s in banks, credit uni	ons,
rokerage houses, pension funds, cooper	atives, associations, and other fin	ancial institutions.		
No				
Yes. Fill in the details.				
	Last 4 digits of account number	Type of account or	Date account was	Last balance bef
		instrument	closed, sold, moved,	closing or transf
· .			or transferred	
Name of Financial Institution		-		
Name of Financial Institution	xxxx	Checking	·	\$
Number Street	•	Savings		
Number Street		Money market		
	•			
	•	Brokerage		
City State ZIP Code			na a ga concernante a su que concernante a su a ga a ga a concernante a su a ga a ga a concernante a su a ga a concernante a su a concernante a	
City State ZIP Code		Brokerage		n ananononononononononononononononononon
		Brokerage		saanoon oo aanoo oo a
City State ZIP Code Name of Financial Institution		Brokerage		sa ten en un
Name of Financial Institution	XXX	Brokerage Other Checking Savings		sa oo
		 Brokerage Other Checking Savings Money market 		\$
Name of Financial Institution		 Brokerage Other Checking Savings Money market Brokerage 	~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~	\$
Name of Financial Institution Number Street		 Brokerage Other Checking Savings Money market 		saannon varaanna saan ah
Name of Financial Institution		 Brokerage Other Checking Savings Money market Brokerage 		s
Name of Financial Institution Number Street City State ZIP Code	-	 Brokerage Other Checking Savings Money market Brokerage Other 		\$
Name of Financial Institution Number Street	-	 Brokerage Other Checking Savings Money market Brokerage Other 	or other depository	\$ for
Name of Financial Institution Number Street City State ZIP Code Do you now have, or did you have within	-	 Brokerage Other Checking Savings Money market Brokerage Other 	or other depository	\$ for
Name of Financial Institution Number Street City State ZIP Code Do you now have, or did you have within securities, cash, or other valuables?	-	 Brokerage Other Checking Savings Money market Brokerage Other 	c or other depository	\$ for
Name of Financial Institution Number Street City State ZIP Code Do you now have, or did you have within securities, cash, or other valuables? No	-	 Brokerage Other Checking Savings Money market Brokerage Other 		Do you s
Name of Financial Institution Number Street City State ZIP Code Do you now have, or did you have within securities, cash, or other valuables? No	- 1 year before you filed for bankrup	 Brokerage Other Checking Savings Money market Brokerage Other 		\$ for Do you s have it?
Name of Financial Institution Number Street City State ZIP Code Do you now have, or did you have within securities, cash, or other valuables? No	- 1 year before you filed for bankrup	 Brokerage Other Checking Savings Money market Brokerage Other 		Do you s
Name of Financial Institution Number Street City State ZIP Code Do you now have, or did you have within securities, cash, or other valuables? No No Yes. Fill in the details.	1 year before you filed for bankrup Who else had access to it?	 Brokerage Other Checking Savings Money market Brokerage Other 		Do you s have it?
Name of Financial Institution Number Street City State ZIP Code Do you now have, or did you have within securities, cash, or other valuables? No	- 1 year before you filed for bankrup	 Brokerage Other Checking Savings Money market Brokerage Other 		Do you s have it?
Name of Financial Institution Number Street City State ZIP Code Do you now have, or did you have within securities, cash, or other valuables? No Yes. Fill in the details.	1 year before you filed for bankrup Who else had access to it? Name	 Brokerage Other Checking Savings Money market Brokerage Other 		Do you s have it?
Name of Financial Institution Number Street City State ZIP Code Do you now have, or did you have within securities, cash, or other valuables? No No Yes. Fill in the details.	1 year before you filed for bankrup Who else had access to it?	 Brokerage Other Checking Savings Money market Brokerage Other 		Do you s have it?
Name of Financial Institution Number Street City State ZIP Code Do you now have, or did you have within securities, cash, or other valuables? No Yes. Fill in the details.	1 year before you filed for bankrup Who else had access to it? Name Number Street	 Brokerage Other Checking Savings Money market Brokerage Other 		Do you s have it?
Name of Financial Institution Number Street City State ZIP Code Do you now have, or did you have within securities, cash, or other valuables? No Yes. Fill in the details.	1 year before you filed for bankrup Who else had access to it? Name	 Brokerage Other Checking Savings Money market Brokerage Other 		Do you s have it?

Debtor 1	First Name Middle Name La	st Name Ca	ase number (if known)	
22. Have	e vou stored property in a storage unit	or place other than your home within 1 year	r before you filed for bankruptcy?	
	Yes. Fill in the details.		Describe the contents	Do you s
		Who else has or had access to it?	Describe the coments	have it?
-	Name of Storage Facility	Name	-	🖸 Yes
		· · · · · · · · · · · · · · · · · · ·	_	An and another an
	Number Street	Number Street		international data
-		CityState ZIP Code	-	
	City State ZIP Code			ų ₁
-	City City			-
Part	9: Identify Property You Hold	or Control for Someone Else		.*
		someone else owns? Include any property y	you borrowed from are storing for	
	you hold or control any property that the hold in trust for someone.	someone else owns r include any property y	you borrowed from, are storing for,	
2 I	No			
	Yes. Fill in the details.			
		Where is the property?	Describe the property	Value
	Owner's Name			\$
	Number Street	Number Street		-
		City State ZIP Code		
	City State ZIP Code	City State ZIP Code		
	10: Give Details About Environ e purpose of Part 10, the following det	initions apply:		
For th En ha inc Sit it c	10: Give Details About Environ the purpose of Part 10, the following det invironmental law means any federal, st zardous or toxic substances, wastes, cluding statutes or regulations control te means any location, facility, or proper or used to own, operate, or utilize it, in	nmental Information initions apply: ate, or local statute or regulation concernin or material into the air, land, soil, surface wa ling the cleanup of these substances, waste erty as defined under any environmental law cluding disposal sites.	ater, groundwater, or other medium, es, or material. v, whether you now own, operate, or uti	
For th En ha inc Sit it c Ha	10: Give Details About Environ the purpose of Part 10, the following det avironmental law means any federal, st zardous or toxic substances, wastes, cluding statutes or regulations control the means any location, facility, or proper or used to own, operate, or utilize it, in azardous material means anything an e	nmental Information initions apply: ate, or local statute or regulation concerning or material into the air, land, soil, surface wa ling the cleanup of these substances, waste erty as defined under any environmental law cluding disposal sites. environmental law defines as a hazardous w	ater, groundwater, or other medium, es, or material. v, whether you now own, operate, or uti	
For th En ha inc Siti it c Ha su	10: Give Details About Environ the purpose of Part 10, the following det avironmental law means any federal, st zardous or toxic substances, wastes, cluding statutes or regulations control te means any location, facility, or propor or used to own, operate, or utilize it, in azardous material means anything an e- bistance, hazardous material, pollutant	nmental Information initions apply: ate, or local statute or regulation concernin or material into the air, land, soil, surface wi ling the cleanup of these substances, waste erty as defined under any environmental law cluding disposal sites. Invironmental law defines as a hazardous w t, contaminant, or similar term.	ater, groundwater, or other medium, es, or material. v, whether you now own, operate, or uti vaste, hazardous substance, toxic	
For th En ha inc Siti it c Ha su	10: Give Details About Environ the purpose of Part 10, the following det avironmental law means any federal, st zardous or toxic substances, wastes, cluding statutes or regulations control te means any location, facility, or propor or used to own, operate, or utilize it, in azardous material means anything an e- bistance, hazardous material, pollutant	nmental Information initions apply: ate, or local statute or regulation concerning or material into the air, land, soil, surface wa ling the cleanup of these substances, waste erty as defined under any environmental law cluding disposal sites. environmental law defines as a hazardous w	ater, groundwater, or other medium, es, or material. v, whether you now own, operate, or uti vaste, hazardous substance, toxic	
For th En ha inc Sit it c Ha su Report	10: Give Details About Environ the purpose of Part 10, the following det invironmental law means any federal, st zardous or toxic substances, wastes, cluding statutes or regulations control to means any location, facility, or propor or used to own, operate, or utilize it, in azardous material means anything an el abstance, hazardous material, pollutant rt all notices, releases, and proceeding	initions apply: ate, or local statute or regulation concernin or material into the air, land, soil, surface wi ling the cleanup of these substances, waste erty as defined under any environmental law cluding disposal sites. Invironmental law defines as a hazardous w t, contaminant, or similar term.	ater, groundwater, or other medium, es, or material. v, whether you now own, operate, or uti vaste, hazardous substance, toxic they occurred.	ilize
For th For th ha inc Sit it c Ha su Report 24. Ha	10: Give Details About Environ the purpose of Part 10, the following det invironmental law means any federal, st zardous or toxic substances, wastes, cluding statutes or regulations control the means any location, facility, or proper or used to own, operate, or utilize it, inv azardous material means anything an e- hibstance, hazardous material, pollutant rt all notices, releases, and proceeding as any governmental unit notified you th	nmental Information initions apply: ate, or local statute or regulation concernin or material into the air, land, soil, surface wi ling the cleanup of these substances, waste erty as defined under any environmental law cluding disposal sites. Invironmental law defines as a hazardous w t, contaminant, or similar term.	ater, groundwater, or other medium, es, or material. v, whether you now own, operate, or uti vaste, hazardous substance, toxic they occurred.	ilize
For th En ha ind Sit it c Ha su Report 24. Ha	10: Give Details About Environ the purpose of Part 10, the following det avironmental law means any federal, st zardous or toxic substances, wastes, cluding statutes or regulations control the means any location, facility, or proper or used to own, operate, or utilize it, in azardous material means anything an el abstance, hazardous material, pollutant et all notices, releases, and proceeding as any governmental unit notified you the No	initions apply: ate, or local statute or regulation concernin or material into the air, land, soil, surface wi ling the cleanup of these substances, waste erty as defined under any environmental law cluding disposal sites. Invironmental law defines as a hazardous w t, contaminant, or similar term.	ater, groundwater, or other medium, es, or material. v, whether you now own, operate, or uti vaste, hazardous substance, toxic they occurred.	ilize
For th For th ha inc Sit it c Ha su Report 24. Ha	10: Give Details About Environ the purpose of Part 10, the following det avironmental law means any federal, st zardous or toxic substances, wastes, cluding statutes or regulations control the means any location, facility, or proper or used to own, operate, or utilize it, in azardous material means anything an el abstance, hazardous material, pollutant et all notices, releases, and proceeding as any governmental unit notified you the No	initions apply: ate, or local statute or regulation concerning or material into the air, land, soil, surface we ling the cleanup of these substances, waste erty as defined under any environmental law cluding disposal sites. Invironmental law defines as a hazardous we t, contaminant, or similar term. Is that you know about, regardless of when hat you may be liable or potentially liable un	ater, groundwater, or other medium, es, or materiał. w, whether you now own, operate, or uti vaste, hazardous substance, toxic they occurred. der or in violation of an environmental	ilize law?
For th En ha ind Sit it c Ha su Report 24. Ha	10: Give Details About Environ the purpose of Part 10, the following det avironmental law means any federal, st zardous or toxic substances, wastes, cluding statutes or regulations control the means any location, facility, or proper or used to own, operate, or utilize it, in azardous material means anything an el abstance, hazardous material, pollutant et all notices, releases, and proceeding as any governmental unit notified you the No	initions apply: ate, or local statute or regulation concerning or material into the air, land, soil, surface we ling the cleanup of these substances, waste erty as defined under any environmental law cluding disposal sites. environmental law defines as a hazardous we t, contaminant, or similar term. as that you know about, regardless of when hat you may be liable or potentially liable un	ater, groundwater, or other medium, es, or material. v, whether you now own, operate, or uti vaste, hazardous substance, toxic they occurred.	ilize law?
For th En ha ind Sit it c Ha su Report 24. Ha	10: Give Details About Environ the purpose of Part 10, the following det avironmental law means any federal, st zardous or toxic substances, wastes, cluding statutes or regulations control the means any location, facility, or proper or used to own, operate, or utilize it, in azardous material means anything an el abstance, hazardous material, pollutant et all notices, releases, and proceeding as any governmental unit notified you the No	initions apply: ate, or local statute or regulation concerning or material into the air, land, soil, surface we ling the cleanup of these substances, waste erty as defined under any environmental law cluding disposal sites. Invironmental law defines as a hazardous we t, contaminant, or similar term. Is that you know about, regardless of when hat you may be liable or potentially liable un	ater, groundwater, or other medium, es, or materiał. w, whether you now own, operate, or uti vaste, hazardous substance, toxic they occurred. der or in violation of an environmental	ilize law?
For th En ha ind Sit it c Ha su Report 24. Ha	10: Give Details About Environ the purpose of Part 10, the following det avironmental law means any federal, st zardous or toxic substances, wastes, cluding statutes or regulations control the means any location, facility, or proper or used to own, operate, or utilize it, in azardous material means anything an el abstance, hazardous material, pollutant et all notices, releases, and proceeding as any governmental unit notified you the No	initions apply: ate, or local statute or regulation concerning or material into the air, land, soil, surface we ling the cleanup of these substances, waste erty as defined under any environmental law cluding disposal sites. Invironmental law defines as a hazardous we t, contaminant, or similar term. Is that you know about, regardless of when hat you may be liable or potentially liable un	ater, groundwater, or other medium, es, or materiał. w, whether you now own, operate, or uti vaste, hazardous substance, toxic they occurred. der or in violation of an environmental	ilize
For th En ha ind Sit it c Ha su Report 24. Ha	10: Give Details About Environ the purpose of Part 10, the following det avironmental law means any federal, st zardous or toxic substances, wastes, cluding statutes or regulations control the means any location, facility, or proper or used to own, operate, or utilize it, in azardous material means anything an el abstance, hazardous material, pollutant et all notices, releases, and proceeding as any governmental unit notified you the No Yes. Fill in the details.	initions apply: ate, or local statute or regulation concerning or material into the air, land, soil, surface wa ling the cleanup of these substances, waste erty as defined under any environmental law cluding disposal sites. Invironmental law defines as a hazardous wa contaminant, or similar term. Its that you know about, regardless of when hat you may be liable or potentially liable un Governmental unit	ater, groundwater, or other medium, es, or materiał. w, whether you now own, operate, or uti vaste, hazardous substance, toxic they occurred. der or in violation of an environmental	ilize law?
For th En ha ind Sit it c Ha su Report 24. Ha	10: Give Details About Environ the purpose of Part 10, the following det avironmental law means any federal, st zardous or toxic substances, wastes, cluding statutes or regulations control the means any location, facility, or proper or used to own, operate, or utilize it, in azardous material means anything an el abstance, hazardous material, pollutant et all notices, releases, and proceeding as any governmental unit notified you the No Yes. Fill in the details.	initions apply: ate, or local statute or regulation concernin or material into the air, land, soil, surface wa ling the cleanup of these substances, waste erty as defined under any environmental law cluding disposal sites. Invironmental law defines as a hazardous w t, contaminant, or similar term. Its that you know about, regardless of when hat you may be liable or potentially liable un Governmental unit	ater, groundwater, or other medium, es, or materiał. w, whether you now own, operate, or uti vaste, hazardous substance, toxic they occurred. der or in violation of an environmental	ilize law?
For th En ha ind Sit it c Ha su Report 24. Ha	10: Give Details About Environ the purpose of Part 10, the following det avironmental law means any federal, st zardous or toxic substances, wastes, cluding statutes or regulations control the means any location, facility, or proper or used to own, operate, or utilize it, in azardous material means anything an el abstance, hazardous material, pollutant et all notices, releases, and proceeding as any governmental unit notified you the No Yes. Fill in the details.	initions apply: ate, or local statute or regulation concerning or material into the air, land, soil, surface wa ling the cleanup of these substances, waste erty as defined under any environmental law cluding disposal sites. Invironmental law defines as a hazardous wa contaminant, or similar term. Its that you know about, regardless of when hat you may be liable or potentially liable un Governmental unit	ater, groundwater, or other medium, es, or materiał. w, whether you now own, operate, or uti vaste, hazardous substance, toxic they occurred. der or in violation of an environmental	ilize law?
For th En ha ind Sit it c Ha su Report 24. Ha	Io: Give Details About Environmental law means any federal, streadous or toxic substances, wastes, cluding statutes or regulations control to means any location, facility, or proporrused to own, operate, or utilize it, in exardous material means anything an existance, hazardous material, pollutant all notices, releases, and proceeding is any governmental unit notified you to No Yes. Fill in the details.	initions apply: ate, or local statute or regulation concerning or material into the air, land, soil, surface wa ling the cleanup of these substances, waste erty as defined under any environmental law cluding disposal sites. Invironmental law defines as a hazardous wa contaminant, or similar term. Its that you know about, regardless of when hat you may be liable or potentially liable un Governmental unit Enviro Governmental unit Enviro	ater, groundwater, or other medium, es, or materiał. w, whether you now own, operate, or uti vaste, hazardous substance, toxic they occurred. der or in violation of an environmental	ilize law?
For th En ha ind Sit it c Ha su Report 24. Ha	10: Give Details About Environ the purpose of Part 10, the following det avironmental law means any federal, st zardous or toxic substances, wastes, cluding statutes or regulations control the means any location, facility, or proper or used to own, operate, or utilize it, in azardous material means anything an el abstance, hazardous material, pollutant et all notices, releases, and proceeding as any governmental unit notified you the No Yes. Fill in the details.	initions apply: ate, or local statute or regulation concerning or material into the air, land, soil, surface wa ling the cleanup of these substances, waste erty as defined under any environmental law cluding disposal sites. Invironmental law defines as a hazardous wa contaminant, or similar term. Its that you know about, regardless of when hat you may be liable or potentially liable un Governmental unit Enviro Governmental unit Enviro	ater, groundwater, or other medium, es, or materiał. w, whether you now own, operate, or uti vaste, hazardous substance, toxic they occurred. der or in violation of an environmental	ilize law?
For th Enha ind Siti it (Image: Hasu Report 24. Ha	10: Give Details About Environmental law means any federal, stazardous or toxic substances, wastes, cluding statutes or regulations controllate means any location, facility, or proper used to own, operate, or utilize it, invazardous material means anything an elabstance, hazardous material, pollutant all notices, releases, and proceeding as any governmental unit notified you the Yes. Fill in the details. No Yes. Fill in the details. Name of site	initions apply: ate, or local statute or regulation concerning or material into the air, land, soil, surface wa ling the cleanup of these substances, waste erty as defined under any environmental law cluding disposal sites. Invironmental law defines as a hazardous wa contaminant, or similar term. Its that you know about, regardless of when hat you may be liable or potentially liable un Governmental unit Enviro Governmental unit Enviro	ater, groundwater, or other medium, es, or materiał. w, whether you now own, operate, or uti raste, hazardous substance, toxic they occurred. Inder or in violation of an environmental commental law, if you know it	ilize law?

ve you notified any governmental ur	nit of any release of hazardous material?		
No			
Yes. Fill in the details.		Section 2014	
	Governmental unit	Environmental law, if you know it	Date of notic
			-
Name of site	Governmental unit		
Number Street	Number Street		
	· · · · · · · · · · · · · · · · · · ·		
	City State ZIP Code		
City State ZIP Cod	le		
ve you been a party in any judicial o	r administrative proceeding under any er	nvironmental law? Include settlements an	a orders.
No			
Yes. Fill in the details.			Status of t
	Court or agency	Nature of the case	case
Case title			-
	Court Name	-	Pendir
			🔲 On app
	Number Street	-	Conclu
Case number	City State ZIP Code	-	
11: Give Details About You	Business or Connections to Any B	Business	ousiness?
 Give Details About Your thin 4 years before you filed for ban A sole proprietor or self-emplo A member of a limited liability of A partner in a partnership 	r Business or Connections to Any B kruptcy, did you own a business or have yed in a trade, profession, or other activi company (LLC) or limited liability partner	Business any of the following connections to any b ity, either full-time or part-time	ousiness?
 Give Details About Your thin 4 years before you filed for ban A sole proprietor or self-emplo A member of a limited liability A partner in a partnership An officer, director, or managing 	r Business or Connections to Any B kruptcy, did you own a business or have yed in a trade, profession, or other activi company (LLC) or limited liability partner ng executive of a corporation	Business any of the following connections to any b ity, either full-time or part-time ship (LLP)	pusiness?
 Give Details About Your thin 4 years before you filed for ban A sole proprietor or self-emplo A member of a limited liability A partner in a partnership An officer, director, or managing 	r Business or Connections to Any B kruptcy, did you own a business or have yed in a trade, profession, or other activi company (LLC) or limited liability partner	Business any of the following connections to any b ity, either full-time or part-time ship (LLP)	ousiness?
 Give Details About Your thin 4 years before you filed for ban A sole proprietor or self-emplo A member of a limited liability A partner in a partnership An officer, director, or managin An owner of at least 5% of the No. None of the above applies. Go 	r Business or Connections to Any B kruptcy, did you own a business or have yed in a trade, profession, or other activi company (LLC) or limited liability partner ng executive of a corporation voting or equity securities of a corporation to Part 12.	Business e any of the following connections to any b ity, either full-time or part-time ship (LLP) on	ousiness?
 Give Details About Your thin 4 years before you filed for ban A sole proprietor or self-emplo A member of a limited liability A partner in a partnership An officer, director, or managin An owner of at least 5% of the No. None of the above applies. Go 	r Business or Connections to Any B kruptcy, did you own a business or have yed in a trade, profession, or other activi company (LLC) or limited liability partner ng executive of a corporation voting or equity securities of a corporation to Part 12. d fill in the details below for each busine	Business any of the following connections to any b ity, either full-time or part-time ship (LLP) on	
 11: Give Details About Your thin 4 years before you filed for ban A sole proprietor or self-emplo A member of a limited liability A partner in a partnership An officer, director, or managin An owner of at least 5% of the No. None of the above applies. Go 	r Business or Connections to Any B kruptcy, did you own a business or have yed in a trade, profession, or other activi company (LLC) or limited liability partner ng executive of a corporation voting or equity securities of a corporation to Part 12.	Business any of the following connections to any b ity, either full-time or part-time rship (LLP) on ess. Employer Identification nur	nber
 Give Details About Your thin 4 years before you filed for ban A sole proprietor or self-emplo A member of a limited liability A partner in a partnership An officer, director, or managin An owner of at least 5% of the No. None of the above applies. Go 	r Business or Connections to Any B kruptcy, did you own a business or have yed in a trade, profession, or other activi company (LLC) or limited liability partner ng executive of a corporation voting or equity securities of a corporation to Part 12. d fill in the details below for each busine	Business any of the following connections to any b ity, either full-time or part-time ship (LLP) on	nber
 Give Details About Your thin 4 years before you filed for ban A sole proprietor or self-emplo A member of a limited liability A partner in a partnership An officer, director, or managin An owner of at least 5% of the No. None of the above applies. Go Yes. Check all that apply above an 	r Business or Connections to Any B kruptcy, did you own a business or have yed in a trade, profession, or other activi company (LLC) or limited liability partner ng executive of a corporation voting or equity securities of a corporation to Part 12. d fill in the details below for each busine	Business any of the following connections to any b ity, either full-time or part-time rship (LLP) on ess. Employer Identification nur	nber
 Give Details About Your thin 4 years before you filed for ban A sole proprietor or self-emplo A member of a limited liability A partner in a partnership An officer, director, or managin An owner of at least 5% of the No. None of the above applies. Go Yes. Check all that apply above an 	r Business or Connections to Any B kruptcy, did you own a business or have yed in a trade, profession, or other activi company (LLC) or limited liability partner ing executive of a corporation voting or equity securities of a corporation to Part 12. d fill in the details below for each busine Describe the nature of the business	Business any of the following connections to any b ity, either full-time or part-time ship (LLP) on ess. Employer Identification nur Do not include Social Secu EIN:	nber
Give Details About Your thin 4 years before you filed for ban A sole proprietor or self-emplo A member of a limited liability of A partner in a partnership An officer, director, or managin An owner of at least 5% of the No. None of the above applies. Go Yes. Check all that apply above an Business Name	r Business or Connections to Any B kruptcy, did you own a business or have yed in a trade, profession, or other activi company (LLC) or limited liability partner ng executive of a corporation voting or equity securities of a corporation to Part 12. d fill in the details below for each busine	Business any of the following connections to any b ity, either full-time or part-time rship (LLP) on ess. Employer Identification nur Do not include Social Secu	nber
Give Details About Your thin 4 years before you filed for ban A sole proprietor or self-emplo A member of a limited liability of A partner in a partnership An officer, director, or managin An owner of at least 5% of the No. None of the above applies. Go Yes. Check all that apply above an Business Name	r Business or Connections to Any B kruptcy, did you own a business or have yed in a trade, profession, or other activi company (LLC) or limited liability partner ing executive of a corporation voting or equity securities of a corporation to Part 12. d fill in the details below for each busine Describe the nature of the business	Business any of the following connections to any b ity, either full-time or part-time ship (LLP) on ess. Employer Identification nur Do not include Social Secu EIN:	nber rity number or ITI
Give Details About Your thin 4 years before you filed for ban A sole proprietor or self-emplo A member of a limited liability of A partner in a partnership An officer, director, or managin An owner of at least 5% of the No. None of the above applies. Go Yes. Check all that apply above an Business Name	r Business or Connections to Any B kruptcy, did you own a business or have yed in a trade, profession, or other activi company (LLC) or limited liability partner ing executive of a corporation voting or equity securities of a corporation to Part 12. Id fill in the details below for each busine Describe the nature of the business	Business any of the following connections to any bity, either full-time or part-time rship (LLP) on ess. Employer Identification num Do not include Social Secu EIN: Dates business existed	nber rity number or ITI
I1: Give Details About Your thin 4 years before you filed for ban A sole proprietor or self-emplo A member of a limited liability A partner in a partnership An officer, director, or managin An owner of at least 5% of the No. None of the above applies. Go Yes. Check all that apply above an Business Name Number Street	r Business or Connections to Any B kruptcy, did you own a business or have yed in a trade, profession, or other activi company (LLC) or limited liability partner ing executive of a corporation voting or equity securities of a corporation to Part 12. Id fill in the details below for each busine Describe the nature of the business	Business any of the following connections to any bity, either full-time or part-time ship (LLP) on sss. Employer Identification nur Do not include Social Secu EIN: Dates business existed From To Employer Identification nur	nber rity number or ITII
11: Give Details About Your thin 4 years before you filed for ban A sole proprietor or self-emplo A member of a limited liability A partner in a partnership An officer, director, or managin An owner of at least 5% of the No. None of the above applies. Go Yes. Check all that apply above an Business Name Number Street	r Business or Connections to Any B kruptcy, did you own a business or have yed in a trade, profession, or other activi company (LLC) or limited liability partner ing executive of a corporation voting or equity securities of a corporation to Part 12. Id fill in the details below for each busines Describe the nature of the business Name of accountant or bookkeeper	Business any of the following connections to any bity, either full-time or part-time ship (LLP) on sss. Employer Identification num Do not include Social Secu EIN: Dates business existed From To	nber rity number or ITII
Give Details About Your thin 4 years before you filed for ban A sole proprietor or self-emplo A member of a limited liability of A partner in a partnership An officer, director, or managin An owner of at least 5% of the No. None of the above applies. Go Yes. Check all that apply above an Business Name Number Street City State	r Business or Connections to Any B kruptcy, did you own a business or have yed in a trade, profession, or other activi company (LLC) or limited liability partner ing executive of a corporation voting or equity securities of a corporation to Part 12. Id fill in the details below for each busines Describe the nature of the business Name of accountant or bookkeeper	Business any of the following connections to any bity, either full-time or part-time rship (LLP) on bess. Employer Identification num Do not include Social Secu EIN: Dates business existed From To Employer Identification num Dates business existed From To	nber rity number or ITII
Give Details About Your thin 4 years before you filed for ban A sole proprietor or self-emplo A member of a limited liability of A partner in a partnership An officer, director, or managin An owner of at least 5% of the officer. No. None of the above applies. Go Yes. Check all that apply above an Business Name Number Street City State	r Business or Connections to Any B kruptcy, did you own a business or have yed in a trade, profession, or other activi- company (LLC) or limited liability partner ing executive of a corporation voting or equity securities of a corporation to Part 12. Id fill in the details below for each busines Describe the nature of the business Name of accountant or bookkeeper de	Business any of the following connections to any bity, either full-time or part-time rship (LLP) on ess. Employer Identification num Do not include Social Secu EIN: Dates business existed From To Employer Identification num Dates business existed From To Employer Identification num Do not include Social Secu EiN:	nber rity number or ITII
I1: Give Details About Your thin 4 years before you filed for ban A sole proprietor or self-emplo A member of a limited liability of A partner in a partnership An officer, director, or managin An owner of at least 5% of the No. None of the above applies. Go Yes. Check all that apply above an Business Name City State Business Name	r Business or Connections to Any B kruptcy, did you own a business or have yed in a trade, profession, or other activi company (LLC) or limited liability partner ing executive of a corporation voting or equity securities of a corporation to Part 12. Id fill in the details below for each busines Describe the nature of the business Name of accountant or bookkeeper	Business any of the following connections to any bity, either full-time or part-time rship (LLP) on bess. Employer Identification num Do not include Social Secu EIN: Dates business existed From To Employer Identification num Dates business existed From To	nber rity number or ITII

ŝ

			Describe the nature of the busines	5	Employer Identification number Do not include Social Security number or ITI
	Business Name				EIN:
	Number Street		Name of accountant or bookkeepe	r	Dates business existed
	City	State ZIP Code	,		From To
	Univ			• .	
			otcy, did you give a financial state	ment to anyon	e about your business? Include all financial
	tutions, creditors,	or other parties.			
I N I Y	lo ′es. Fill in the detai	is below		2	
	es. Fill in ule detai	is below.			
			Date issued		
	Name		MM / DD / YYYY		
	Number Street				
			1 .		
	City	State ZIP Code	•		
					-
12	2. Sign Below				
12	2: Sign Below				
12 ha	ave read the answe	ers on this <i>Stateme</i>	nt of Financial Affairs and any atta	ichments, and	I declare under penalty of perjury that the
ha	ave read the answe	correct. I understa	nd that making a false statement,	concealing pr	I declare under penalty of perjury that the operty, or obtaining money or property by frau
ha ans n c	ave read the answe swers are true and connection with a b	correct. I understa bankruptcy case ca	nt of Financial Affairs and any atta nd that making a false statement, n result in fines up to \$250,000, or	concealing pr	operty, or obtaining money or property by frau
ha ans n c	ave read the answe	correct. I understa bankruptcy case ca	nd that making a false statement,	concealing pr	operty, or obtaining money or property by frau
ha ans n c 18 (ave read the answe swers are true and connection with a t U.S.C. §§ 152, 134	correct. I understa bankruptcy case ca	nd that making a false statement, n result in fines up to \$250,000, or	concealing pr	operty, or obtaining money or property by frau
ha ans n c 18 (ave read the answe swers are true and connection with a t U.S.C. §§ 152, 134	correct. I understa bankruptcy case ca	nd that making a false statement, n result in fines up to \$250,000, or	concealing pr · imprisonmen	operty, or obtaining money or property by frau
ha ans n c 18	ave read the answe swers are true and connection with a t U.S.C. §§ 152, 134	correct. I understa bankruptcy case ca 1, 1519, and 3571.	nd that making a false statement, n result in fines up to \$250,000, or	concealing pr · imprisonmen	operty, or obtaining money or property by frau
ha ans n c 18	ave read the answe swers are true and connection with a b U.S.C. §§ 152, 134	correct. I understa bankruptcy case ca 1, 1519, and 3571.	nd that making a false statement, n result in fines up to \$250,000, or	concealing pr · imprisonmen	operty, or obtaining money or property by frau
ha ans n c 18 (ave read the answe swers are true and connection with a b U.S.C. §§ 152, 1341 	correct. I understa bankruptcy case ca 1, 1519, and 3571.	nd that making a false statement, n result in fines up to \$250,000, or Signature of Det Date	concealing pr imprisonmen	operty, or obtaining money or property by frau t for up to 20 years, or both.
	ave read the answe swers are true and connection with a b U.S.C. §§ 152, 1341 	correct. I understa bankruptcy case ca 1, 1519, and 3571.	nd that making a false statement, n result in fines up to \$250,000, or Signature of Det Date	concealing pr imprisonmen	operty, or obtaining money or property by frau
ha ans n c 18 (X	ave read the answe swers are true and connection with a b U.S.C. §§ 152, 134 	correct. I understa bankruptcy case ca 1, 1519, and 3571.	nd that making a false statement, n result in fines up to \$250,000, or Signature of Det Date	concealing pr imprisonmen	operty, or obtaining money or property by frau t for up to 20 years, or both.
	ave read the answe swers are true and connection with a b U.S.C. §§ 152, 1341 	correct. I understa bankruptcy case ca 1, 1519, and 3571.	nd that making a false statement, n result in fines up to \$250,000, or Signature of Det Date	concealing pr imprisonmen	operty, or obtaining money or property by frau t for up to 20 years, or both.
	ave read the answe swers are true and connection with a b U.S.C. §§ 152, 1341 Signature of Debtor Date You attach addition	correct. I understa bankruptcy case ca 1, 1519, and 3571.	nd that making a false statement, n result in fines up to \$250,000, or Signature of Det Date	concealing pr imprisonmen	operty, or obtaining money or property by frau t for up to 20 years, or both.
	ave read the answe swers are true and connection with a b U.S.C. §§ 152, 1341 Signature of Debtor Date You attach addition No Yes	correct. I understa bankruptcy case ca 1, 1519, and 3571.	nd that making a false statement, n result in fines up to \$250,000, or Signature of Det Date Statement of Financial Affairs for	concealing pr imprisonmen ntor 2 Individuals Fil	operty, or obtaining money or property by frau t for up to 20 years, or both.
Did	ave read the answe swers are true and connection with a b U.S.C. §§ 152, 134 Signature of Debtor Date I you attach addition No Yes	correct. I understa bankruptcy case ca 1, 1519, and 3571.	nd that making a false statement, n result in fines up to \$250,000, or Signature of Det Date	concealing pr imprisonmen ntor 2 Individuals Fil	operty, or obtaining money or property by frau t for up to 20 years, or both.
Did	ave read the answe swers are true and connection with a b U.S.C. §§ 152, 134 Signature of Debtor Date I you attach addition No Yes	correct. I understa bankruptcy case ca 1, 1519, and 3571.	nd that making a false statement, n result in fines up to \$250,000, or Signature of Det Date Statement of Financial Affairs for	concealing pr imprisonmen notor 2 Individuals Fil	operty, or obtaining money or property by frau t for up to 20 years, or both. <i>ing for Bankruptcy</i> (Official Form 107)? tcy forms?
Did	ave read the answe swers are true and connection with a b U.S.C. §§ 152, 134 Signature of Debtor Date I you attach addition No Yes	correct. I understa bankruptcy case ca 1, 1519, and 3571.	nd that making a false statement, n result in fines up to \$250,000, or Signature of Det Date Statement of Financial Affairs for	concealing pr imprisonmen notor 2 Individuals Fil	operty, or obtaining money or property by frau t for up to 20 years, or both.
Did	ave read the answe swers are true and connection with a b U.S.C. §§ 152, 134 Signature of Debtor Date I you attach addition No Yes	correct. I understa bankruptcy case ca 1, 1519, and 3571.	nd that making a false statement, n result in fines up to \$250,000, or Signature of Det Date Statement of Financial Affairs for	concealing pr imprisonmen notor 2 Individuals Fil	operty, or obtaining money or property by frau t for up to 20 years, or both. <i>ing for Bankruptcy</i> (Official Form 107)? tcy forms? Attach the <i>Bankruptcy Petition Preparer's Notice</i> ,
Did	ave read the answe swers are true and connection with a b U.S.C. §§ 152, 134 Signature of Debtor Date I you attach addition No Yes	correct. I understa bankruptcy case ca 1, 1519, and 3571.	nd that making a false statement, n result in fines up to \$250,000, or Signature of Det Date Statement of Financial Affairs for	concealing pr imprisonmen notor 2 Individuals Fil	operty, or obtaining money or property by frau t for up to 20 years, or both. <i>ing for Bankruptcy</i> (Official Form 107)? tcy forms? Attach the <i>Bankruptcy Petition Preparer's Notice</i> ,
Did	ave read the answe swers are true and connection with a b U.S.C. §§ 152, 134 Signature of Debtor Date I you attach addition No Yes	correct. I understa bankruptcy case ca 1, 1519, and 3571.	nd that making a false statement, n result in fines up to \$250,000, or Signature of Det Date Statement of Financial Affairs for	concealing pr imprisonmen notor 2 Individuals Fil	operty, or obtaining money or property by frau t for up to 20 years, or both. <i>ing for Bankruptcy</i> (Official Form 107)? tcy forms? Attach the <i>Bankruptcy Petition Preparer's Notice</i> ,
	ave read the answe swers are true and connection with a b U.S.C. §§ 152, 134 Signature of Debtor Date I you attach addition No Yes	correct. I understa bankruptcy case ca 1, 1519, and 3571.	nd that making a false statement, n result in fines up to \$250,000, or Signature of Det Date Statement of Financial Affairs for	concealing pr imprisonmen notor 2 Individuals Fil	operty, or obtaining money or property by frau t for up to 20 years, or both. <i>ing for Bankruptcy</i> (Official Form 107)? tcy forms? Attach the <i>Bankruptcy Petition Preparer's Notice,</i> <i>Declaration, and Signature</i> (Official Form 119).

Fill in this inf	ormation to identify ye	our case:	
Debtor 1	First Name	Middle Name	Last Name
Debtor 2 (Spouse, if filing)	First Name	Middle Name	Last Name
United States E Case number	Bankruptcy Court for the: _	· ·	District of(State)
(if known)			

Official Form 108

Statement of Intention for Individuals Filing Under Chapter 7 12/15

If you are an individual filing under chapter 7, you must fill out this form if:

- creditors have claims secured by your property, or
- you have leased personal property and the lease has not expired.

You must file this form with the court within 30 days after you file your bankruptcy petition or by the date set for the meeting of creditors, whichever is earlier, unless the court extends the time for cause. You must also send copies to the creditors and lessors you list on the form.

If two married people are filing together in a joint case, both are equally responsible for supplying correct information.

Both debtors must sign and date the form.

Be as complete and accurate as possible. If more space is needed, attach a separate sheet to this form. On the top of any additional pages, write your name and case number (if known).

Part 1: List Your Creditors Who Have Secured Claims

1. For any creditors that you listed in Part 1 of Schedule D: Creditors Who Have Claims Secured by Property (Official Form 106D), fill in the information below.

	half of the state of the state state the state of the sta	
Identify the creditor and the property that is collateral	What do you intend to do with the property that secures a debt?	Did you claim the property as exempt on Schedule C?
Creditor's	Surrender the property.	D No
	Retain the property and redeem it.	Yes
Description of property securing debt:	Retain the property and enter into a Reaffirmation Agreement.	
securing debt.	Retain the property and [explain]:	-
Creditor's name:	Surrender the property.	No No
	Retain the property and redeem it.	Yes
Description of property securing debt:	Retain the property and enter into a Reaffirmation Agreement.	
	Retain the property and [explain]:	-
Creditor's	Surrender the property.	No
name:	Retain the property and redeem it.	C Yes
Description of property securing debt:	Retain the property and enter into a Reaffirmation Agreement.	
	Retain the property and [explain]:	<u>.</u>
Creditor's	Surrender the property.	No No
name:	Retain the property and redeem it.	C Yes
Description of property securing debt:	Retain the property and enter into a Reaffirmation Agreement.	
	Retain the property and [explain]:	
	·····	

Debtor 1

First Name Middle Name

Case number (If known)_

Part 2: List Your Unexpired Personal Property Leases

Last Name

For any unexpired personal property lease that you listed in Schedule G: Executory Contracts and Unexpired Leases (Official Form 106G), fill in the information below. Do not list real estate leases. Unexpired leases are leases that are still in effect; the lease period has not yet ended. You may assume an unexpired personal property lease if the trustee does not assume it. 11 U.S.C. § 365(p)(2).

essor's name:	No
Description of leased roperty:	Yes
lessor's name:	No No
Description of leased property:	Yes
Lessor's name:	No
Description of leased property:	Yes
Lessor's name:	No No
Description of leased property:	Yes
Lessor's name:	No
Description of leased property:	Yes
Lessor's name:	No
Description of leased property:	Yes
Lessor's name:	No No
Description of leased property:	C Yes

Under penalty of perjury, I declare that I have indicated my intention about any property of my estate that secures a debt and any personal property that is subject to an unexpired lease.

Signature of Debtor 1	Signature of Debtor 2	-
Date	Date	

UNITED STATES BANKRUPTCY COURT SOUTHERN DISTRICT OF WEST VIRGINIA

In re:

Case No.____

Chapter_____

Debtor(s).

VERIFICATION OF CREDITOR MATRIX

The above named debtor(s), and attorney for debtor(s) if applicable, hereby verify(ies) that the attached mailing matrix of creditors is complete, correct and consistent with the debtor(s)'s schedules to the best of my (our) knowledge.

Date:		Signature of Debtor:
Date:		Signature of Joint Debtor, if any:
Date:	·	Signature of Attorney for Debtor(s), if any:

Fill in this information to identi	fy your case:		Check one box only as directed in this form a
			Form 122A-1Supp:
Debtor 1 First Name	Middle Name	Last Name	1. There is no presumption of abuse.
Debtor 2 Spouse, if filing) First Name United States Bankruptcy Court for the	Middle Name	Last Name District of	2. The calculation to determine if a presumpti abuse applies will be made under Chapter Means Test Calculation (Official Form 122)
Case number(If known)			3. The Means Test does not apply now becau qualified military service but it could apply

Official Form 122A-1

Chapter 7 Statement of Your Current Monthly Income

12/15

Be as complete and accurate as possible. If two married people are filing together, both are equally responsible for being accurate. If more space is needed, attach a separate sheet to this form. Include the line number to which the additional information applies. On the top of any additional pages, write your name and case number (if known). If you believe that you are exempted from a presumption of abuse because you do not have primarily consumer debts or because of qualifying military service, complete and file *Statement of Exemption from Presumption of Abuse Under § 707(b)(2)* (Official Form 122A-1Supp) with this form.

Part 1: Calculate Your Current Monthly Income

1. What is your marital and filing status? Check one only.

Not married. Fill out Column A, lines 2-11.

Married and your spouse is filing with you. Fill out both Columns A and B, lines 2-11.

Married and your spouse is NOT filing with you. You and your spouse are:

Living in the same household and are not legally separated. Fill out both Columns A and B, lines 2-11.

Living separately or are legally separated. Fill out Column A, lines 2-11; do not fill out Column B. By checking this box, you declare under penalty of perjury that you and your spouse are legally separated under nonbankruptcy law that applies or that you and your spouse are living apart for reasons that do not include evading the Means Test requirements. 11 U.S.C. § 707(b)(7)(B).

Fill in the average monthly income that you received from all sources, derived during the 6 full months before you file this bankruptcy case. 11 U.S.C. § 101(10A). For example, if you are filing on September 15, the 6-month period would be March 1 through August 31. If the amount of your monthly income varied during the 6 months, add the income for all 6 months and divide the total by 6. Fill in the result. Do not include any income amount more than once. For example, if both spouses own the same rental property, put the income from that property in one column only. If you have nothing to report for any line, write \$0 in the space.

					Column A Debtor 1	Column B Debtor 2 or non-filing spouse	
2.	Your gross wages, salary, tips, bonuses, overtime, and commissions (before all payroll deductions).					\$	
3.	 Alimony and maintenance payments. Do not include payments from a spouse if Column B is filled in. 					\$	
4.	4. All amounts from any source which are regularly paid for household expenses of you or your dependents, including child support. Include regular contributions from an unmarried partner, members of your household, your dependents, parents, and roommates. Include regular contributions from a spouse only if Column B is not filled in. Do not include payments you listed on line 3.			\$	\$		
5.	Net income from operating a business, profession, or farm Gross receipts (before all deductions)	Debtor 1 \$	Debtor 2 \$				
	Ordinary and necessary operating expenses	- \$	- \$		× *		
	Net monthly income from a business, profession, or farm	\$	\$	Copy here	\$	\$	
6.	Net income from rental and other real property Gross receipts (before all deductions)	Debtor 1 \$	Debtor 2 \$				
	Ordinary and necessary operating expenses	- \$	- \$				
	Net monthly income from rental or other real property	\$	\$	Copy here	\$	\$	
7.	Interest, dividends, and royalties				\$	\$	

Debtor 1	Case number (if known)							
First Name Middle Name Last Name								
	Column A Debtor 1	Column B Debtor 2 or non-filing spouse						
8. Unemployment compensation	• • • • • • • • • • • • • • • • • • •	¢						
Do not enter the amount if you contend that the amount received was a benefit	φ	Φ						
under the Social Security Act. Instead, list it here:								
For you \$								
For your spouse\$								
 Pension or retirement income. Do not include any amount received that was a benefit under the Social Security Act. 	\$	\$						
10. Income from all other sources not listed above. Specify the source and amount. Do not include any benefits received under the Social Security Act or payments receive as a victim of a war crime, a crime against humanity, or international or domestic terrorism. If necessary, list other sources on a separate page and put the total below.	ed							
	\$	\$						
	\$	\$						
Total amounts from separate pages, if any.	+ \$	+ \$						
11. Calculate your total current monthly income. Add lines 2 through 10 for each column. Then add the total for Column A to the total for Column B.	\$	+	= Total current					
Part 2: Determine Whether the Means Test Applies to You			monthly income					
12. Calculate your current monthly income for the year. Follow these steps:								
12a. Copy your total current monthly income from line 11.	Co	py line 11 here 🗲	\$					
Multiply by 12 (the number of months in a year).			x 12					
12b. The result is your annual income for this part of the form.		12b.	\$					
13. Calculate the median family income that applies to you. Follow these steps:								
Fill in the state in which you live.								
Fill in the number of people in your household.								
Fill in the median family income for your state and size of household.		13.	\$					
To find a list of applicable median income amounts, go online using the link specified in the separate instructions for this form. This list may also be available at the bankruptcy clerk's office.								
14. How do the lines compare?								
14a: Line 12b is less than or equal to line 13. On the top of page 1, check box 1, 7 Go to Part 3.	here is no presumption	n of abuse.						
14b. Line 12b is more than line 13. On the top of page 1, check box 2, <i>The presumption of abuse is determined by Form 122A-2.</i> Go to Part 3 and fill out Form 122A–2.								
Part 3: Sign Below			·					
By signing here, I declare under penalty of perjury that the information on this	statement and in any a	attachments is true an	d correct.					
×								
	Signature of Debtor 2	-						
Signature of Debtor 1	Signature of Debtor 2							
	Date	_	~ · · ·					
MM / DD / YYYY								
If you checked line 14a, do NOT fill out or file Form 122A-2.								
If you checked line 14b, fill out Form 122A-2 and file it with this form.								

Fill in this in	formation to ident	ify your case:	
Debtor 1	-	-	
	First Name	Middle Name	Last Name
Debtor 2			
(Spouse, if filing)	First Name	Middle Name	Last Name
United States B	ankruptcy Court for the	e:	District of (State)
Case number			
(If known)			
	·		·

Official Form 122A-1Supp

Statement of Exemption from Presumption of Abuse Under § 707(b)(2) 12/15

File this supplement together with Chapter 7 Statement of Your Current Monthly Income (Official Form 122A-1), if you believe that you are exempted from a presumption of abuse. Be as complete and accurate as possible. If two married people are filing together, and any of the exclusions in this statement applies to only one of you, the other person should complete a separate Form 122A-1 if you believe that this is required by 11 U.S.C. § 707(b)(2)(C).

P	art 1:	Identify the Kind of Debts You Have					
1	Are your debts primarily consumer debts? Consumer debts are defined in 11 U.S.C. § 101(8) as "incurred by an individual primarily for a personal, family, or household purpose." Make sure that your answer is consistent with the answer you gave at line 16 of the Voluntary Petition for Individuals Filing for Bankruptcy (Official Form 101).						
	🖵 No	No. Go to Form 122A-1; on the top of page 1 of that form, check box 1, <i>There is no presumption of abuse,</i> and sign Part 3. Then submit this supplement with the signed Form 122A-1.					
	🗋 Ye	es. Go to Part 2.					
P	art 2:	Determine Whether Military Service Provisions Apply to You					
2. Are you a disabled veteran (as defined in 38 U.S.C. § 3741(1))?							
	I No. Go to line 3.						
	 Yes. Did you incur debts mostly while you were on active duty or while you were performing a homeland defense activity? 10 U.S.C. § 101(d)(1)); 32 U.S.C. § 901(1). 						
	No. Go to line 3.						
	Yes. Go to Form 122A-1; on the top of page 1 of that form, check box 1, <i>There is no presumption of abuse,</i> and sign Part 3. Then submit this supplement with the signed Form 122A-1.						
3	3. Are you or have you been a Reservist or member of the National Guard?						
	□ No. Complete Form 122A-1. Do not submit this supplement.						
	Yes. Were you called to active duty or did you perform a homeland defense activity? 10 U.S.C. § 101(d)(1); 32 U.S.C. § 901(1).						
	No. Complete Form 122A-1. Do not submit this supplement.						
		Yes. Check any one of the following categories that applies:					
		 I was called to active duty after September 11, 2001, for at least 90 days and remain on active duty. 	If you checked one of the categories to the left, go to Form 122A-1. On the top of page 1 of Form 122A-1,				
		□ I was called to active duty after September 11, 2001, for at least 90 days and was released from active duty on, which is fewer than 540 days before I file this bankruptcy case.	check box 3, <i>The Means Test does not apply now</i> , and sign Part 3. Then submit this supplement with the signed Form 122A-1. You are not required to fill out the rest of Official Form 122A-1 during the exclusion period. The				
		 I am performing a homeland defense activity for at least 90 days. I performed a homeland defense activity for at least 90 days, 	exclusion period means the time you are on active duty or are performing a homeland defense activity, and for				
-		ending on, which is fewer than 540 days before I file this bankruptcy case.	540 days afterward. 11 U.S.C. § 707(b)(2)(D)(ii). If your exclusion period ends before your case is closed, you may have to file an amended form later.				

Fill in this in	formation to identify you	r case:		Check the appropriate bo lines 40 or 42:
Debtor 1				According to the calculati
Debtor 2	First Name	Middle Name	Last Name	this Statement:
(Spouse, if filing)		Middle Name	Last Name	1. There is no presum
	Bankruptcy Court for the:		(State)	2. There is a presum
(If known)				Check if this is an a

Official Form 122A-2

Chapter 7 Means Test Calculation

12/15

To fill out this form, you will need your completed copy of Chapter 7 Statement of Your Current Monthly Income (Official Form 122A-1).

Be as complete and accurate as possible. If two married people are filing together, both are equally responsible for being accurate. If more space is needed, attach a separate sheet to this form. Include the line number to which the additional information applies. On the top of any additional pages, write your name and case number (if known).

Р	art 1:	Determine Your Adjusted Income			
1.	Сору ус	ur total current monthly income	Copy line 11 from Offici	ial Form 122A-1 here ➔	\$
2	Did you	fill out Column B in Part 1 of Form 122A–1?			
Ĩ	•	Fill in \$0 for the total on line 3.			
		Is your spouse filing with you?			
		No. Go to line 3.			
		Yes. Fill in \$0 for the total on line 3.			
3.	househ On line regularly	your current monthly income by subtracting any part of your signal expenses of you or your dependents. Follow these steps: 11, Column B of Form 122A–1, was any amount of the income you your dependents? Fill in 0 for the household expenses of you or your dependents?			
				1.1	
	L Yes.	Fill in the information below:			
	F	tate each purpose for which the income was used or example, the income is used to pay your spouse's tax debt or to support eople other than you or your dependents	Fill in the amount you are subtracting from your spouse's income		
			\$		
			\$		
		· · · · · · · · · · · · · · · · · · ·	+ \$		
	Т	otal	\$	Copy total here	
	1.1				
4	Adjust	your current monthly income. Subtract the total on line 3 from line	e 1.		\$

Debtor 1

First Name Middle Name

Last Name

Case number (if known)

Pa	art 2:	Calculate Your Deductions from Your In	come
i.	answer	rnal Revenue Service (IRS) issues National and the questions in lines 6-15. To find the IRS stand n. This information may also be available at the	Local Standards for certain expense amounts. Use these amounts to dards, go online using the link specified in the separate instructions for bankruptcy clerk's office.
ł	actual ex	he expense amounts set out in lines 6-15 regardles openses if they are higher than the standards. Do no ot deduct any operating expenses that you subtrac	s of your actual expense. In later parts of the form, you will use some of your ot deduct any amounts that you subtracted from your spouse's income in line 3 ted from income in lines 5 and 6 of Form 122A–1.
	If your ex	penses differ from month to month, enter the average	age expense.
	Wheneve	er this part of the form refers to you, it means both	you and your spouse if Column B of Form 122A-1 is filled in.
	5. The	number of people used in determining your de	ductions from income
	Fill plus	in the number of people who could be claimed as e s the number of any additional dependents whom yo number of people in your household.	xemptions on your federal income tax return,
	Nation	al Standards You must use the IRS National	Standards to answer the questions in lines 6-7.
	6. Foo in th	od, clothing, and other items: Using the number of the number of the number of the dollar amount for food, clothing, and other items.	of people you entered in line 5 and the IRS National Standards, fill
	fill i und actu	n the dollar amount for out-of-pocket health care. T	ther of people you entered in line 5 and the IRS National Standards, he number of people is split into two categories—people who are der people have a higher IRS allowance for health care costs. If your may deduct the additional amount on line 22.
	,7a.	Out-of-pocket health care allowance per person	\$
	7b.	Number of people who are under 65	, X
	7c.	Subtotal. Multiply line 7a by line 7b.	\$ Copy here ➔ \$
	P	eople who are 65 years of age or older	
	7d.	Out-of-pocket health care allowance per person	\$
	7e.	Number of people who are 65 or older	X
	7f.	Subtotal. Multiply line 7d by line 7e.	\$ Copy here → + \$
	7g.	Total. Add lines 7c and 7f	\$ Copy total here ➔

D	e	b	to	r	1

First Name

Last Name

Case number (if known)

ocal Standards	You must use the	IRS Local Standards to	o answer the questions i	in lines 8-15.			
	ion from the IRS, the es into two parts:	e U.S. Trustee Program	m has divided the IRS	Local Standa	ard for housing	for	
		d operating expenses	s				
	lities – Mortgage or						
		se the U.S. Trustee P					
) find the chart, go his chart may also	online using the link be available at the ba	specified in the separat ankruptcy clerk's office.	te instructions for this fo	om.			
Housing and ut dollar amount lis	ilities – Insurance an ted for your county fo	nd operating expense r insurance and operati	s: Using the number of ing expenses.	people you e	ntered in line 5,	fill in the \$_	
Housing and ut	ilities – Mortgage or	rent expenses:					
9a. Using the nu	mber of people you e	ntered in line 5, fill in th	ne dollar amount listed		\$		
	ity for mongage of re-						
9b. Total averag	e monthly payment for	or all mortgages and ot	her debts secured by yo	our home.			
contractually	the total average mor due to each secured Then divide by 60.	nthly payment, add all a creditor in the 60 mont	amounts that are ths after you file for				
Name of the	creditor		Average monthly payment				
			\$	_			
			\$	-			
			+ s				
			r	-			
	Total aver	age monthly payment	\$	Copy here➔	-\$	Repeat this amount on line 33a.	
	ge or rent expense.					Сору	
Subtract lin	e 9b (<i>total average m</i> se): If this amount is le	onthly payment) from li ess than \$0, enter \$0	ine 9a (<i>mortgage or</i>		\$	here	
. If you claim tha the calculation	t the U.S. Trustee P of your monthly exp	rogram's division of t penses, fill in any add	he IRS Local Standard itional amount you cla	t for housing aim.	is incorrect an	id affects \$_	
Explain							
why:							
	54.						
Local transport	ation expenses: Che	eck the number of vehic	cles for which you claim	an ownership	or operating ex	pense.	
0. Go to line	e 14.						
	e 12.						
1. Go to line							
-	Go to line 12.						
—	Go to line 12.		·				
2 or more.	on expense: Usina ti	he IRS Local Standards	s and the number of veh your Census region or	nicles for which metropolitans	h you claim the statistical area		
2 or more.	on expense: Usina ti	ne IRS Local Standards	s and the number of veh your Census region or	nicles for whic metropolitan s	h you claim the statistical area.	\$_	

Chapter 7 Means Test Calculation

page 3

Vehicle ownership or lease expense: Using the IRS Local Standards, calculate the net ownership or lease expense for each vehicle below. You may not claim the expense if you do not make any loan or lease payments on the vehicle. In addition, you may not claim the expense for more than two vehicles. Vehicle 1 Describe Vehicle 1: 13a. Ownership or leasing costs using IRS Local Standard. \$	
for each vehicle below. You may not claim the expense if you do not make any loan or lease payments on the vehicle. In addition, you may not claim the expense for more than two vehicles. Vehicle 1 Describe Vehicle 1:	
13a. Ownership or leasing costs using IRS Local Standard	
13a. Ownership or leasing costs using IRS Local Standard	
13a. Ownership or leasing costs using IRS Local Standard.	
13b. Average monthly payment for all debts secured by Vehicle 1. Do not include costs for leased vehicles.	
To calculate the average monthly payment here and on line 13e, add all amounts that are contractually due to each secured creditor in the 60 months after you filed for bankruptcy. Then divide by 60.	
Name of each creditor for Vehicle 1 Average monthly payment	
Ψ	
+ s	
Repeat this	9
Total average monthly payment \$ amount on	
line 33b.	
Copy net	
13c. Net Vehicle 1 ownership or lease expense Vehicle 1 Subtract line 13b from line 13a. If this amount is less than \$0, enter \$0. \$	
Subtract line 13b from line 13a, if this amount is less than 40, enter 40.	\$
Vehicle 2 Describe Vehicle 2:	
13d. Ownership or leasing costs using IRS Local Standard	
13e. Average monthly payment for all debts secured by Vehicle 2.	
Do not include costs for leased vehicles.	
Name of each creditor for Vehicle 2 Average monthly	
Name of each creditor for Vehicle 2 Average monthly payment	
\$\$	
\$\$	
\$\$ + \$	
\$\$	
S Copy S Copy S Repeat thi amount or	
<pre>\$\$</pre> Total average monthly payment \$ Copy here→ -\$ Repeat this amount or line 33c.	
Total average monthly payment \$ here - \$ amount or line 33c.	
Total average monthly payment \$ \$ amount or line 33c. 13f. Net Vehicle 2 ownership or lease expense Copy net Vehicle 2	
Total average monthly payment \$here → -\$ amount or line 33c. 13f. Net Vehicle 2 ownership or lease expense Copy net Vehicle 2 Subtract line 13e from 13d. If this amount is less than \$0, enter \$0, \$	1
Total average monthly payment \$	1
Total average monthly payment \$	1
Total average monthly payment \$	1
Total average monthly payment \$here - \$ amount on line 33c. 13f. Net Vehicle 2 ownership or lease expense Subtract line 13e from 13d. If this amount is less than \$0, enter \$0 \$ \$ Copy net Vehicle 2 expense here > Public transportation expense: If you claimed 0 vehicles in line 11, using the IRS Local Standards, fill in the	1

Chapter 7 Means Test Calculation

r 1		Case number (if known)	
First Name Middle N	ame Last Name		
ther Necessary Expenses		tions listed above, you are allowed your monthly expenses for	
	the following IRS categories.		alientaiten.
Taxaa, The total monthly	mount that you will actually owe for	federal, state and local taxes, such as income taxes, self-	
employment taxes social	security taxes, and Medicare taxes.	You may include the monthly amount withheld from your	\$
pay for these taxes. Howe	ver, if you expect to receive a tax refu	und, you must divide the expected refund by 12 and	· · ·
subtract that number from	the total monthly amount that is with	held to pay for taxes.	
Do not include real estate,	sales, or use taxes.		
	The total monthly neural deductions	that your job requires, such as retirement contributions	
union dues, and uniform c		that your job requires, such as retirement contributions,	
		as voluntary 401(k) contributions or payroll savings.	\$
	·		
Life insurance: The total	monthly premiums that you pay for y	our own term life insurance. If two married people are filing	
together, include payment	s that you make for your spouse's ter	rm life insurance. Do not include premiums for life	
insurance on your depend	ents, for a non-filing spouse's life ins	urance, or for any form of life insurance other than term.	\$
Court-ordered payments	: The total monthly amount that you	pay as required by the order of a court or administrative	
agency, such as spousal of	or child support payments.		\$
Do not include payments of	on past due obligations for spousal of	r child support. You will list these obligations in line 35.	♥
Education. The total	thly amount that you pay for education	on that is either required.	
		on man's enner required.	
as a condition for your j		no public advection is available for similar convince	\$
for your physically or me	entally challenged dependent child if	no public education is available for similar services.	Ť
Childeare: The total mont	by amount that you pay for childcare	e, such as babysitting, daycare, nursery, and preschool.	
	or any elementary or secondary sch		\$
Do not include payments	of any elementary of secondary some	oo oddodaon.	
Additional health care ex	penses, excluding insurance cost	ts: The monthly amount that you pay for health care that	
is required for the health a	nd welfare of you or your dependent	s and that is not reimbursed by insurance or paid by a	
	clude only the amount that is more the ance or health savings accounts sho		\$
Fayments for health insur	ance of nearth savings accounts show		-
O-flowed to low how on a	telephone convision: The total man	thly amount that you pay for telecommunication services for	
vou and your dependents.	such as pagers, call waiting, caller in	dentification, special long distance, or business cell phone	
service, to the extent nece	ssary for your health and welfare or	that of your dependents or for the production of income, if it	+ \$
is not reimbursed by your			
Do not include payments	or basic home telephone, internet ar	nd cell phone service. Do not include self-employment 2A-1, or any amount you previously deducted.	
expenses, such as mose			P
Add all of the expenses	allowed under the IRS expense all	owances.	e
Add lines 6 through 23.			°
	•		L

	or 1 Case number (if known)	
Ado	Additional Expense Deductions These are additional deductions allowed by the Means Test. Note: Do not include any expense allowances listed in lines 6-24	.	
	 Health insurance, disability insurance, and health savings account expenses. The monthly existence, disability insurance, and health savings accounts that are reasonably necessary for you dependents. 	penses for health rself, your spouse, or your	
	Health insurance \$		
	Disability insurance \$		
	Health savings account + \$		
	Total \$ Copy total	here 🗲 S	j
	Do you actually spend this total amount?		
	No. How much do you actually spend?		
	Yes		
6.	6. Continuing contributions to the care of household or family members. The actual monthly excontinue to pay for the reasonable and necessary care and support of an elderly, chronically ill, or your household or member of your immediate family who is unable to pay for such expenses. The include contributions to an account of a qualified ABLE program. 26 U.S.C. § 529A(b).	disabled member of	\$
7.	7. Protection against family violence. The reasonably necessary monthly expenses that you incur of you and your family under the Family Violence Prevention and Services Act or other federal law By law, the court must keep the nature of these expenses confidential.	to maintain the safety s that apply.	§
		enting expenses on line 0	
8.	8. Additional home energy costs. Your home energy costs are included in your insurance and oper		
	If you believe that you have home energy costs that are more than the home energy costs include 8, then fill in the excess amount of home energy costs.		\$
	You must give your case trustee documentation of your actual expenses, and you must show that claimed is reasonable and necessary.	the additional amount	
9.	9. Education expenses for dependent children who are younger than 18. The monthly expenses per child) that you pay for your dependent children who are younger than 18 years old to attend a elementary or secondary school. You must give your case trustee documentation of your actual expenses, and you must explain who are younger than 18. The monthly expenses	private or public	\$
	reasonable and necessary and not already accounted for in lines 6-23.	ato of adjustment	
	* Subject to adjustment on 4/01/16, and every 3 years after that for cases begun on or after the d	ate of adjustment.	
0.	 Additional food and clothing expense. The monthly amount by which your actual food and cloth higher than the combined food and clothing allowances in the IRS National Standards. That amou 5% of the food and clothing allowances in the IRS National Standards. 	nt cannot be more than	\$
	To find a chart showing the maximum additional allowance, go online using the link specified in the this form. This chart may also be available at the bankruptcy clerk's office.	e separate instructions for	
	You must show that the additional amount claimed is reasonable and necessary.		
	1. Continuing charitable contributions. The amount that you will continue to contribute in the form	of cash or financial +	\$
1.	instruments to a religious or charitable organization. 26 U.S.C. § 170(c)(1)-(2).		
1.	instruments to a religious or chantable organization. 20 0.3.0. § 170(0)(1)-(2).		-
	2. Add all of the additional expense deductions.		\$

Г

	t Name Middle Name Las	st Name	Case ha	Imber (# known)	-	
ductions	for Debt Payment					
For debt	s that are secured by an interes	st in property that you own	, including home mo	rtgages, vehicle		
To calcula	ad other secured debt, fill in lin ate the total average monthly pay	vment, add all amounts that a	are contractually due to	each secured		
creditor in	n the 60 months after you file for l	bankruptcy. Then divide by 6	0.			
M	ortgages on your home:			Average monthly payment		
	py line 9b here			\$		
La	ans on your first two vehicles:					
	py line 13b here.			\$		
33c. Co	py line 13e here.		······ >	\$		
	t other secured debts:					
10	Name of each creditor for other	Identify property that	Does payment			
	secured debt	secures the debt	include taxes or insurance?			
	(CARAC	18500V (1.000007022), "96780801(1) - 546667646671	No No	\$		
		-	Yes	T		
			No Ves	\$		
		<u> </u>	Yes	+ \$		
33e. Total	average monthly payment. Add I	lines 33a through 33d.		\$	Copy total here	\$
Are	debts that you listed in line 33 property necessary for your s	secured by your primary re upport or the support of yo	sidence, a vehicle, our dependents?			
or other	Go to line 35. State any amount that you must listed in line 33, to keep possess Next divide by 60 and fill in the i	sion of your property (called th	to the payments			
or other	State any amount that you must listed in line 33, to keep possess Next, divide by 60 and fill in the in Name of the creditor	sion of your property (called th information below.	to the payments he <i>cure amount</i>).	Monthly cure amount		
or other	State any amount that you must listed in line 33, to keep possess Next, divide by 60 and fill in the in Name of the creditor	sion of your property (called the information below.	to the payments he <i>cure amount</i>).			
or other	State any amount that you must listed in line 33, to keep possess Next, divide by 60 and fill in the in Name of the creditor	sion of your property (called the information below.	to the payments he <i>cure amount</i>). al cure punt			
or other	State any amount that you must listed in line 33, to keep possess Next, divide by 60 and fill in the in Name of the creditor	sion of your property (called the information below.	to the payments the cure amount). al cure bunt + 60 =			
or other	State any amount that you must listed in line 33, to keep possess Next, divide by 60 and fill in the in Name of the creditor	sion of your property (called the information below.	to the payments the <i>cure amount</i>). al cure bunt + 60 = + 60 = + 60 =		Copy total	\$
or other	State any amount that you must listed in line 33, to keep possess Next, divide by 60 and fill in the in Name of the creditor	sion of your property (called the information below.	to the payments he <i>cure amount</i>). al cure bunt + 60 = + 60 =		Copy total here→	\$
Or other	State any amount that you must listed in line 33, to keep possess Next, divide by 60 and fill in the in Name of the creditor	sion of your property (called the information below. entify property that cures the debt Support the debt Support the debt Support the support	to the payments he cure amount). al cure bunt $\div 60 =$ $\div 60 =$ $\div 60 =$ Total			\$
Do you that are No.	State any amount that you must listed in line 33, to keep possess Next, divide by 60 and fill in the in Name of the creditor lide set 	sion of your property (called the information below. entify property that cures the debt amo\$\$\$\$	to the payments he <i>cure amount</i>). al cure bunt 			\$
Do you that are No.	State any amount that you must listed in line 33, to keep possess Next, divide by 60 and fill in the ii Name of the creditor lde sec 	sion of your property (called the information below. entify property that cures the debt amo s	to the payments he <i>cure amount</i>). al cure bunt 			\$

First Name Middle Name Last Name		Case number (# kn	own)		
6. Are you eligible to file a case under Chapter 13? 1 For more information, go online using the link for Bank	1 U.S.C. § 109(e).	e separate			
instructions for this form. Bankruptcy Basics may also	be available at the bankrupto	y clerk's office.			
No. Go to line 37.					
Yes. Fill in the following information.					
Projected monthly plan payment if you were fi	iling under Chapter 13	¢			
		Ψ			
Current multiplier for your district as stated on Administrative Office of the United States Cou North Carolina) or by the Executive Office for other districts).	urts (for districts in Alabama a	nd II X	· .		
To find a list of district multipliers that includes link specified in the separate instructions for the available at the bankruptcy clerk's office.	s your district, go online using his form. This list may also be	the		· .	
Average monthly administrative expense if yo	were filing under Chapter 1	3 \$		Copy total here	\$
. Add all of the deductions for debt payment. Add lines 33e through 36					\$
otal Deductions from Income					
Add all of the allowed deductions.					
Copy line 24, All of the expenses allowed under IRS expense allowances	\$				
Copy line 32, All of the additional expense deductions	······ 3				
Copy line 37, All of the deductions for debt payment	+\$				
	·				
Total deduction	ns \$	Copy total h	ere		\$
art 3: Determine Whether There Is a Presum	nption of Abuse				
. Calculate monthly disposable income for 60 months	s				
39a. Copy line 4, adjusted current monthly income	. \$				
39b. Copy line 38, Total deductions	- \$				
	· · · · · · · · · · · · · · · · · · ·				
39c. Monthly disposable income. 11 U.S.C. § 707(b)(Subtract line 39b from line 39a.	2)\$	Copy here	\$		
					·
For the next 60 months (5 years)			x 60		
				Сору	
39d. Total. Multiply line 39c by 60.	· · · ·		\$	here 🗲	\$
			ξ		
). Find out whether there is a presumption of abuse. (Check the box that applies:				
The line 39d is less than \$7,475*. On the top of part to Part 5.		1, There is no pre	esumption of ab	use. Go	
The line 39d is more than \$12,475*. On the top of may fill out Part 4 if you claim special circumstance	f page 1 of this form, check bo s. Then go to Part 5.	ox 2, There is a p	resumption of a	<i>buse</i> . You	
□ The line 39d is at least \$7,475*, but not more that	an \$12,475*. Go to line 41.				
* Subject to adjustment on 4/01/16, and every 3 ye	ears after that for cases filed o	on or after the dat	te of adjustmen		

or 1	First Name Last Name .			
	Fill in the amount of your total nonpriority unsecured debt. If you filled out A Summary of Your Assets and Liabilities and Certain Statistical Information Schedules (Official Form 106Sum), you may refer to line 3b on that form.	c		-
		» x .25		
41b.	25% of your total nonpriority unsecured debt. 11 U.S.C. § 707(b)(2)(A)(i)(I). Multiply line 41a by 0.25.	\$	Copy here →	\$
is end	mine whether the income you have left over after subtracting all allowed deductions ough to pay 25% of your unsecured, nonpriority debt. k the box that applies:			
	ine 39d is less than line 41b. On the top of page 1 of this form, check box 1, There is no presu	umption of abuse.		
	o to Part 5.			
	ine 39d is equal to or more than line 41b. On the top of page 1 of this form, check box 2, <i>The f abuse.</i> You may fill out Part 4 if you claim special circumstances. Then go to Part 5.	ere is a presumption	7	
t 4:	Give Details About Special Circumstances			
No.	have any special circumstances that justify additional expenses or adjustments of current ble alternative? 11 U.S.C. § 707(b)(2)(B). Go to Part 5. Fill in the following information. All figures should reflect your average monthly expense or incor for each item. You may include expenses you listed in line 25.	me adjustment	for which	there is no
■ No. ■ Yes.	ble alternative? 11 U.S.C. § 707(b)(2)(B). Go to Part 5. Fill in the following information. All figures should reflect your average monthly expense or inco	me adjustment	for which	there is no
■ No. ■ Yes.	ble alternative? 11 U.S.C. § 707(b)(2)(B). Go to Part 5. Fill in the following information. All figures should reflect your average monthly expense or inco for each item. You may include expenses you listed in line 25. You must give a detailed explanation of the special circumstances that make the expenses or in adjustments necessary and reasonable. You must also give your case trustee documentation o	me adjustment	y expense	there is no
No.	ble alternative? 11 U.S.C. § 707(b)(2)(B). Go to Part 5. Fill in the following information. All figures should reflect your average monthly expense or incor for each item. You may include expenses you listed in line 25. You must give a detailed explanation of the special circumstances that make the expenses or in adjustments necessary and reasonable. You must also give your case trustee documentation o expenses or income adjustments.	me adjustment ncome of your actual Average monthi	y expense	there is no
No.	ble alternative? 11 U.S.C. § 707(b)(2)(B). Go to Part 5. Fill in the following information. All figures should reflect your average monthly expense or incor for each item. You may include expenses you listed in line 25. You must give a detailed explanation of the special circumstances that make the expenses or in adjustments necessary and reasonable. You must also give your case trustee documentation o expenses or income adjustments.	me adjustment ncome of your actual Average monthi	y expense	there is no
No.	ble alternative? 11 U.S.C. § 707(b)(2)(B). Go to Part 5. Fill in the following information. All figures should reflect your average monthly expense or incor for each item. You may include expenses you listed in line 25. You must give a detailed explanation of the special circumstances that make the expenses or in adjustments necessary and reasonable. You must also give your case trustee documentation o expenses or income adjustments.	me adjustment ncome of your actual Average monthi	y expense	there is no
No.	ble alternative? 11 U.S.C. § 707(b)(2)(B). Go to Part 5. Fill in the following information. All figures should reflect your average monthly expense or incor for each item. You may include expenses you listed in line 25. You must give a detailed explanation of the special circumstances that make the expenses or in adjustments necessary and reasonable. You must also give your case trustee documentation o expenses or income adjustments.	me adjustment ncome of your actual Average monthi	y expense	there is no
■ No. ■ Yes.	ble alternative? 11 U.S.C. § 707(b)(2)(B). Go to Part 5. Fill in the following information. All figures should reflect your average monthly expense or incor for each item. You may include expenses you listed in line 25. You must give a detailed explanation of the special circumstances that make the expenses or in adjustments necessary and reasonable. You must also give your case trustee documentation o expenses or income adjustments.	me adjustment ncome of your actual Average monthi	y expense	there is no
No.	ble alternative? 11 U.S.C. § 707(b)(2)(B). Go to Part 5. Fill in the following information. All figures should reflect your average monthly expense or incor for each item. You may include expenses you listed in line 25. You must give a detailed explanation of the special circumstances that make the expenses or in adjustments necessary and reasonable. You must also give your case trustee documentation o expenses or income adjustments.	me adjustment ncome of your actual Average monthi	y expense	there is no
easonal	ble alternative? 11 U.S.C. § 707(b)(2)(B). Go to Part 5. Fill in the following information. All figures should reflect your average monthly expense or incor for each item. You may include expenses you listed in line 25. You must give a detailed explanation of the special circumstances that make the expenses or in adjustments necessary and reasonable. You must also give your case trustee documentation o expenses or income adjustments.	me adjustment ncome of your actual Average monthi	y expense	there is no
reasonal	ble alternative? 11 U.S.C. § 707(b)(2)(B). Go to Part 5. Fill in the following information. All figures should reflect your average monthly expense or inco for each item. You may include expenses you listed in line 25. You must give a detailed explanation of the special circumstances that make the expenses or in adjustments necessary and reasonable. You must also give your case trustee documentation of expenses or income adjustments. Give a detailed explanation of the special circumstances 	me adjustment ncome of your actual Average monthil or income adjus \$ \$ \$ \$ \$	y expense stment	
reasonal	ble alternative? 11 U.S.C. § 707(b)(2)(B). Go to Part 5. Fill in the following information. All figures should reflect your average monthly expense or inco for each item. You may include expenses you listed in line 25. You must give a detailed explanation of the special circumstances that make the expenses or in adjustments necessary and reasonable. You must also give your case trustee documentation o expenses or income adjustments. Give a detailed explanation of the special circumstances 	me adjustment ncome of your actual Average monthil or income adjus \$ \$ \$ \$ \$	y expense stment	
t 5:	ble alternative? 11 U.S.C. § 707(b)(2)(B). Go to Part 5. Fill in the following information. All figures should reflect your average monthly expense or inco for each item. You may include expenses you listed in line 25. You must give a detailed explanation of the special circumstances that make the expenses or in adjustments necessary and reasonable. You must also give your case trustee documentation of expenses or income adjustments. Give a detailed explanation of the special circumstances 	me adjustment ncome of your actual Average monthil or income adjus \$ \$ \$ \$ \$	y expense stment	
reasonal	ble alternative? 11 U.S.C. § 707(b)(2)(B). Go to Part 5. Fill in the following information. All figures should reflect your average monthly expense or inco for each item. You may include expenses you listed in line 25. You must give a detailed explanation of the special circumstances that make the expenses or in adjustments necessary and reasonable. You must also give your case trustee documentation o expenses or income adjustments. Give a detailed explanation of the special circumstances Give a detail	me adjustment	y expense stment	

COMMITTEE NOTE

Official Form 101A, Initial Statement About an Eviction Judgment Against You, and Official Form 101B, Statement About Payment of an Eviction Judgment Against You, are new forms promulgated as part of the Forms Modernization Project. They replace the "Certification by a Debtor Who Resides as a Tenant of Residential Property" section on Official Form 1, Voluntary Petition. The forms apply only in cases of individual debtors.

Official Form 101A explains that debtors need to complete and file the form only if their landlord has a judgment for possession or an eviction judgment against them and they rent their residence. The form further explains that if the debtor wishes to stay in their residence for 30 days after filing their bankruptcy petition, the certification must be completed. The form adds references to the provisions in the Bankruptcy Code that specify when debtor-tenants subject to eviction may remain in their residence after filing for bankruptcy.

The form eliminates the checkboxes that the debtor has served the landlord with the certification and paid the court the rent that would be due during the 30 days after the filing of the bankruptcy petition. Instead, debtors are required to certify under penalty of perjury that the rent has been paid to the court, and the instructions direct debtors to serve a copy of the statement on the landlord.

The form eliminates the checkbox that the debtor claims there are circumstances under applicable nonbankruptcy law under which the debtor would be permitted to cure the monetary default that gave rise to the judgment for possession (or eviction judgment) and remain in residence. Instead, debtors are required to certify under penalty of perjury that they have the right to stay in their residence under state law or other nonbankruptcy law by paying their landlord the entire delinquent amount. **Official Form 101B** is new. If debtors wish to stay in their residence for more than 30 days after filing the petition, they must complete, file, and serve the form within 30 days after the petition is filed. Under Official Form 101B, debtors certify under penalty of perjury that they have the right to stay in their residence under state law or other nonbankruptcy law by paying their landlord the entire delinquent amount and that they have paid their landlord the entire amount owed as stated in the judgment for possession or in the eviction judgment.

	Fill in this in	formation to identify	your case:		
	Debtor 1				
		First Name	Middle Name	Last Name	
	Debtor 2		Middle Name	Last Name	
	(Spouse, if filing)	First Name	Middle Name	Last Name	
•	United States E	ankruptcy Court for the:		District of	
ļ					(State)
	Case number (If known)				
Ļ					

Official Form 101A

Initial Statement About an Eviction Judgment Against You

12/15

File this form with the court and serve a copy on your landlord when you first file bankruptcy only if:

you rent your residence; and

your landlord has obtained a judgment for possession in an eviction, unlawful detainer action, or

similar proceeding (called eviction judgment) against you to possess your residence.

Landlord's name				
Landlord's address	Number Street			
	City	State	ZIP Code	

If you want to stay in your rented residence after you file your case for bankruptcy, also complete the certification below.

l c	ertify under penalty of	perjury that:
		ther nonbankruptcy law that applies to the judgment for possession (<i>eviction judgment</i>), ay in my residence by paying my landlord the entire delinquent amount.
		nkruptcy court clerk a deposit for the rent that would be due during the 30 days after I file on for Individuals Filing for Bankruptcy (Official Form 101).
	×	×
	Signature of Debte	tor 1 Signature of Debtor 2
	Date MM / DD	/ YYYY Date
ŝ	tay of Eviction: (a)	First 30 days after bankruptcy. If you checked both boxes above, signed the form to certify that both apply, and served your landlord with a copy of this statement, the automatic stay under 11 U.S.C. § 362(a)(3) will apply to the continuation of the eviction against you for 30 days after you file your <i>Voluntary Petition for Individuals Filing for Bankruptcy</i> (Official Form 101).
	(b)	Stay after the initial 30 days. If you wish to stay in your residence after that 30-day period and continue to receive the protection of the automatic stay under 11 U.S.C. § 362(a)(3), you must pay the entire delinquent amount to your landlord as stated in the eviction judgment before the 30-day period ends. You must also fill out <i>Statement About Payment of an Eviction Judgment Against You</i> (Official Form 101B), file it with the bankruptcy court, and serve your landlord a copy of it before the 30-day period ends.

11 U.S.C. §§ 362(b)(22) and 362(l)

Official Form 101A

Initial Statement About an Eviction Judgment Against You

Fill in this information to identify your case:							
Debtor 1							
	First Name		Middle Name	Last Name			
Debtor 2							
(Spouse, if filing) First Name		Middle Name	Last Name			
United States	Bankruptcy Cou	urt for the:		District of			
				(State)			
Case number							
(If known)					_		

Official Form 101B

Statement About Payment of an Eviction Judgment Against You 12/15

Fill out this form only if:

- you filed Initial Statement About an Eviction Judgment Against You (Official Form 101A); and
- you served a copy of Form 101A on your landlord; and
- you want to stay in your rented residence for more than 30 days after you file your Voluntary Petition for Individuals Filing for Bankruptcy (Official Form 101).

File this form within 30 days after you file your Voluntary Petition for Individuals Filing for Bankruptcy (Official Form 101). Also serve a copy on your landlord within that same time period.

Certification About Applicable Law and Payment of Eviction Judgment

I certify under penalty of perjury that (Check all that apply):

- Under the state or other nonbankruptcy law that applies to the judgment for possession (*eviction judgment*), I have the right to stay in my residence by paying my landlord the entire delinquent amount.
- Within 30 days after I filed my Voluntary Petition for Individuals Filing for Bankruptcy (Official Form 101), I have paid my landlord the entire amount I owe as stated in the judgment for possession (eviction judgment).

Signature of Debtor 1	Signature of Debtor 2
Date MM / _DD _ / YYYY	Date MM / DD / YYYY

You must serve your landlord with a copy of this form.

Check the Bankruptcy Rules (<u>www.uscourts.gov/rulesandpolicies/rules.aspx</u>) and the court's local website (go to <u>http://www.uscourts.gov/Court_Locator.aspx</u> to find your court's website) for any specific requirements that you might have to meet to serve this statement.

Official Form 101B

Statement About Payment of an Eviction Judgment Against You

					1.11
	Fill in this information to ic	tentify y	our case:		
-	United States Bankruptcy Co	ourt for t	he:		
		strict of	State	-	1 A.
	Case number (If known):	-			

Official Form 121

Statement About Your Social Security Numbers

12/15

Use this form to tell the court about any Social Security or federal Individual Taxpayer Identification numbers you have used. Do not file this form as part of the public case file. This form must be submitted separately and must not be included in the court's public electronic records. Please consult local court procedures for submission requirements.

To protect your privacy, the court will not make this form available to the public. You should not include a full Social Security Number or Individual Taxpayer Number on any other document filed with the court. The court will make only the last four digits of your numbers known to the public. However, the full numbers will be available to your creditors, the U.S. Trustee or bankruptcy administrator, and the trustee assigned to your case.

Making a false statement, concealing property, or obtaining money or property by fraud in connection with a bankruptcy case can result in fines up to \$250,000, or imprisonment for up to 20 years, or both. 18 U.S.C. §§ 152, 1341, 1519, and 3571.

Part 1: Tell the Court	About Yourself and Your spouse if Your Spouse is	Filing With You
	For Debtor 1:	For Debtor 2 (Only If Spouse Is Filing):
1. Your name		
	First name	First name
	Middle name	Middle name
	Last name	Last name
Part 2: Tell the Court	About all of Your Social Security or Federal Indivi	idual Taxpayer Identification Numbers
2. All Social Security		
Numbers you have used		<u> </u>
4504		
	You do not have a Social Security number.	You do not have a Social Security number.
3. All federal Individual Taxpayer	9	9
Identification Numbers (ITIN) you have used	9	9
	You do not have an ITIN.	You do not have an ITIN.
Part 3: Sign Below		· · · · · · · · · · · · · · · · · · ·
-	Under penalty of perjury, I declare that the information I have provided in this form is true and correct.	Under penalty of perjury, I declare that the information I have provided in this form is true and correct.
	*	*
- 1	Signature of Debtor 1	Signature of Debtor 2
	Date	Date
	MM / DD / YYYY	MM / DD / YYYY

Statement About Your Social Security Numbers

Form payadvic		
		NKRUPTCY COURT t of West Virginia
In Re: Debtor(;	s)	Case No.: Chapter: Judge: B. McKay Mignault
		Y OF PERJURY CONCERNING TO 11 U.S.C. SECTION 521(a)(1)(B)(iv)
I*, evidence of pay because:	(Debtor's name), state ment received within 60 days before the days	e that I did not file copies of all payment advices or other the of the filing of the petition, by me from any employer
(1)	I was not employed during the period im case(mediately preceding the filing of the above-referenced state the dates that you were not employed);
(2)	I was employed during the period immed did not receive any payment advices or o days before the filing of the petition;	liately preceding the filing of the above referenced case but ther evidence of payment from my employer within 60
(3)	I am self employed and do not receive ar	y evidence of payment;
(4)	Other (please explain)	
	under penalty of perjury that I have read the moviedge, information and belief.	foregoing statements and that they are true and accurate to
Dated this	day of	, 20
		(Signature of Debtor)
* A separate for	rm must be filed for each Debtor.	

Fill in this in	formation to identify	your case:			
Debtor 1					
Deblor	First Name	Middle Name	Last Name		_
Debtor 2				-	
(Spouse, if filing)	First Name	Middle Name	Last Name		
United States E	Bankruptcy Court for the:		District of	(State)	-
Case number				(State)	
(II KHOWII)					

Check if this is an amended filing

Official Form 103A

Part 1:

Application for Individuals to Pay the Filing Fee in Installments

Be as complete and accurate as possible. If two married people are filing together, both are equally responsible for supplying correct information.

Pa	rt 1: Specify Your Proposed Payment	Timetable		
1.	Which chapter of the Bankruptcy Code are you choosing to file under?	 Chapter 7 Chapter 11 Chapter 12 Chapter 13 		
2.	You may apply to pay the filing fee in up to four installments. Fill in the amounts you propose to pay and the dates you plan to pay them. Be sure all dates are business days. Then add the payments you propose to pay. You must propose to pay the entire fee no later than 120 days after you file this bankruptcy case. If the court approves your application, the court will set your final payment timetable. Total	You propose to pay \$	 With the filing of the petition On or before this date 	MM / DD / YYYY Mtire fee for the chapter you checked in line 1. 1.
Pa	rt 2: Sign Below	· · · · · · · · · · · · · · · · · · ·	· · · · · · · · · · · · · · · · · · ·	
u	y signing here, you state that you are unable inderstand that:			
	You must pay your entire filing fee before you preparer, or anyone else for services in conne			n attorney, bankruptcy petition
	You must pay the entire fee no later than 120 debts will not be discharged until your entire for		nkruptcy, unless the court later e	extends your deadline. Your
	If you do not make any payment when it is du may be affected.	e, your bankruptcy case may	be dismissed, and your rights in	other bankruptcy proceedings
x	×		×	
	Signature of Debtor 1	ignature of Debtor 2	Your attorn	ey's name and signature, if you used one
	Date DD / YYYY	MM / DD / YYYY	Date MM	/ DD / YYYY
Offic	tial Form 103A	Application for Individuals t	o Pay the Filing Fee in Installn	nents

12/15

Fill in this in	formation to identify	your case:			
Debtor 1	First Name	Middle Name		Last Name	
Debtor 2 (Spouse, if filing)	First Name	Middle Name		Last Name	
United States E	Bankruptcy Court for the:		District of		
Case number (If known)	······································				

Check if this is an amended filing

Official Form 103B Application to Have the Chapter 7 Filing Fee Waived

12/15

Be as complete and accurate as possible. If two married people are filing together, both are equally responsible for supplying correct information. If more space is needed, attach a separate sheet to this form. On the top of any additional pages, write your name and case number (if known).

Ρ	art 1: Tell the Court About Yo	our Family and Your F	amily's Income		
1.	What is the size of your family? Your family includes you, your spouse, and any dependents listed on Schedule J: Your Expenses (Official Form 106J).	Check all that apply: You Your spouse Your dependents	How many dependents?	Total number of per	ople
2.	Fill in your family's average monthly income. Include your spouse's income if your spouse is living with you, even if your spouse is not filing. Do not include your spouse's income if you are separated and your spouse is not filing with you.	value (if known) of any nor that you receive, such as fo Supplemental Nutrition Ass subsidies. If you have already filled on line 10 of that schedule.	spouse's income. Include the h-cash governmental assistance ood stamps (benefits under the sistance Program) or housing ut Schedule I: Your Income, see vernmental assistance that you	You Your spouse 4 Subtotal	That person's average monthly net income (take-home pay) \$
3.	Do you receive non-cash governmental assistance?	No Yes. Describe	Type of assistance		
4.	Do you expect your family's average monthly net income to increase or decrease by more than 10% during the next 6 months?	No Yes. Explain			
5	Tell the court why you are unable to installments within 120 days. If you h circumstances that cause you to not be fee in installments, explain them.	have some additional			

ebtor 1	First Name Middle	Name	Last Name				Case number (#	f known)	
Part 2:	Tell the Court				8				
Include a reported	e your average mo amounts paid by any I on line 2. ave already filled out	y governme	ent assistar		\$				
line 22 f	rom that form.								
who is r	e expenses cover not included in you rted in line 1?	anyone ır family	No Yes.	Identify who					
regulari expense If you ha Schedul	nyone other than y ly pay any of these es? ave already filled ou <i>le I: Your Income</i> , co m line 11.	t	No Yes.	How much do			is contributions'	? \$ montl	nly
monthly	expect your avera y expenses to incre se by more than 10 t 6 months?	ease or	No Yes.	Explain					
Part 3:	Tell the Court	About Ye	our Prope	erty	_				
If you hav	e already filled out	Schedule	A/B: Prop	erty (Official	Form 100	A/B) attach	copies to this	application and go	to Part 4.
n you nav	e alleady miled out	. oomoduno		.,					
Example your wa	uch cash do you have les: Money you have allet, in your home, a hen you file this app	e in Ind on	Cash:		\$			a.	
11. Bank a of mon	ccounts and other ey?	deposits			Institut	ion name:			Amount:
money	les: Checking, savin market, or other fina ts; certificates of de in banks, credit unic	ncial posit;	Checking Savings a						\$ \$
brokera	ige houses, and oth institutions. If you ha	er	Other fina	ncial accounts:					\$
more th same ir	nan one account with Institution, list each. I 401(k) and IRA acc	n the Do not	Other financial accounts:					\$	
	ome? (if you own it	outright or						Oursetuslus	•
•	chasing it) <i>les:</i> House, condom	inium	Number	Street				Current value: Amount you owe	\$
manufa	actured home, or mo	bile home	City			State	ZIP Code	on mortgage and liens:	\$
13. Other r	real estate?							Current value:	\$
			Number	Street				Amount you owe	e
			City			State	ZIP Code	on mortgage and liens:	φ
14. The ve	hicles you own?		Make:						
	les: Cars, vans, truc		Model:					Current value:	\$
	utility vehicles, moto s, boats	orcycles,	Year:					Amount you owe on liens:	\$
			Mileage						
			Make: Model:					Current value:	\$
			Year:			•		Amount you owe	
			Mileage					on liens:	\$

ebtor 1 First Name Middle Name	Last Name Case number (if known)	
 Other assets? Do not include household items and clothing. 	Describe the other assets: Current value: Amount you owe on liens:	\$ \$
6. Money or property due you? <i>Examples</i> : Tax refunds, past due or lump sum alimony, spousal support, child support, maintenance, divorce or property settlements, Social Security benefits, workers' compensation, personal injury recovery	\$ No \$ Yes	believe you will likely receivent in the next 180 days? . Explain:
Part 4: Answer These Additio	nal Questions	
17. Have you paid anyone for services for this case, including filling out this application, the bankruptcy filing package, or the schedules?	No Yes. Whom did you pay? Check all that apply: An attorney A bankruptcy petition preparer, paralegal, or typing service Someone else	How much did you pay? \$
18. Have you promised to pay or do you expect to pay someone for services for your bankruptcy case?	No Yes. Whom do you expect to pay? Check all that apply: An attorney A bankruptcy petition preparer, paralegal, or typing service Someone else	How much do you expect to pay? \$
19. Has anyone paid someone on your behalf for services for this case?	No Yes. Who was paid on your behalf? Check all that apply: Who paid? Check all that apply: An attorney Parent Brother or sister paralegal, or typing service Someone else Pastor or clergy Someone else	How much did someone else pay? \$
20. Have you filed for bankruptcy within the last 8 years?	No Yes. District When Case num MM/ DD/ YYYY	ber
	District When Case num	1ber
	District When Case num MM/ DD/ YYYY	
Part 5:Sign BelowBy signing here under penalty of pethat the information I provided in this	rjury, I declare that I cannot afford to pay the filing fee either in full or in instal s application is true and correct.	Iments. I also declare
Signature of Debtor 1	Signature of Debtor 2	
Date	Date MM / DD / YYYY	

Notice Required by 11 U.S.C. § 342(b) for Individuals Filing for Bankruptcy (Form 2010)

This notice is for you if:

You are an individual filing for bankruptcy, and

Your debts are primarily consumer debts. Consumer debts are defined in 11 U.S.C. § 101(8) as "incurred by an individual primarily for a personal, family, or household purpose."

The types of bankruptcy that are available to individuals

Individuals who meet the qualifications may file under one of four different chapters of the Bankruptcy Code:

- Chapter 7 Liquidation
- Chapter 11— Reorganization
- Chapter 12— Voluntary repayment plan for family farmers or fishermen
- Chapter 13— Voluntary repayment plan for individuals with regular income

You should have an attorney review your decision to file for bankruptcy and the choice of chapter.

Chapter 7: Liquidation

	\$245	filing fee
	\$75	administrative fee
+	\$15	trustee surcharge
	\$335	total fee

Chapter 7 is for individuals who have financial difficulty preventing them from paying their debts and who are willing to allow their nonexempt property to be used to pay their creditors. The primary purpose of filing under chapter 7 is to have your debts discharged. The bankruptcy discharge relieves you after bankruptcy from having to pay many of your pre-bankruptcy debts. Exceptions exist for particular debts, and liens on property may still be enforced after discharge. For example, a creditor may have the right to foreclose a home mortgage or repossess an automobile.

However, if the court finds that you have committed certain kinds of improper conduct described in the Bankruptcy Code, the court may deny your discharge.

You should know that even if you file chapter 7 and you receive a discharge, some debts are not discharged under the law. Therefore, you may still be responsible to pay:

- most taxes;
- most student loans;
- domestic support and property settlement obligations;

- most fines, penalties, forfeitures, and criminal restitution obligations; and
- certain debts that are not listed in your bankruptcy papers.

You may also be required to pay debts arising from:

- fraud or theft;
- fraud or defalcation while acting in breach of fiduciary capacity;
- intentional injuries that you inflicted; and
- death or personal injury caused by operating a motor vehicle, vessel, or aircraft while intoxicated from alcohol or drugs.

If your debts are primarily consumer debts, the court can dismiss your chapter 7 case if it finds that you have enough income to repay creditors a certain amount. You must file *Chapter 7 Statement of Your Current Monthly Income* (Official Form 122A–1) if you are an individual filing for bankruptcy under chapter 7. This form will determine your current monthly income and compare whether your income is more than the median income that applies in your state.

If your income is not above the median for your state, you will not have to complete the other chapter 7 form, the *Chapter 7 Means Test Calculation* (Official Form 122A–2).

If your income is above the median for your state, you must file a second form —the *Chapter 7 Means Test Calculation* (Official Form 122A–2). The calculations on the form—sometimes called the *Means Test*—deduct from your income living expenses and payments on certain debts to determine any amount available to pay unsecured creditors. If

your income is more than the median income for your state of residence and family size, depending on the results of the *Means Test*, the U.S. trustee, bankruptcy administrator, or creditors can file a motion to dismiss your case under § 707(b) of the Bankruptcy Code. If a motion is filed, the court will decide if your case should be dismissed. To avoid dismissal, you may choose to proceed under another chapter of the Bankruptcy Code.

If you are an individual filing for chapter 7 bankruptcy, the trustee may sell your property to pay your debts, subject to your right to exempt the property or a portion of the proceeds from the sale of the property. The property, and the proceeds from property that your bankruptcy trustee sells or liquidates that you are entitled to, is called *exempt property*. Exemptions may enable you to keep your home, a car, clothing, and household items or to receive some of the proceeds if the property is sold.

Exemptions are not automatic. To exempt property, you must list it on *Schedule C: The Property You Claim as Exempt* (Official Form 106C). If you do not list the property, the trustee may sell it and pay all of the proceeds to your creditors.

Chapter 11: Reorganization

	\$1,167	filing fee
+	\$550	administrative fee
	\$1,717	total fee

Chapter 11 is often used for reorganizing a business, but is also available to individuals. The provisions of chapter 11 are too complicated to summarize briefly.

Read These Important Warnings

Because bankruptcy can have serious long-term financial and legal consequences, including loss of your property, you should hire an attorney and carefully consider all of your options before you file. Only an attorney can give you legal advice about what can happen as a result of filing for bankruptcy and what your options are. If you do file for bankruptcy, an attorney can help you fill out the forms properly and protect you, your family, your home, and your possessions.

Although the law allows you to represent yourself in bankruptcy court, you should understand that many people find it difficult to represent themselves successfully. The rules are technical, and a mistake or inaction may harm you. If you file without an attorney, you are still responsible for knowing and following all of the legal requirements.

You should not file for bankruptcy if you are not eligible to file or if you do not intend to file the necessary documents.

Bankruptcy fraud is a serious crime; you could be fined and imprisoned if you commit fraud in your bankruptcy case. Making a false statement, concealing property, or obtaining money or property by fraud in connection with a bankruptcy case can result in fines up to \$250,000, or imprisonment for up to 20 years, or both. 18 U.S.C. §§ 152, 1341, 1519, and 3571.

Chapter 12: Repayment plan for family farmers or fishermen

	\$200	filing fee
+	\$75	administrative fee
	\$275	total fee

Similar to chapter 13, chapter 12 permits family farmers and fishermen to repay their debts over a period of time using future earnings and to discharge some debts that are not paid.

Chapter 13: Repayment plan for individuals with regular income

	\$235	filing fee
+	\$75	administrative fee
	\$310	total fee

Chapter 13 is for individuals who have regular income and would like to pay all or part of their debts in installments over a period of time and to discharge some debts that are not paid. You are eligible for chapter 13 only if your debts are not more than certain dollar amounts set forth in 11 U.S.C. § 109. Under chapter 13, you must file with the court a plan to repay your creditors all or part of the money that you owe them, usually using your future earnings. If the court approves your plan, the court will allow you to repay your debts, as adjusted by the plan, within 3 years or 5 years, depending on your income and other factors.

After you make all the payments under your plan, many of your debts are discharged. The debts that are not discharged and that you may still be responsible to pay include:

- domestic support obligations,
- most student loans,
- certain taxes,
- debts for fraud or theft,
- debts for fraud or defalcation while acting in a fiduciary capacity,
- most criminal fines and restitution obligations,
- certain debts that are not listed in your bankruptcy papers,
- certain debts for acts that caused death or personal injury, and
- certain long-term secured debts.

Warning: File Your Forms on Time

Section 521(a)(1) of the Bankruptcy Code requires that you promptly file detailed information about your creditors, assets, liabilities, income, expenses and general financial condition. The court may dismiss your bankruptcy case if you do not file this information within the deadlines set by the Bankruptcy Code, the Bankruptcy Rules, and the local rules of the court.

For more information about the documents and their deadlines, go to: http://www.uscourts.gov/forms/bankruptcy-forms

Bankruptcy crimes have serious consequences

- If you knowingly and fraudulently conceal assets or make a false oath or statement under penalty of perjury—either orally or in writing—in connection with a bankruptcy case, you may be fined, imprisoned, or both.
- All information you supply in connection with a bankruptcy case is subject to examination by the Attorney General acting through the Office of the U.S. Trustee, the Office of the U.S. Attorney, and other offices and employees of the U.S. Department of Justice.

Make sure the court has your mailing address

The bankruptcy court sends notices to the mailing address you list on *Voluntary Petition for Individuals Filing for Bankruptcy* (Official Form 101). To ensure that you receive information about your case, Bankruptcy Rule 4002 requires that you notify the court of any changes in your address. A married couple may file a bankruptcy case together—called a *joint case*. If you file a joint case and each spouse lists the same mailing address on the bankruptcy petition, the bankruptcy court generally will mail you and your spouse one copy of each notice, unless you file a statement with the court asking that each spouse receive separate copies.

Understand which services you could receive from credit counseling agencies

The law generally requires that you receive a credit counseling briefing from an approved credit counseling agency. 11 U.S.C. § 109(h). If you are filing a joint case, both spouses must receive the briefing. With limited exceptions, you must receive it within the 180 days **before** you file your bankruptcy petition. This briefing is usually conducted by telephone or on the Internet.

In addition, after filing a bankruptcy case, you generally must complete a financial management instructional course before you can receive a discharge. If you are filing a joint case, both spouses must complete the course.

You can obtain the list of agencies approved to provide both the briefing and the instructional course from: <u>http://www.uscourts.gov/services-forms/bankruptcy/credit-courseling-and-debtor-education-courses</u>.

In Alabama and North Carolina, go to: http://www.uscourts.gov/servicesforms/bankruptcy/credit-counseling-anddebtor-education-courses.

If you do not have access to a computer, the clerk of the bankruptcy court may be able to help you obtain the list.